

# Fluctuations and correlations in weakly asymmetric simple exclusion on a ring subject to an atypical current Benoit Dagallier

## ▶ To cite this version:

Benoit Dagallier. Fluctuations and correlations in weakly asymmetric simple exclusion on a ring subject to an atypical current. 2023. hal-04281437

# HAL Id: hal-04281437 https://hal.science/hal-04281437

Preprint submitted on 12 Nov 2023

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial - NoDerivatives 4.0 International License

# Fluctuations and correlations in weakly asymmetric simple exclusion on a ring subject to an atypical current

Benoit Dagallier\*

#### Abstract

We consider the weakly asymmetric simple exclusion process on a ring, driven out of equilibrium by tilting the dynamics so as to enforce a macroscopic current of particles on a large time interval. In this current-biased dynamics, the tilt by the current makes the dynamics non-local, non homogeneous and induces long-range correlations. We compute the correlation structure in the large time, large size limit for a certain range of asymmetry and current strength, recovering heuristic results of Bodineau et al. [15]. In addition, in this range of parameters, we characterise the full dynamics of fluctuations around the optimal density profile in the current-biased dynamics. The key ingredient at the microscopic scale is a precise relative entropy estimate at the level of correlations. We also discuss how to remove the (technical) restriction on the range of parameters.

# 1 Introduction

Consider a large number N of particles in a box, evolving in time and interacting locally. Assume that the dynamics of these particles can be modelled by a Markov chain  $\mathbb{P}^N$  on a finite, but large state space  $\Omega_N$  Consider also an observable  $\mathcal{O}_T$  of the trajectory of the Markov chain on a time interval [0, T], say of the following form: if  $(\eta_t)_{t\leq T}$  denotes a trajectory,

$$\mathcal{O}_T((\eta_t)_{t \le T}) = \int_0^T f(\eta_t) \, dt + \sum_{t \le T} g(\eta_{t_-}, \eta_t).$$
(1.1)

Above,  $f: \Omega_N \to \mathbb{R}$  is a test function that could for instance be the density of particles in a box. The function  $g: \Omega_N^2 \to \mathbb{R}$  counts changes in the chain and could for instance correspond to the activity or current.

Suppose also that we have some information on the observable  $\mathcal{O}_T$  in the sense of a large deviation principle, written informally as follows:

$$\frac{1}{T}\log \mathbb{P}^{N}(\mathcal{O}_{T} \approx Tx) \underset{T \to \infty}{=} -I_{f,g}^{N}(x), \qquad x \in \mathbb{R},$$
(1.2)

where  $I_{f,g}^N \ge 0$  is a rate function that may be known e.g. through results of Donsker and Varadhan [24]. The broad, informal question we wish to discuss is the following.

<sup>\*</sup>Courant Institute of Mathematical Sciences, New York University. E-mail: bd2543@nyu.edu

**Question.** Can one describe trajectories in the rare event  $\{\mathcal{O}_T \approx Tx\}$  in the large size, long time limits?

## **1.1 Biased dynamics**

The description of trajectories in a rare dynamical event has received much attention in the physics community in recent years, see [14, 15, 33, 18, 19, 3, 2] for a non exhaustive list of references as well as the review [32]. Studying trajectories in the rare event  $\{\mathcal{O}_T \approx Tx\}$  when  $N, T \gg 1$  boils down, following a standard large deviation idea, to studying typical trajectories under a tilted dynamics of the form (recall the expression (1.1) of  $\mathcal{O}_T$ ):

$$d\mathbb{P}^{N,f,g}_{\lambda,T}\big((\eta_t)_{t\leq T}\big) = \frac{\exp\left[\lambda\mathcal{O}_T\right]}{\mathbb{E}^N\big[\exp\left[\lambda\mathcal{O}_T\right]\big]} d\mathbb{P}^N\big((\eta(t))_{t\leq T}\big).$$
(1.3)

If  $\lambda \in \mathbb{R}$  is chosen appropriately as a function of x, then typical trajectories under (1.3) when N, T are large indeed correspond to trajectories in  $\{\mathcal{O}_T \approx Tx\}$ . We refer to [17] for a discussion of the choice of  $\lambda$ . A schematic description of trajectories in (1.3) is presented on Figure 1, see also [23] for a more in-depth discussion.

Although the original dynamics  $\mathbb{P}^N$  is a homogeneous Markov chain with local jump rates, the tilt by the observable  $\mathcal{O}_T$  of (1.1) makes the dynamics (1.3) much more complicated: its jump rates are not explicit, now time-dependent and non-local. In addition, depending on the tilt strength  $\lambda$ , typical trajectories under the biased dynamics (1.3) may have qualitatively different behaviour than under the original dynamics  $\mathbb{P}^N$ , for instance featuring dynamical phase transitions. Let us illustrate this point and more generally explain how the biased dynamics (1.3) can be studied by considering the example of the Weakly Asymmetric Simple Exclusion Process (WASEP for short) on a ring, tilted by the current. This is the dynamics that we will focus on in this article.

## **1.2** Macroscopic fluctuation theory and driven process

The WASEP on a ring with weak asymmetry  $E \in \mathbb{R}$  is an interacting particle system where particles follow nearest-neighbour random walks on a discrete torus with N sites, jumping to the right/left with rate proportional to 1 + E/N and 1 - E/N respectively. The only interaction between the particles comes from an exclusion rule: there can be at most one particle per site (see (2.2) for a precise definition).

Scaling limits of this model for both the density and current of particles have been studied in detail, see e.g. [38, Chapter 10], [6] and more recently [9] for joint large deviations in the large time, large N limits. In particular, it is known that there is typically a macroscopic current of particles on any time interval [0, T]. Also and interestingly, the dynamics is conjectured to undergo a dynamical phase transition. To state what this means, consider  $\lfloor \rho N \rfloor$  particles in the system ( $\rho \in (0, 1)$ ) and the event:

the time-integrated, space averaged current  $Q_T$  is equal to qT,  $q \in \mathbb{R}$ . (1.4)

One may ask about the typical density profile in this event in the large N, large T limit. The conjecture, formulated e.g. in [5, 14] and supported by numerical simulations [26], states that the result depends on the relationship between the tilt strength  $\lambda$  and the asymmetry E:

- if  $\frac{q^2}{\rho(1-\rho)} E^2\rho(1-\rho) > -\pi^2$ , then the optimal density profile is homogeneous in space and time. (1.5)
- if  $\frac{q^2}{\rho(1-\rho)} E^2\rho(1-\rho) < -\pi^2$ , then the optimal density profile is a travelling wave,

inhomogeneous in time and/or space.

In [14] the conjecture (1.5) was formulated assuming no discontinuous phase transition take place. Such transitions were later heuristically shown not to happen [3]. Note that in the symmetric simple exclusion process (E = 0), (1.5) predicts that the optimal density profile is flat for any value of  $\lambda$ . This is known to be true, consistent with the fact that this model does not have a dynamical phase transition [6]. The conjecture (1.5) has not been established rigorously in either region. It is only known that far above the threshold the flat density profile is optimal, and that far below the threshold a travelling wave profile is better [6] (but optimality of travelling waves is unknown).

A second interesting property of trajectories in the event (1.4) is that one expects the current constraint to create a long-range spatial correlation structure. The two-point correlation structure in particular was computed heuristically in [15], but nothing is known rigorously.

Let us now present different approaches employed in the literature to study trajectories in a rare event that can in particular be used to study the conjecture (1.5). These approaches are based on the study of WASEP dynamics tilted by the time-integrated, space-averaged current  $Q_T$ ; a particular example of (1.3). This dynamics, referred to as the *current-biased* dynamics throughout this article (see (2.7) for a precise definition), reads:

$$\mathrm{d}\mathbb{P}_{\lambda,E,T}^{curr,N} = \frac{1}{\mathbb{E}_E^N \left[ e^{\lambda N Q_T} \right]} e^{\lambda N Q_T} \mathrm{d}\mathbb{P}_E^N.$$
(1.6)

Above,  $\mathbb{P}_E^N, \mathbb{E}_E^N$  are the probability/expectations associated with the WASEP dynamics. In the exponential, the macroscopic current  $Q_T$  is multiplied by a factor N. In view of large deviations for the current in the WASEP [6], this ensures that (1.6) concentrates on trajectories in which the macroscopic current  $Q_T$  takes different values from the one under  $\mathbb{P}_E^N$ .

A major difficulty in studying the current-biased dynamics (1.6) comes from the fact that the state space is very large when  $N \gg 1$ . The first approach that can be used to study (1.6) completely bypasses the large state space difficulty through the formalism of the Macroscopic Fluctuation Theory (MFT), see the review [7]. There, one starts from a coarse grained, lowdimensional description of the microscopic dynamics, the details of which enter only through the choice of a diffusion coefficient and mobility. The long-time behaviour of the tilted dynamics is then directly studied at the level of the dynamical large deviation functional, see e.g. [5, 11, 49] and the recent [2]. Information on small fluctuations and correlations under (1.3) can be deduced from formal expansions of the large deviation functional. This is



Figure 1: Schematic representation of a trajectory of the current-biased dynamics (1.6). At times far before 0 and far after T, the influence of the current bias is not felt. In the yellow regions, the trajectory moves towards/away from a region of the state space with atypical current. It stays in this state (green region) for all but a negligible fraction of the time interval [0, T] when  $T \gg 1$ .

in particular how two-point correlations were computed in the current-biased dynamics [15]. Note that in this approach, one first takes the large N limit, then the large time limit.

The validity of the MFT has been established for a number of microscopic models (see the review [7], the book [38] as well as [4, 6, 10, 28, 16]), by proving dynamical large deviations for the density and/or current of particles. Thus, at the level of density large deviations, the measure (1.6) is well understood. In particular, for these microscopic models, dealing with the large size of the microscopic state seems to be a technical problem only, since the effective MFT description captures the physical features of the models. This should be the general picture.

On the other hand, the rigorous large deviation results do not give information on fluctuations, which require control of the dynamics on a finer scale. In particular and as already mentioned, there are no rigorous result on the correlation structure under the current-biased dynamics.

Another approach that has received a lot of attention in the physics literature consists in looking for an effective, but still microscopic description of the dynamics in the large time limit at fixed N. The macroscopic,  $N \gg 1$  limit in this approach is thus taken second, after the long time limit.

Let us describe this approach in more detail. At each fixed N but in the long time limit, typical trajectories in the rare event  $\{Q_T \approx Tq\}$  are expected to feature an instantaneous current approximately equal to q for all but a small portion of times in [0, T]. When the time T is large, one then expects the dynamics (1.6) to accurately be described in terms of a time-homogeneous Markovian dynamics with jump rates mimicking the effect of a constant current q. This homogeneous process, the *driven process* in the language of [18] (see Figure 1), can be built rigorously at each N, for general dynamics of the form (1.3), in terms of spectral elements of an explicit operator built from the initial dynamics and the observable by which the dynamics is biased. The driven process approach was expounded in detail in [18, 19] and also [33, 17, 34], following earlier work, see e.g. [27].

One advantage of the driven process framework compared to the MFT is that it can be defined directly at the microscopic level. Taking the large N limit is however difficult, as defin-

ing the driven process for N large requires solving a high-dimensional spectral problem, the limit of which must then be studied. Results have mostly bypassed this difficulty by studying the driven process in low-dimensional systems (i.e. at the macroscopic level) [47, 45, 22]. The recent paper [23] however managed to characterise the driven process in the large N limit for two interacting particle systems: independent particles, and the symmetric simple exclusion process connected with boundary reservoirs. In this latter case, the reservoir densities are assumed to be small, which together with the symmetry of the original dynamics enabled the authors to solve the spectral problem defining the driven process in a perturbative fashion. In this perturbative regime, the scaling limit of the driven process is identified in [23] and it is shown that the macroscopic fluctuation theory and driven process approach are consistent.

While one expects long time and scaling limits to commute at both the large deviation and fluctuation level, consistent with the validity of the MFT picture; proving the commutation is not an easy problem. It has been achieved only very recently, at the large deviation level and with a different motivation, for models such as the weakly asymmetric simple exclusion process on the on the torus in any dimension [9] or the one-dimensional simple exclusion with boundary reservoirs [13] (see also [8] for diffusions with small noise). It is not clear whether these commutation results are strong enough to apply also at the level of fluctuations.

### **1.3** Approximate driven process

In this paper, we study small fluctuations around the typical density profile in the currentbiased dynamics (1.6), which cannot be deduced from existing large deviation results. In a portion of the sub-critical regime (1.5) (i.e. before the dynamical phase transition), we obtain a full description of the non-local dynamics of density fluctuations at times  $1 \ll t \ll T$ (the green region of Figure 1) and compute the two-point correlation structure in this same region. The expression of two-point correlations that we find agrees with the one derived in [15] through a heuristic expansion of the large deviation functional.

Our approach builds on the driven process framework as in [22, 23], but with a different flavour.

The main difficulty in the driven process approach is to solve a high-dimensional spectral problem and study its large N limit. We bypass this difficulty by building an *approximate driven process*, that is not the correct one for each fixed N but has the correct macroscopic behaviour. This is done by using information on the macroscopic behaviour of the dynamics, so we take the large N limit first before the large T limit. The idea is that observables of interest, such as the density the current of particles or correlations, are determined at the macroscopic level by low-dimensional equations. These macroscopic equations are much easier to solve than the high-dimensional microscopic problem. One can then use the macroscopic data to build an informed microscopic approximation of the current-biased dynamics, the above mentioned approximate driven process (see Section 2.2 for more details).

Compared to the genuine driven process, the downside is that our approximate driven process has less structure. We therefore need additional stability estimates at the microscopic level compared to [23] to ensure that our approximate driven process indeed describes the microscopic dynamics to the desired precision. These estimates turn out to be quite involved and unfortunately restrict the range of asymmetries E and of the parameter  $\lambda$  in (1.6) that we can treat (extensions are discussed in Section 2.5).

The main technical tool to prove that the approximate driven process is close to the microscopic dynamics is a very precise relative entropy estimate. The relative entropy method was initially introduced by Yau [48] to study hydrodynamic limits. It consists in comparing the law of the dynamics to a simple measure that only retains macroscopic information on the observable under study, usually the density of particles. Jara and Menezes [36, 37] considerably improved the method, obtaining good enough estimates to study density fluctuations. In a recent paper [13], Bodineau and the author introduced a conceptual refinement of the relative entropy method, building on the techniques of Jara and Menezes. The main, though simple idea is that one can get better relative entropy bounds (meaning better control on the law of the dynamics) by adding more macroscopic information on the measure used for comparison, in particular by adding a correlation structure. A similar decomposition of the law of the dynamics was already suggested in [21]. The resulting better relative entropy bounds can be used in several contexts. They were for instance used in [13] to obtain long time large deviations for two-point correlations, whereas in the present paper they are key to proving that our approximate driven process indeed accurately describes fluctuations under the current-biased dynamics.

The article is structured as follows. Section 2 contains definitions and results, with Section 2.4 detailing the structure of the proof of the main result, the characterisation of the density fluctuation process under the current-biased dynamics. Extensions of the result are discussed in Section 2.5. Section 3 contains a collection of preliminary results useful from Section 4 onwards where we start the proof of the main result. In particular, we study in Section 3 fluctuations for a family of dynamics the approximate driven process is later shown to belong to. The main technical input, the relative entropy estimate, is also established in this section. Section 4 identifies a candidate for the approximate driven process and provides a stability estimate to ensure that this candidate indeed contains all information on fluctuations under the current biased dynamics. At this point the proof of our main can be reduced to estimates involving only the limiting  $(N \gg 1)$  fluctuations under the approximate driven process. Section 5 contains the long-time analysis of these fluctuations and concludes the proof of the main result. The two-point correlations structure is computed in Section 6 and compared with [15]. Useful concentration of measure estimates and other technical results are gathered in the appendix.

# 2 Definition and results

#### 2.1 Model and notations

**The WASEP.** For  $N \in \mathbb{N}_{\geq 1}$ , Let  $\Omega_N = \{0, 1\}^{\mathbb{T}_N}$  denote the state space of the system, with  $\mathbb{T}_N := \mathbb{Z}/N\mathbb{Z}$  the discrete torus with N sites. A point  $i \in \mathbb{T}_N$  is called a site, while elements  $\eta = (\eta_i)_{i \in \mathbb{T}_N} \in \Omega_N$  are called (particle) configurations, with  $\eta_i = 1$  if there is a particle at site  $i \in \mathbb{T}_N$ , and  $\eta_i = 0$  otherwise. The quantity  $\eta_i$  is the occupation number at site  $i \in \mathbb{T}_N$ . For  $\eta \in \Omega_N$  and  $i, j \in \mathbb{T}_N$ , define the configuration  $\eta^{i,j} \in \Omega_N$  with exchanged occupation

numbers at sites i, j:

$$\eta^{i,j}(k) = \begin{cases} \eta(\ell) & \text{if } \ell \notin \{i,j\}, \\ \eta(j) & \text{if } \ell = i, \\ \eta(i) & \text{if } \ell = j. \end{cases}$$
(2.1)

The WASEP with asymmetry  $E \in \mathbb{R}$  is the dynamics with generator  $N^2 L_E$ , acting on  $\phi : \Omega_N \to \mathbb{R}$  according to:

$$N^{2}L_{E}\phi(\eta) = N^{2}\sum_{i\in\mathbb{T}_{N}} c_{E}(\eta,\eta^{i,i+1}) \big[\phi(\eta^{i,i+1}) - \phi(\eta)\big],$$
(2.2)

with:

$$c_E(\eta, \eta^{i,i+1}) = c(\eta, \eta^{i,i+1})e^{-(\eta_{i+1} - \eta_i)E/N}, \quad c(\eta, \eta^{i,i+1}) := \eta_i(1 - \eta_{i+1}) + \eta_{i+1}(1 - \eta_i).$$
(2.3)

Note that the number of particles  $\sum_{i \in \mathbb{T}_N} \eta_i$  is conserved by the dynamics (2.2). The factor  $N^2$  in front of the generator (2.2) corresponds to a diffusive rescaling of time. Trajectories  $\eta(\cdot) = (\eta(t))_{t\geq 0}$  belong to the Skorokhod space  $\mathcal{D}(\mathbb{R}_+, \Omega_N)$  of left-continuous, right-limited  $\Omega_N$ -valued functions, and we write  $\mathbb{P}_E^{\mu^N}, \mathbb{E}_E^{\mu^N}$  for the probability/expectation of the dynamics starting from an initial probability distribution  $\mu^N$  on  $\Omega_N$ .

The invariant measures of the WASEP are well-understood (see e.g. [38], Chapter 3-4). For  $\rho \in [0, 1]$ , let  $\nu_{\rho}^{N}$  denote the Bernoulli product measure with parameter  $\rho$ :

$$\nu_{\rho}^{N} := \bigotimes_{i \in \mathbb{T}_{N}} \operatorname{Ber}(\rho), \qquad \operatorname{Ber}(\rho) : x \in \{0, 1\} \mapsto \rho x + (1 - \rho)(1 - x).$$
(2.4)

Then  $\nu_{\rho}^{N}$  is invariant for the WASEP:

$$\forall i \in \mathbb{T}_N, \forall \eta \in \Omega_N, \qquad c_E(\eta, \eta^{i,i+1})\nu_\rho^N(\eta) = \sum_{j \in \mathbb{T}_N} c_E(\eta^{j,j+1}, \eta)\nu_\rho^N(\eta^{j,j+1}).$$
(2.5)

In particular, for each  $\rho \in [0, 1]$ , correlations under the product measure  $\nu_{\rho}^{N}$  are trivial.

**The current.** Let  $Q_T$  denote the space averaged, time-integrated current on [0, T] (T > 0): if  $N_T(i \to j)$  is the number of jumps from site  $i \in \mathbb{T}_N$  to site  $j \in \mathbb{T}_N$  on [0, T],

$$\forall \eta(\cdot) \in \mathcal{D}(\mathbb{R}_+, \Omega_N), \qquad Q_T(\eta(\cdot)) := \frac{1}{N^2} \sum_{i \in \mathbb{T}_N} \left[ N_T(i \to i+1) - N_T(i+1 \to i) \right]. \tag{2.6}$$

The additional factor of  $N^{-1}$  reflects the diffusive scaling in time: if there is a macroscopic current of particles, i.e. if the net number of particles having gone through a bond  $(i, i+1), i \in \mathbb{T}_N$  up to time T is of order N, then  $Q_T$  is of order 1 in N.

The current-biased dynamics. We now define the dynamics of interest in this work. For a time T > 0,  $\lambda \in \mathbb{R}$  and an initial probability distribution  $\mu^N$  of particles, define the *current-biased dynamics*  $\mathbb{P}^{curr,\mu^N}_{\lambda,E,T}$  on  $\mathcal{D}(\mathbb{R}_+,\Omega_N)$  as:

$$\mathbb{P}_{\lambda,E,T}^{curr,\mu^{N}}(\cdot) = \frac{\mathbb{E}_{E}^{\mu^{N}} \left[ \mathbf{1}.e^{\lambda NQ_{T}} \right]}{\mathbb{E}_{E}^{\mu^{N}} \left[ e^{\lambda NQ_{T}} \right]}.$$
(2.7)

Under (2.7), by [6], the probability of observing a macroscopic current  $q = \rho(1 - \rho)(\lambda + E)$ of particles on [0, T] goes to 1 when N becomes large for a wide range of initial conditions enforcing a density  $\rho \in (0, 1)$  of particles. The sub-critical range (1.5) of parameters  $\lambda, E$  for which there is no dynamical phase transition can then be rewritten as:

$$\rho(1-\rho)\lambda(\lambda+2E) > -\pi^2. \tag{2.8}$$

We stress again that, in contrast to the WASEP dynamics, the tilt by the current in (2.7) makes the dynamics non-local, inhomogeneous in time and induces long-range correlations (see [15] for heuristics and [18] for a general discussion of properties of tilted dynamics). Note also that we consider an "annealed" version of the dynamics, where both numerator and denominator in (2.7) are averaged on the initial condition. Consequences of this assumption and "quenched" dynamics are discussed in Section 2.5.

Initial condition, fluctuations and correlations. In this article, we aim to describe the fluctuations and correlations of the dynamics (2.7) when N, then T are large, for subcritical parameters  $\lambda$ , E (i.e. satisfying (2.8)) and in the intermediate regime where there is a macroscopic instantaneous current (the green region of Figure 1). In the text, we refer to 0, T as time boundaries and may refer to times  $1 \ll t \ll T$  as being far away from the time boundaries. For simplicity we will work at density 1/2, with initial condition in (2.7) given by:

$$\mu^{N}(\eta) = \nu_{1/2}^{N}(\eta) := \bigotimes_{i \in \mathbb{T}_{N}} \operatorname{Ber}(1/2)(\eta_{i}), \qquad \eta \in \Omega_{N}.$$
(2.9)

Other initial conditions are discussed in Section 2.5. Since we focus on the sub-critical regime, the density profile under the current-biased dynamics is typically going to be constant equal to 1/2. The fluctuation and correlation fields of interest are then defined as follows. Write:

$$\bar{\eta}_i := \eta_i - \frac{1}{2}, \qquad \eta \in \Omega_N, i \in \mathbb{T}_N.$$
(2.10)

The fluctuation field  $Y^N$  is a distribution, acting on test functions  $f: \mathbb{T} \to \mathbb{R}$  according to:

$$Y^{N}(f) = Y^{N}(f)(\eta) = \frac{1}{N^{1/2}} \sum_{i \in \mathbb{T}_{N}} \bar{\eta}_{i} f(i/N) \qquad \eta \in \Omega_{N}.$$
 (2.11)

The correlation field  $\Pi^N$  is defined as the "square" of  $Y^N$  in the following sense: if  $f_1, f_2 : \mathbb{T} \to \mathbb{R}$ , write  $f_1 \otimes f_2(x, y) = f_1(x)f_2(y)$  for  $x, y \in \mathbb{T}$ . Then:

$$\Pi^{N}(f_{1} \otimes f_{2}) = \frac{1}{4} Y^{N}(f_{1}) Y^{N}(f_{2}) - \frac{1}{4N} \sum_{i \in \mathbb{T}_{N}} (\bar{\eta}_{i})^{2} f_{1}(i/N) f_{2}(i/N).$$
(2.12)

The factor 1/4 is a convenient normalisation. More generally, for  $\phi : \mathbb{T}^2 \to \mathbb{R}$ , we define:

$$\Pi^{N}(\phi) = \frac{1}{4N} \sum_{i \neq j \in \mathbb{T}_{N}} \bar{\eta}_{i} \bar{\eta}_{j} \phi(i/N, j/N) \qquad \eta \in \Omega_{N}.$$
(2.13)

In the text, "correlations" always refers to two-point correlations of the form  $\bar{\eta}_i \bar{\eta}_j$ ,  $i \neq j \in \mathbb{T}_N$ unless otherwise mentioned. More generally, *n*-point correlations  $(n \in \mathbb{N}_{\geq 1})$  refer to any product of the form  $\bar{\eta}_{i_1}...\bar{\eta}_{i_n}$ . By convention, if  $\phi : \mathbb{T}^n \to \mathbb{R}$ , we write  $\phi_{i_1,...,i_n}$  for  $\phi(i_1/N,...,i_n/N)$ ,  $i_1,...,i_n \in \mathbb{T}_N^n$ . Letters  $i, j, \ell$  are reserved for discrete indices, while x, y, z denote continuous variables.

**Spaces of test functions.** Let  $S'(\mathbb{T}), S'(\mathbb{T}^2)$  denote the space of distributions on  $\mathbb{T}, \mathbb{T}^2$  respectively. The fluctuation field  $Y^N$  is viewed as a random element in  $S'(\mathbb{T})$ . To study of the correlation field  $\Pi^N$ , we will have to consider test functions which are continuous, but have discontinuous normal derivatives across the diagonal  $D = \{(x, x) : x \in \mathbb{T}\}$  of  $\mathbb{T}^2$ . This is related to the discontinuity of out of equilibrium correlations across the diagonal D, see [44, 13], and in particular would not be needed to study correlations correlations at equilibrium [30]. Write for short  $\{x \leq y\} := \{(x, y) \in \mathbb{T}^2 : x \leq y\}, \{x \geq y\} := \{(x, y) \in \mathbb{T}^2 : x \geq y\}$ . The set of test functions  $C_D^{\infty}(\mathbb{T}^2)$  is given by all continuous functions on  $\mathbb{T}^2$  such that their restrictions on  $\{x \leq y\}$  and  $\{x \geq y\}$  are  $C^{\infty}$  (the derivatives of the extensions on the diagonal D need not coincide)

$$C_D^{\infty}(\mathbb{T}^2) = \left\{ \phi \in C^0(\mathbb{T}^2) : \phi_{|\{x \le y\}} \in C^{\infty}(\{x \le y\}), \phi_{|\{x \ge y\}} \in C^{\infty}(\{x \ge y\}) \right\}.$$
(2.14)

The correlation field  $\Pi^N$  is seen as a random distribution in the set  $\mathcal{S}'_D(\mathbb{T}^2)$  of bounded linear forms on  $C^{\infty}_D(\mathbb{T}^2)$ .

A trajectory  $\eta(\cdot) \in \mathcal{D}(\mathbb{R}_+, \Omega_N)$  induces a process  $Y_{\cdot}^N = (Y_t^N)_{t \geq 0}$ , which we refer to as the fluctuation process. It is a random element of the Skorokhod space  $\mathcal{D}(\mathbb{R}_+, \mathcal{S}'(\mathbb{T}))$ , equipped with its usual topology, see e.g. [12]. The correlation process similarly denotes the random element  $\Pi_{\cdot}^N = (\Pi_t^N)_{t \geq 0}$  of  $\mathcal{D}(\mathbb{R}_+, \mathcal{S}'_D(\mathbb{T}^2))$ .

## 2.2 Goal and approach: the approximate driven process

In this section, we define what we mean by approximate driven process as mentioned in Section 1.3. We seek to characterise fluctuations and correlations for the current-biased dynamics (2.7), in the large N, then large T limits; for intermediate times far away from the time boundaries 0, T (corresponding to the green region in Figure 1). In this regime, for sub-critical values (2.8) of  $\lambda, E$  for which the best way to realise the current constraint is by having a constant density profile, one expects the current-biased dynamics to be well approximated by a stationary Markov process as explained in the introduction. Our aim, then, is to find a Markov process  $\tilde{\mathbb{P}}^N$  which describes fluctuations and correlations under (2.7) in this intermediate time range. For this Markov process  $\tilde{\mathbb{P}}^N$  to describe the fluctuations under the current biased-dynamics it should satisfy, for a good choice of initial condition  $\mu^N$ , any bounded continuous  $F : \mathcal{D}(\mathbb{R}_+, \mathcal{S}'(\mathbb{T})) \to \mathbb{R}$  and each  $t, T, \tau > 0$  with  $t + \tau \leq T$ :

$$\int F((Y^N)_{t \le s \le t+\tau}) \, \mathrm{d}\mathbb{P}^{curr,\mu^N}_{\lambda,E,T} = \int F((Y^N_s)_{0 \le s \le \tau}) \, \mathrm{d}\tilde{\mathbb{P}}^N + \varepsilon^N_{\tau,t,T}(F), \tag{2.15}$$

with  $\varepsilon_{\tau,t,T}^N(F)$  a term that must be small, in the large N limit, for times away from the time boundaries:

$$\lim_{t \to \infty} \limsup_{T \to \infty} \limsup_{N \to \infty} \left| \varepsilon_{\tau, t, T}^N(F) \right| = 0.$$
(2.16)

We call any  $\tilde{\mathbb{P}}^N$  satisfying (2.15)–(2.16) an *approximate driven process* in analogy with the driven process of [18].

Let us now give more details on the construction of an approximate driven process. To find such a process, we build the jump rates of  $\tilde{\mathbb{P}}^N$  explicitly, according to the following considerations.

• For any T > 0 and  $\lambda \neq 0$ , the typical current  $Q_T$  under the current-biased dynamics (2.7) is different from the current under the WASEP  $\mathbb{P}^N_E$ . One expects the currentbiased dynamics to be better approximated by a dynamics that has the correct macroscopic current. The simplest way to modify  $\mathbb{P}^N_E$  into a dynamics that has the correct current is to change the asymmetry from E to  $\lambda + E$ , with jump rates:

$$c_{\lambda+E}(\eta, \eta^{i,i+1}) := c_E(\eta, \eta^{i,i+1}) e^{-N^{-1}(\eta_{i+1} - \eta_i)\lambda}, \qquad i \in \mathbb{T}_N, \eta \in \Omega_N.$$
(2.17)

The resulting WASEP dynamics  $\mathbb{P}_{\lambda+E}^N$  is, by construction, a good approximation of the current-biased dynamics as far as the density and current of particles are concerned. However, it does not have the correct correlation structure (recall that product Bernoulli measures with constant parameter are invariant for the WASEP). It thus does not have the same density fluctuations and therefore cannot serve as an approximate driven process as defined in (2.15).

• We then tune observables on the next finer scale than the current, that is we tune the jump rates to obtain a dynamics that has the same two-point correlation structure as the current-biased dynamics. We expect these correlations to be non-local, so we need non-local jump rates. An effective way to tune correlations without changing the density/current in the large N limit [13] is to consider, for a symmetric function  $h: \mathbb{T}^2 \to \mathbb{R}$  that we call a bias below, the jump rates:

$$c_{h,\lambda+E}(\eta,\eta^{i,i+1}) := c_{\lambda+E}(\eta,\eta^{i,i+1})e^{\Pi^{N}(h)(\eta^{i,i+1}) - \Pi^{N}(h)(\eta)}, \qquad i \in \mathbb{T}_{N}, \eta \in \Omega_{N}.$$
(2.18)

Let  $\mathbb{P}_{h,\lambda+E}^N$  be the corresponding dynamics. We then optimise the bias h so that  $\mathbb{P}_{h,\lambda+E}^N$  and the current-biased dynamics are, loosely speaking, as close to each other as possible.

As stated in the next section, tuning the current/density and two-point correlations is enough and  $\mathbb{P}_{h,\lambda+E}^{N}$  is an approximate driven process in the sense of (2.15) for a suitable bias h.

#### 2.3 Results

A first result of this work is the computation of the correlation structure under the currentbiased dynamics in the green region of Figure 1, in Proposition 2.3. The correlation structure is related to the optimal bias h that one must choose in order for  $\mathbb{P}^N_{h,\lambda+E}$  to be an approximate driven process. This bias is characterised next. We start with some notations. Any  $\phi \in C_D^{\infty}(\mathbb{T}^2)$  is identified with the kernel operator  $\phi f = \int_{\mathbb{T}} \phi(x, \cdot) f(\cdot) dx$  for  $f \in \mathbb{L}^2(\mathbb{T})$ . We say that  $\phi$  is a positive kernel if:

$$\forall f \in \mathbb{L}^2(\mathbb{T}), \qquad \int_{\mathbb{T}^2} f(x)\phi(x,y)f(y)\,dx\,dy \ge 0. \tag{2.19}$$

A negative kernel corresponds to the opposite sign. Define also:

$$\sigma := \rho(1-\rho)|_{\rho=1/2} = \frac{1}{4}.$$
(2.20)

For  $\lambda, E \in \mathbb{R}$ , consider the following differential equation with unknown  $h : \mathbb{T} \to \mathbb{R}$ :

$$\begin{cases} h''(x) - \frac{\sigma}{2} \int_{\mathbb{T}} h'(x-z)h'(z) \, dz = 0 \quad \text{for } x \in (0,1), \\ h'(0^+) - h'(1^-) = 2\lambda(\lambda + 2E). \end{cases}$$
(2.21)

Solutions to (2.21) are continuous periodic functions with derivative having a jump.

**Proposition 2.1.** Let  $\lambda, E \in \mathbb{R}$  be sub-critical parameters at density 1/2 as in (2.8):

$$\sigma\lambda(\lambda+2E) > -\pi^2. \tag{2.22}$$

There is then a unique family  $(h_{\lambda,E})_{\lambda,E \text{ sub-critical}}$  of solutions in  $C^0(\mathbb{T}) \cap C^{\infty}([0,1])$  of (2.21) such that  $\int_{\mathbb{T}} h_{\lambda,E}(x) dx = 0$  and:

 $(\lambda, E) \mapsto ||h_{\lambda,E}||_2$  is a continuous function vanishing on the line  $\lambda = 0.$  (2.23)

The function  $h_{\lambda,E}$  is explicitly given by:

$$\forall x \in \mathbb{T}, \qquad h_{\lambda,E}(x) = \frac{\sqrt{2}}{\sigma} \sum_{\ell \ge 1} \left( 1 - \left[ 1 + \frac{\sigma\lambda(\lambda + 2E)}{\pi^2 \ell^2} \right]^{1/2} \right) \sqrt{2} \cos(2\pi\ell x). \tag{2.24}$$

The kernel  $(x, y) \in \mathbb{T}^2 \mapsto h_{\lambda,E}(x - y)$ , still denoted  $h_{\lambda,E}$ , is a symmetric kernel in  $C_D^{\infty}(\mathbb{T}^2)$ with eigenvalues given by the Fourier coefficients of  $h_{\lambda,E}$ . In particular,  $h_{\lambda,E}$  is a negative kernel if  $\lambda(\lambda + 2E) \geq 0$ , a positive kernel if  $\lambda(\lambda + 2E) \leq 0$  and the critical line can be expressed in terms of the behaviour of the eigenvalues of  $h_{\lambda,E}$ :

$$\sigma\lambda(\lambda+2E) > -\pi^2 \quad \Leftrightarrow \quad \sigma h_{\lambda,E} \text{ has leading eigenvalue strictly smaller than 1.}$$
 (2.25)

**Remark 2.2.** The family  $(h_{\lambda,E})$  is not unique without the continuity requirement (2.23), see Appendix C. This requirement is however physically meaningful. Indeed, the current-biased dynamics with  $\lambda = 0$  case corresponds to the WASEP  $\mathbb{P}_E^N$  for which there is no correlation in the steady state. Proposition 2.3 shows that h determines correlations under the currentbiased dynamics which in particular depend continuously on  $\lambda$ .

In the following, the subscripts  $\lambda, E$  will be dropped and we simply write h for  $h_{\lambda,E}$ .

**Proposition 2.3** (Correlations under the current biased dynamics at times  $1 \ll t \ll T$ ). Let  $\lambda, E \in \mathbb{R}$ . Assume that  $\lambda(\lambda + 2E) \geq 0$  and is small enough (in particular  $\lambda, E$  are sub-critical (2.22)). Then, for any  $f_1, f_2 \in C^{\infty}(\mathbb{T})$ :

$$\lim_{T \to \infty} \lim_{N \to \infty} \mathbb{E}_{\lambda, E, T}^{curr, \nu_{1/2}^{N}} [Y_{T/2}^{N}(f_1) Y_{T/2}^{N}(f_2)] = \frac{1}{4} \int_{\mathbb{T}^2} f_1(x) \big( (\sigma id + k_{curr}) f_2 \big)(x) \, dx, \tag{2.26}$$

with  $k_{curr} \in C^{\infty}([0,1]) \cap C^0(\mathbb{T})$  identified with the kernel  $(x,y) \in \mathbb{T}^2 \mapsto k_{curr}(x-y)$  and given explicitly by:

$$k_{curr}(x) = \sqrt{2}\sigma \sum_{\ell \ge 1} \left[ \frac{1}{\sqrt{1 + \frac{\lambda(\lambda + 2E)\sigma}{\pi^2 \ell^2}}} - 1 \right] \sqrt{2} \cos(2\pi\ell x).$$
(2.27)

In addition,  $k_{curr}$  is related to h through:

$$(\sigma^{-1}id - h)^{-1} = \sigma id + k_{curr}.$$
(2.28)

The expression (2.27) of  $k_{curr}$  agrees with the one derived in [15]. Yet another equivalent formulation of the sub-critical region (2.22) is then as the region in which  $k_{curr}$  has bounded largest eigenvalue.

Proposition 2.3 is obtained as a corollary of the next theorem, the main result of this work, which characterises not just the spatial correlation structure but the full dynamics of fluctuations under the current-biased dynamics: we show that  $\mathbb{P}_{h,\lambda+E}^{N}$  is an approximate driven process in the sense of (2.15). To state the result precisely, introduce a discrete approximation  $\nu_{h}^{N}$  of a Gaussian measure:

$$\nu_h^N := \frac{e^{2\Pi^N(h)}}{\mathcal{Z}_h^N} \,\nu_{1/2}^N. \tag{2.29}$$

Above,  $\mathcal{Z}_h^N$  is a normalisation factor. Properties of this measure are analysed in Appendix C, in particular it converges to a Gaussian field with covariance  $\sigma id + k_{curr}$ . It is technical but convenient to work with the dynamics  $\mathbb{P}_{h,\lambda+E}^N$  started from  $\nu_h^N$  as it is a good approximation of its invariant measure, see Theorems 2.5–2.7 below.

**Theorem 2.4.** Let  $\lambda, E \in \mathbb{R}$ . Assume that  $\lambda(\lambda + 2E) \geq 0$  and is small enough (in particular (2.22) holds). Then  $\mathbb{P}_{h,\lambda+E}^{\nu_h^N}$  is an approximate driven process for the current-biased dynamics  $\mathbb{P}_{\lambda,E,T}^{curr,\nu_{1/2}^N}$ . That is, for any bounded continuous  $F : \mathcal{D}(\mathbb{R}_+, \mathcal{S}'(\mathbb{T})) \to \mathbb{R}$  and  $t, T, \tau > 0$  with  $t + \tau \leq T$ :

$$\mathbb{E}_{\lambda,E,T}^{curr,\nu_{1/2}^N} \left[ F\left( (Y^N)_{t \le s \le t+\tau} \right) \right] = \mathbb{E}_{h,\lambda+E}^{\nu_h^N} \left[ F\left( (Y^N_s)_{0 \le s \le \tau} \right) \right] + \varepsilon_{[t,t+\tau]}^{N,T}(F), \tag{2.30}$$

with:

$$\limsup_{t \to \infty} \limsup_{T \to \infty} \limsup_{N \to \infty} \sup_{t \to \infty} \left| \varepsilon_{[t, t+\tau]}^{N, T}(F) \right| = 0.$$
(2.31)

In (2.30),  $t, \tau$  can also be taken to diverge with T provided  $T - \tau$  diverges as well.

We expect the claim of Theorem 2.4 to be valid throughout the sub-critical regime (2.22), with some care required about the initial condition. This and extensions of Theorem 2.4 are discussed in Section 2.5. Let us however already mention that Theorem 2.4 is the only theorem for which we need to restrict to small  $\lambda(\lambda + 2E) \geq 0$ . All other Theorems are valid in the entire sub-critical regime (2.22) as we shall see.

Equation (2.30) implies that the fluctuation process under the current biased dynamics converges to the limiting fluctuations under  $\mathbb{P}^N_{h,\lambda+E}$ . These fluctuations can be characterised precisely, as stated next in Theorem 2.5 where a more general study of dynamics of the form  $\mathbb{P}^N_{\tilde{h},\tilde{\lambda}}$  is carried out for  $\tilde{\lambda} \in \mathbb{R}$  and  $\tilde{h} \in C^{\infty}_D(\mathbb{T}^2)$  (i.e. not just for  $\tilde{h} = h$  or  $\tilde{\lambda} = \lambda + E$ ). **Theorem 2.5.** Let  $\tilde{\lambda} \in \mathbb{R}$  and let  $\tilde{h} \in C_D^{\infty}(\mathbb{T}^2)$  be symmetric and such that  $\sigma \tilde{h}$  has leading

eigenvalue strictly below 1. Define  $\nu_{\tilde{h}}^{N}$  as in (2.29). The law of  $(Y^{N}, \Pi^{N})$  under  $\mathbb{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}}$  then converges weakly in  $\mathcal{D}(\mathbb{R}_{+}, \mathcal{S}'(\mathbb{T}) \times \mathcal{S}'_{D}(\mathbb{T}^{2}))$  to the law of a couple  $(Y, \Pi)$  satisfying:

- 1. The law  $\nu_{\tilde{h}}^{\infty}$  of  $Y_0$  is a Gaussian field on  $\mathcal{S}'(\mathbb{T})$ , with covariance  $C_{\tilde{h}} := (\sigma^{-1}id \tilde{h})^{-1}$ . Moreover, Y. and  $\Pi$ . are stationary processes.
- 2. For any  $\phi \in C_D^{\infty}(\mathbb{T}^2)$ , if  $\phi^s$  denotes the symmetric part of  $\phi$ , then  $\Pi_{\cdot}(\phi) = \Pi_{\cdot}(\phi^s)$ .
- 3. Y. is uniquely characterised by the following martingale problem. For each test function  $f \in C^1(\mathbb{R}_+, C^{\infty}(\mathbb{T}))$ , the processes  $M_{\cdot}(f), \mathfrak{M}_{\cdot}(f)$  are continuous martingales with respect to the canonical filtration  $(\mathcal{F}_t)_{t>0}$  generated by Y.:

$$\forall t \ge 0, \qquad M_t(f) = Y_t(f_t) - Y_0(f_t) - \int_0^t Y_s \Big(\partial_s f_s + \mathcal{L}_{\tilde{h}} f_s\Big) ds$$
$$\mathfrak{M}_t(f) = \Big(M_t(f)\Big)^2 - 2\sigma \int_0^t \|\partial_x f_s\|_2^2 dx, \qquad (2.32)$$

with  $\mathcal{L}_{\tilde{h}}$  the operator acting on  $f \in C^{\infty}(\mathbb{T})$  according to:

$$\mathcal{L}_{\tilde{h}}f(x) = f''(x) + \sigma \int_{\mathbb{T}} \partial_2 \tilde{h}(x, z) f'(z) \, dz = (id - \sigma \tilde{h}) \Delta f(x), \qquad x \in \mathbb{T}.$$
 (2.33)

4. The processes  $Y_{\cdot}, \Pi_{\cdot}$  are related as follows:

$$\forall (f_1, f_2) \in C^{\infty}(\mathbb{T})^2, \qquad \Pi_{\cdot}(f_1 \otimes f_2) = \frac{1}{4} Y_{\cdot}(f_1) Y_{\cdot}(f_2) - \frac{\sigma}{4} \int_{\mathbb{T}} f_1(x) f_2(x) \, dx. \tag{2.34}$$

5. (Non-smooth test functions and bound on correlations) The process  $\Pi$  admits a unique extension to a process in  $\mathcal{D}([0,+\infty), \mathcal{S}'_{D}(\mathbb{T}^{2}))$  (i.e. also defined on test functions that may have discontinuities on the diagonal). If  $p \in \mathbb{N}$  and  $t_1, ..., t_p \geq 0, \phi_1, ..., \phi_p \in$  $\mathcal{C}_D^{\infty}(\mathbb{T}^2)$ , then the vector  $\left(\Pi_{t_1}^N(\phi_1), ..., \Pi_{t_p}^N(\phi_p)\right)$  converges weakly to  $\left(\Pi_{t_1}(\phi_1), ..., \Pi_{t_p}(\phi_p)\right)$ . In addition, there is  $C(\tilde{h}, \tilde{\lambda}) > 0$  such that, for each  $\varepsilon > 0$ :

$$\mathbf{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{\infty}}\left(|\Pi_{t}(\phi_{1})| > \varepsilon\right) \leq \frac{C(\tilde{h},\tilde{\lambda})\|\phi_{1}\|_{\infty}}{\varepsilon}, \qquad t \geq 0.$$

$$(2.35)$$

• In the special case where  $\tilde{h}$  commutes with the Laplacian, the operator Remark 2.6.  $\mathcal{L}_{\tilde{h}}$  is self-adjoint and the associated fluctuations are reversible. This is the case for translation-invariant  $\tilde{h}$ , in particular the fluctuations under the current-biased dynamics are reversible as in that case  $\tilde{h} = h$ .

• Note that the parameter  $\tilde{\lambda}$  does not appear in the martingale problem of Theorem (2.5). This is due to the fact that we consider fluctuations around density 1/2. At density  $\rho \neq 1/2$ , there would be a term proportional to  $\tilde{\lambda}\sigma'(\rho)\int_0^t Y_s(\partial_x f_s)\,ds$  in  $M_t(f)$  in (2.32), with  $\sigma(\rho) = \rho(1-\rho)$  (and in particular even for translation-invariant h the resulting fluctuations would not be reversible).

Item 3 of Theorem 2.5 is essentially proven in [36]. The key ingredient there is a refinement of the relative entropy method of Yau [48]. This refinement has been used in [36]–[37] to characterise out of equilibrium fluctuations in dimension d < 4. Putting Theorem 2.4 and Theorem 2.5 together, one obtains that the fluctuations under the current-biased dynamics in the intermediate regime  $1 \ll t \ll T$  (the green region of Figure 1) are given by the only solution of the infinite-dimensional Ornstein-Uhlenbeck process:

$$dY_t = \mathcal{L}_h^* Y_t \, dt + \sqrt{2\sigma} \, \nabla dW_t, \qquad (2.36)$$

with dW. a space-time white noise and  $\mathcal{L}_h^* = \mathcal{L}_h$  the non-local, self-adjoint operator defined in (2.33), acting on  $Y_t$  by  $\mathcal{L}_h Y_t(f) = Y_t(\mathcal{L}_h f)$  ( $f \in C^{\infty}(\mathbb{T})$ ). The non-locality of the drift term  $\mathcal{L}_h$  is responsible for the non-local correlation structure of the current-biased dynamics.

Although Theorem 2.5 can be proven using the relative entropy bounds of [36], proving that  $\mathbb{P}_{h,\lambda+E}^{\nu_h^N}$  is an approximate driven process for the current biased dynamics as in (2.15) requires yet finer bounds. These bounds, stated next, are the main technical ingredient. Like Theorem 2.5, they hold in the whole sub-critical regime (2.22).

**Theorem 2.7.** Let  $\lambda, E \in \mathbb{R}$  be sub-critical, i.e. satisfy (2.22). Let h be given by Proposition 2.1. For  $t \geq 0$ , let  $\mathbf{H}^{N}(f_{t})$  denote the relative entropy of the law of the dynamics  $f_{t}\nu_{h}^{N}$  with respect to  $\nu_{h}^{N}$ :

$$\mathbf{H}^{N}(f_{t}) := \nu_{h}^{N}(f_{t}\log f_{t}) = \sum_{\eta \in \Omega_{N}} f_{t}(\eta)\log f_{t}(\eta)\nu_{h}^{N}(\eta).$$
(2.37)

There are then  $\gamma, C > 0$  depending only on  $\lambda, E$  such that:

$$\forall t > 0, \qquad \partial_t \mathbf{H}^N(f_t) \le \frac{\mathbf{H}^N(f_t)}{\gamma} + \frac{C}{N^{1/2}}$$
$$\Rightarrow \quad \mathbf{H}^N(f_t) \le \frac{\gamma C e^{t/\gamma}}{N^{1/2}}, \quad t \ge 0.$$
(2.38)

Theorem 2.7 is proven in Section 3.4. Such a relative entropy estimate provides a precise control on the dynamics that can be used in other contexts. In [13], they allowed for the study of the probability of observing long time anomalous correlations in the symmetric simple exclusion process in contact with reservoirs at different densities.

#### 2.4 Heuristics and structure of the proof

Let us now sketch the proof of Theorem 2.4, postponing technical estimates to later sections.

Step 1: Construction of a candidate for the approximate driven process. To obtain a candidate approximate driven process, we carry out the procedure outlined in Section 2.2, successively modifying the jump rates to first obtain a dynamics with the right macroscopic current, then with the right two-point correlations. This is the content of Proposition 4.2, which states that, for any symmetric, translation invariant bias  $\tilde{h} \in C_D^{\infty}(\mathbb{T}^2)$ , the

current-biased dynamics can be rewritten in terms of a tilted dynamics:

$$\mathbb{P}_{\lambda,E,T}^{curr,\nu_{1/2}^{N}}(\cdot) = \frac{\mathbb{E}_{\tilde{h},\lambda+E}^{\nu_{\tilde{h}}^{N}} \left[ \mathbf{1} \cdot \exp\left(-\Pi_{T}^{N}(\tilde{h}) - \Pi_{0}^{N}(\tilde{h}) + \int_{0}^{T} \Pi_{t}^{N}(F_{\lambda,E}(\tilde{h})) \, dt + \int_{0}^{T} \varepsilon_{\tilde{h},\lambda,E}^{N}(\eta_{t}) \, dt \right) \right]}{\mathbb{E}_{\tilde{h},\lambda+E}^{\nu_{\tilde{h}}^{N}} \left[ \exp\left(-\Pi_{T}^{N}(\tilde{h}) - \Pi_{0}^{N}(\tilde{h}) + \int_{0}^{T} \Pi_{t}^{N}(F_{\lambda,E}(\tilde{h})) \, dt + \int_{0}^{T} \varepsilon_{\tilde{h},\lambda,E}^{N}(\eta_{t}) \, dt \right) \right]}$$
(2.39)

Above,  $F_{\lambda,E}$  is a differential operator while the random variable  $\varepsilon_{\tilde{h},\lambda,E}^N$  involves three-point correlations and higher that are expected to be negligible compared to two-point correlations, which one expects to be of order 1 in N.

In order for  $\mathbb{P}^{N}_{\tilde{h},\lambda+E}$  to be an approximate driven process in the sense of (2.15), we need the exponential in (2.39) to be bounded with N, T when N, then T large. This suggests to take  $\tilde{h}$  with  $F_{\lambda,E}(\tilde{h}) = 0$ , which is equivalent to  $\tilde{h}$  being the bias h of Proposition 2.1.

The error terms  $\int_0^T \varepsilon_{h,\lambda,E}^N(\eta_t) dt$ ,  $\Pi_T^N(h)$ ,  $\Pi_0^N(h)$ , however, appear inside exponentials. We check in Section 3 that they have are well-controlled in expectation under the dynamics  $\mathbb{P}_{h,\lambda+E}^{\nu_h^N}$ , using the relative entropy estimate of Theorem 2.7 that is also proven there. Still, it could be that their exponential moment blows up with N, T on a rare event. This would mean that  $\mathbb{P}_{h,\lambda+E}^{\nu_h^N}$  is in fact not an approximate driven process.

Step 2: estimate of error terms. The key Proposition 4.3 provides a control of the exponential terms in (2.39), proving that they indeed remain bounded when N, then T are large. This is achieved through a domination bound of the following form: for any sequence  $(\mathcal{O}_N)$  of measurable events that involve the dynamics up to time T at most,

$$\limsup_{N \to \infty} \mathbb{P}_{\lambda, E, T}^{curr, \nu_{1/2}^{N}}(\mathcal{O}_{N}) \leq C(\lambda, E) \limsup_{N \to \infty} \mathbb{P}_{h, \lambda + E}^{\nu_{h}^{N}}(\mathcal{O}_{N})^{1/2}.$$
(2.40)

This bound is the only claim for which we need the (technical) restriction that  $\lambda(\lambda + 2E) \ge 0$  is small enough.

Equipped with the domination bound (2.40), it is not hard to prove (see Proposition 4.4) that the current-biased dynamics has the same fluctuations at times  $1 \ll t \ll T$  when N, then T are large as the dynamics:

$$\frac{\mathbb{E}_{h,\lambda+E}^{\nu_h^N} \left[ \mathbf{1}.H_1(Y_0^N) H_2(Y_T^N) \right]}{\mathbb{E}_{h,\lambda+E}^{\nu_h^N} \left[ H_1(Y_0^N) H_2(Y_T^N) \right]}.$$
(2.41)

Above,  $H_1, H_2$  are suitable bounded functions of the fluctuation field.

Step 3: fluctuations under  $\mathbb{E}_{h,\lambda+E}^{\nu_h^N}$  and time decorrelation. Fluctuations under (2.41) are then analysed in two stages. The limiting law  $\mathbf{P}_{h,\lambda+E}^{\nu_h^\infty}$  of the fluctuation process under  $\mathbb{P}_{h,\lambda+E}^{\nu_h^N}$  is first characterised in Section 3, in particular proving Theorem 2.5. We prove that, for any bounded  $F: \mathcal{D}(\mathbb{R}_+, \mathcal{S}(\mathbb{T})) \to \mathbb{R}$  and suitable bounded functions  $H_1, H_2$ ,

$$\lim_{N \to \infty} \frac{\mathbb{E}_{h,\lambda+E}^{\nu_h^N} \left[ F\left((Y_s^N)_{s \le T}\right) H_1(Y_0^N) H_2(Y_T^N) \right]}{\mathbb{E}_{h,\lambda+E}^{\nu_h^N} \left[ H_1(Y_0^N) H_2(Y_T^N) \right]} = \frac{\mathbf{E}_{h,\lambda+E}^{\nu_h^\infty} \left[ F\left((Y_s)_{s \le T}\right) H_1(Y_0) H_2(Y_T) \right]}{\mathbf{E}_{h,\lambda+E}^{\nu_h^\infty} \left[ H_1(Y_0) H_2(Y_T) \right]}.$$
 (2.42)

We conclude the proof of Theorem 2.4 in Section 5, showing that the limiting fluctuation process decorrelates in time in the sense that, for each  $t \ge 0$  and each A > 0 with  $t + A \le T$ :

$$\frac{\mathbf{E}_{h,\lambda+E}^{\nu_{h}^{\infty}}\left[F\left((Y_{s})_{t\leq s\leq t+A}\right)H_{1}(Y_{0})H_{2}(Y_{T})\right]}{\mathbf{E}_{h,\lambda+E}\left[H_{1}(Y_{0})H_{2}(Y_{T})\right]} = \mathbf{E}_{h,\lambda+E}^{\nu_{h}^{\infty}}\left[F\left((Y_{s})_{s\leq T}\right)\right] + C_{t,T}(F,H_{1},H_{2}), \quad (2.43)$$

with:

$$\lim_{t \to \infty} \limsup_{T \to \infty} |C_{t,T}(F, H_1, H_2)| = 0.$$
(2.44)

This decorrelation in time boils down to Gaussian computations on a family of finitedimensional diffusions approximating the limiting fluctuation process, constructed as in [31].

### 2.5 Discussion of the results and perspectives

In this article, the large N density fluctuations of the current-biased dynamics at intermediate times  $1 \ll t \ll T$  are characterised. In this time regime, the current biased dynamics are shown to be well-approximated by the explicit, non-local and homogeneous Markov dynamics  $\mathbb{P}_{h,\lambda+E}^{\nu_h^N}$ . Moreover, correlations at times  $1 \ll t \ll T$  under the current-biased dynamics are shown to agree with the expression predicted in [15]. Let us discuss a few limitations and possible extensions.

**Dynamics for times around** 0, T. It is also possible to investigate the behaviour of the current-biased dynamics (2.7) at times t, T - t with t of order 1. This would require no change to the microscopic study, only work on the limiting fluctuation process. Indeed, for such times the limiting fluctuations can still be expressed in terms of  $\mathbf{P}_{h,\lambda+E}^{\nu_h}$  but the time decorrelation (2.43) does not happen, reflecting the fact that initial/final conditions still influence the dynamics. The corresponding fluctuation process therefore becomes inhomogeneous in time.

Annealed and quenched dynamics. The current-biased dynamics (1.6) is defined in an "annealed" way, in the sense that it is a ratio of quantities averaged on the initial condition. One could also ask about results on a "quenched" version of the dynamics. Several variants of quenched dynamics can be envisioned. A fully quenched version, so to speak, would be:

$$\mathbb{E}_{\lambda,E,T}^{q,\mu^{N}}(\cdot) = \sum_{\eta \in \Omega_{N}} \mu^{N}(\eta) \frac{\mathbb{E}_{E}^{\eta}[\mathbf{1}.e^{\lambda NQ_{T}}]}{\mathbb{E}_{E}^{\eta}[e^{\lambda NQ_{T}}]} \qquad \text{for an initial condition } \mu^{N}.$$
(2.45)

If e.g.  $\mu^N = \nu_{1/2}^N$ , the only missing ingredient to prove Theorem 2.4 for the dynamics (2.45) is an equivalent of the domination bound (2.40). It is unclear whether such a bound should hold. If it does, then the above quenched dynamics would have the same fluctuations as the "annealed" dynamics (2.7).

Another possibility is to average on the initial condition while keeping the number of particles fixed, as in the following dynamics started from the uniform distribution  $U_m$  on configurations with  $m \in \{0, ..., N\}$  particles:

$$\mathbb{E}_{\lambda,E,T}^{curr,U_m}(\cdot) = \frac{\mathbb{E}_E^{U_m}[\mathbf{1}.e^{\lambda NQ_T}]}{\mathbb{E}_E^{U_m}[e^{\lambda NQ_T}]}.$$
(2.46)

For this dynamics we expect Theorem 2.4 to hold without much change to the proof. This would be very interesting to prove as the initial condition  $U_m$  has an advantage over  $\nu_{1/2}^N$  as discussed in the next paragraph.

Influence of the initial condition. We now come back to the annealed current-biased dynamics (2.7) considered in the paper and discuss initial conditions more in depth.

Our main result, Theorem 2.4, is a statement about the current-biased dynamics in the intermediate time interval  $1 \ll t \ll T$ . Part of its proof consists in showing a time decorrelation result according to which the effect of the initial condition is not relevant. One therefore expects the statement to be independent of the specific choice  $\nu_{1/2}^N$  of initial condition (and for instance Theorem 2.4 also holds starting from a class of discrete Gaussian measures of the form (2.29), with the same proof).

On one aspect, however, the choice of  $\nu_{1/2}^N$  does matter. That is because this measure allows all possible number of particles, including 0 or N particles where there is no current. Since the number of particles is conserved by the dynamics, this has important consequences. Indeed, start the WASEP dynamics from the uniform measure  $U_m$  on configurations with  $m = \lfloor \rho N \rfloor$  particles ( $\rho \in (0, 1)$ ). A direct computation using [9] shows that the denominator in (4.2) is informally given to leading order in N, T by:

$$\mathbb{E}_{E}^{U_{m}}\left[e^{\lambda NQ_{T}}\right] \approx e^{NT\rho(1-\rho)\lambda(\lambda+2E)}.$$
(2.47)

As all densities are allowed under  $\nu_{1/2}^N$ , as soon as  $\lambda(\lambda+2E) < 0$  the denominator  $\mathbb{E}_E^{\nu_{1/2}^N}[e^{\lambda NQ_T}]$ in the definition (2.7) is therefore dominated by the contributions at 0, N particles when  $T \gg 1$  even though the typical density under  $\nu_{1/2}^N$  is 1/2:

$$\mathbb{E}_{E}^{\nu_{1/2}^{N}} \left[ e^{\lambda N Q_{T}} \right] = \frac{1}{2^{N}} \sum_{m=0}^{N} {N \choose m} \mathbb{E}_{E}^{U_{m}} \left[ e^{\lambda N Q_{T}} \right]$$
$$\approx \frac{1}{2^{N-1}} \left( \mathbb{E}_{E}^{U_{0}} \left[ e^{\lambda N Q_{T}} \right] + o_{T}(1) \right) = \frac{1}{2^{N-1}} \left( 1 + o_{T}(1) \right), \qquad (2.48)$$

where the  $o_T(1)$  is uniform in N. Thus for  $\lambda(\lambda + 2E) < 0$  the "annealed" current-biased dynamics (2.7) started from  $\nu_{1/2}^N$  does not capture any interesting dynamical phenomenon when N, T are large. This is the reason for the restriction  $\lambda(\lambda + 2E) \ge 0$  in Theorem 2.4. In particular, recalling the definition (2.22) of the conjectured critical line, the restriction  $\lambda(\lambda + 2E) \ge 0$  means that only  $\lambda, E$  far from the critical line can be considered.

A natural way to bypass this problem would be to consider an initial condition with a fixed number of particles as in (2.46).

Extending Theorem 2.4 to that case should require no new ideas and would allow for  $\lambda, E$  with  $\lambda(\lambda + 2E) \leq 0$ . Indeed, the structure of the proof of Theorem 2.4 for  $|\lambda(\lambda + 2E)|$  small enough would be identical: the candidate for the driven process and the time-decorrelation result of the limiting fluctuation fields are the same. The key microscopic ingredient, the relative entropy estimate, can also be set up at fixed density without change. To get bounds similar to those of Theorem 2.7, however, we would need an additional input: an equivalent

of Appendix A for the canonical measure:

$$U_{\lfloor N/2 \rfloor,h} := \nu_h^N \Big( \cdot \Big| \sum_i \eta_i = \lfloor N/2 \rfloor \Big).$$
(2.49)

In particular exponential concentration bounds for *n*-point correlations under this measure when  $\lambda(\lambda+2E) \leq 0$ , corresponding to *h* being a positive kernel (recall Proposition 2.1) would be required. We would also need to prove a central limit theorem for density fluctuations, which should be accessible, perhaps using the classical results on mean-field models of [39]. As these estimates are fairly technical and would still not allow to cover the full range of sub-critical  $\lambda, E$ , we did not pursue that generalisation.

The non-perturbative case. Let us now discuss how to remove the smallness assumption on  $\lambda(\lambda + 2E)$  in Theorem 2.4.

The requirement that  $\lambda(\lambda + 2E) \ge 0$  be small comes from a single, but central technical estimate: the domination bound (2.40). We claim that this bound can be proven for any  $\lambda, E$  with  $\lambda(\lambda + 2E) \ge 0$  provided the following estimate holds.

Write  $U_{m,h}$  for the measure  $\nu_h^N$  conditioned to having  $0 \le m \le N$  particles. Assume that, for any  $\lambda, E$  in the sub-critical regime of Proposition 2.1:

 $U_{m,h}$  satisfies a logarithmic Sobolev inequality

with constant  $c(\lambda, E)N^2$  uniform in  $m \in \{0, ..., N\}$ . (2.50)

In other words, assume that there is  $c(\lambda, E)$  independent of N and  $m \in \{0, ..., N\}$  such that, for any density  $f : \Omega_{N,m} \to \mathbb{R}_+$  for  $U_{m,h}$ :

$$U_{m,h}(f \log f) \le c(\lambda, E) N^2 U_{m,h} \Big( \sum_{i \in \mathbb{T}_N} \Big[ \sqrt{f(\eta^{i,i+1})} - \sqrt{f(\eta)} \Big]^2 \Big).$$
 (2.51)

Then Theorem 2.4 with the current-biased dynamics started from  $\nu_{1/2}^N$  holds for any  $\lambda(\lambda + 2E) \geq 0$ . Alternatively, assuming the above log-Sobolev inequality (2.51) for  $m = \lfloor N/2 \rfloor$  only as well as a CLT under the measure  $U_{\lfloor N/2 \rfloor,h}$  of (2.49), the result of Theorem 2.4 for the current-biased dynamics started from  $U_{\lfloor N/2 \rfloor}$  would similarly hold in the whole sub-critical regime (2.22).

The proof of such a logarithmic Sobolev inequality is currently under investigation.

# 3 Preliminaries: fluctuations and correlations under $\mathbb{P}_{h,\lambda+E}^N$ and relative entropy estimate

In this section we establish preliminary results needed in the proof of the main result, Theorem 2.4. We prove Theorem 2.5 characterising fluctuations and correlations under the approximate driven process  $\mathbb{P}_{h,\lambda+E}^N$  (Sections 3.1 to 3.3) and the relative entropy estimate of Theorem 2.7 (Section 3.4).

Throughout, we consider the following more general case: rather than  $\mathbb{P}^{N}_{h,\lambda+E}$ , we study fluctuation and correlation processes for a dynamics of the form  $\mathbb{P}^{N}_{\tilde{h},\tilde{\lambda}}$ , where  $\tilde{\lambda} \in \mathbb{R}$  and

 $\tilde{h} \in C_D^{\infty}(\mathbb{T}^2)$  is symmetric (this set is defined in (2.14)). We assume throughout that  $\sigma \tilde{h}$  has leading eigenvalue strictly below 1: for some parameter  $\delta \in (0, 1)$ :

$$\sup_{f \in \mathbb{L}^2(\mathbb{T}): \|f\|_2 = 1} \int_{\mathbb{T}^2} f(x) \sigma \tilde{h}(x, y) f(y) \, dx \, dy \le (1 - \delta), \qquad \sigma := (1/2)(1 - 1/2) = 1/4.$$
(3.1)

**Remark 3.1.** In the case  $\tilde{h} = h$  with  $\tilde{\lambda} = \lambda + E$ , the assumption (3.1) covers the full subcritical regime, recall Proposition 2.1. In particular and as mentioned above, no limitation on  $\lambda$ , E come from Theorems 2.5–2.7.

The dynamics  $\mathbb{P}^{N}_{\tilde{h},\tilde{\lambda}}$  is started from the discrete Gaussian measure  $\nu^{N}_{\tilde{h}}$  defined as in (2.29) with  $\tilde{h}$  instead of h there. The fluctuations are studied as follows.

- Limit theorems for the law of the fluctuation process  $Y_{\cdot}^{N}$  under a local dynamics ( $\tilde{h} = 0$ ) have been obtained in [36] using  $O_{N}(1)$  estimates on the relative entropy of the law of the dynamics at each time compared to a suitable product measure. The extension to  $\tilde{h} \neq 0$  does not require new arguments provided a similar relative entropy estimate holds. We therefore only provide a sketch of proof, in Section 3.1 and refer to [36] for details.
- The correlation process  $\Pi^N$  is then expressed in terms of  $Y^N$  when acting on smooth test functions. This and an approximation argument characterise its limit uniquely when acting also on test functions in  $C^{\infty}_D(\mathbb{T}^2)$  (recall (2.14)) that are smooth up to singularities on the diagonal. Stationarity of the limiting processes  $Y_{\cdot}, \Pi_{\cdot}$  follows from an explicit formula for the law of  $Y_t$  ( $t \geq 0$ ).

The following proposition contains the entropy estimate, and includes Theorem 2.7 as a special case. The  $o_N(1)$  bound stated below is stronger than required for the proof of Theorem 2.5. Note that the law of the dynamics is compared to the correlated measure  $\nu_{\tilde{h}}^N$  rather than a product measure as in [36]. This is essential for the  $o_N(1)$  bound and does not change induce any change to the proof of Theorem 2.5 (where an  $O_N(1)$  bound is enough) compared to the proof in [36].

**Proposition 3.2.** Let  $f_t \nu_{\tilde{h}}^N$  denote the law of the dynamics  $\mathbb{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^N}$  at time  $t \geq 0$ . There are  $\gamma, C > 0$  independent of  $\tilde{h}$  such that, for each T > 0:

$$\forall N \in \mathbb{N}_{\geq 1}, \qquad \sup_{t \leq T} \mathbf{H}^N(f_t) = \sup_{t \leq T} \leq \frac{\gamma C e^{T/\gamma}}{N^{1/2}}, \qquad \mathbf{H}^N(f_t) := \nu_{\tilde{h}}^N(f_t \log f_t). \tag{3.2}$$

This bound implies that, for each bounded  $\phi : \mathbb{T}^2 \to \mathbb{R}$ :

$$\sup_{N} \sup_{t \le T} \mathbb{E}_{\tilde{h}, \tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \left[ \left| \Pi_{t}^{N}(\phi) \right|^{3/2} \right] \le C(T, \phi).$$

$$(3.3)$$

Moreover, Pinsker's inequalities yields the following useful consequence. For any function  $F_N: \Omega_N \to \mathbb{R}$  and any  $T \ge 0$ ,

$$\sup_{t \le T} \left| \mathbb{E}_{\tilde{h}, \tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \left[ F_{N}(\eta(t)) \right] - \nu_{\tilde{h}}^{N}(F_{N}) \right| \le \|F_{N}\|_{\infty} \sup_{t \le T} \int |f_{t} - 1| \, d\nu_{\tilde{h}}^{N} \le \frac{\|F_{N}\|_{\infty} \sqrt{2\gamma C e^{T/\gamma}}}{N^{1/4}}.$$
(3.4)

The proof of Proposition 3.2 is quite technical, and therefore postponed to Section 3.4 where we also discuss practical applications of this bound.

### 3.1 The limiting fluctuation process

Here we sketch the proof of item 3 in Theorem 2.5 and characterise the law of the fluctuations at time 0 (part of item 1). The proof of item 3 is standard: we prove tightness of  $(Y_t^N)_{t\geq 0}$  in  $\mathcal{D}([0,\infty), \mathcal{S}'(\mathbb{T}))$ , then prove that limit points satisfy a martingale problem that has a unique solution.

#### 3.1.1 Tightness

By Mitoma's criterion [43, Theorem 3.1] (formulated for trajectories on bounded time intervals, but immediately extending to unbounded intervals as in [12, Theorem 16.4]), tightness of  $(Y_t^N)_{t\geq 0}$  in  $\mathcal{D}([0,\infty), \mathcal{S}'(\mathbb{T}))$  follows from tightness of all projections  $(Y_t^N(f))_{t\geq 0}$  in  $\mathcal{D}([0,\infty), \mathbb{R})$  ( $f \in C^{\infty}(\mathbb{T})$ ). By [12, Theorems 16.8–16.10], tightness of  $(Y_t^N(f))_{t\geq 0}$  follows from the following two estimates:

(i) For each  $t \ge 0$ ,

$$\lim_{a \to \infty} \limsup_{N \to \infty} \mathbb{P}_{\tilde{h}, \tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \left( \left| Y_{t}^{N}(f) \right| \ge a \right) = 0.$$
(3.5)

(ii) (Aldous criterion) For  $t \ge 0$ , let  $\mathcal{I}_t$  denote the set of stopping times for  $(Y_s^N(f))_{s\ge 0}$ bounded by  $t \ge 0$ . For each  $t \ge 0$  and  $\varepsilon > 0$ ,

$$\lim_{\delta \to 0} \sup_{\tau \in \mathcal{I}_t} \limsup_{N \to \infty} \mathbb{P}_{\tilde{h}, \tilde{\lambda}}^{\nu_{\tilde{h}}^N} \Big( \big| Y_{\tau+\delta}^N(f) - Y_{\tau}^N(f) \big| \ge \varepsilon \Big) = 0.$$
(3.6)

Both items follow from  $O_N(1)$  bounds on the relative entropy  $\mathbf{H}^N(f_t)$  uniform on compact time intervals as we now show. Recall the entropy inequality:

$$\mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}}[F(\eta_{t})] \leq \frac{\mathbf{H}^{N}(f_{t})}{\gamma} + \frac{1}{\gamma} \log \nu_{\tilde{h}}^{N} \Big[ e^{\gamma F} \Big], \qquad \gamma > 0, F : \Omega_{N} \to \mathbb{R}.$$
(3.7)

Let  $t \ge 0$  and a > 0. For point (i), Markov- and the entropy inequalities give:

$$\mathbb{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}}\Big(\big|Y_{t}^{N}(f)\big| \ge a\Big) \le \frac{1}{a} \mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}}\Big[\big|Y_{t}^{N}(f)\big|\Big] \le \frac{\mathbf{H}^{N}(f_{t})}{a} + \frac{1}{a}\log\nu_{\tilde{h}}^{N}\Big[\exp\big|Y^{N}(f)\big|\Big].$$
(3.8)

Proposition A.4 shows that the last exponential moment is bounded uniformly in N, which yields the claim as the entropy  $\mathbf{H}^{N}(f_{t})$  is also bounded.

For point (ii), the starting point is the semi-martingale decomposition:

$$Y_t^N(f) = Y_0^N(f) + \int_0^T N^2 L_{\tilde{h},\tilde{\lambda}} Y_s^N(f) \, ds + M_t^N(f), \tag{3.9}$$

with  $(M_t^N(f))_{t\geq 0}$  a martingale independent from  $Y_0^N$  and starting at 0. Let  $t \geq 0$ ,  $\varepsilon, \delta > 0$ and  $\tau \in \mathcal{I}_t$  denote a stopping time. Then Markov property gives, recalling that  $f_\tau \nu_{\tilde{h}}^N$  stands for the law of the dynamics at time  $\tau$ :

$$\mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \Big[ \left| Y_{\tau+\delta}^{N}(f) - Y_{\tau}^{N}(f) \right| \Big] = \mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \Big[ \mathbb{E}_{\tilde{h},\tilde{\lambda}}^{f_{\tau}\nu_{\tilde{h}}^{N}} \Big[ \left| Y_{\delta}^{N}(f) - Y_{0}^{N}(f) \right| \Big] \Big] \\
\leq \mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \Big[ \mathbb{E}_{\tilde{h},\tilde{\lambda}}^{f_{\tau}\nu_{\tilde{h}}^{N}} \Big[ \left| \int_{0}^{\delta} N^{2}L_{\tilde{h},\tilde{\lambda}}Y_{s}^{N}(f) \, ds \right| \Big] + \mathbb{E}_{\tilde{h},\tilde{\lambda}}^{f_{\tau}\nu_{\tilde{h}}^{N}} \Big[ \left| M_{\delta}^{N}(f) \right| \Big] \Big]. \quad (3.10)$$

Note that, by Cauchy-Schwarz inequality, the estimate of the martingale term boils down to an estimate on its quadratic variation:

$$\left\langle M_{\delta}^{N}(f)\right\rangle = \int_{0}^{\delta} N^{2} \sum_{i\in\mathbb{T}_{N}} c_{\tilde{h},\tilde{\lambda}}(\eta_{t'},\eta_{t'}^{i,i+1}) \Big[\Pi^{N}(f)(\eta_{t'}^{i,i+1}) - \Pi^{N}(f)(\eta_{t'})\Big]^{2} dt'.$$
(3.11)

Using the entropy inequality as in (3.8) and noticing that  $\tau \leq t$  and that the relative entropy is bounded with N uniformly on  $[0, t + \delta]$ , item (ii) is therefore proven if we can prove that the integrands in (3.10)–(3.11) have bounded exponential moment under  $\nu_{\tilde{h}}^{N}$ . This is indeed the case. As computations similar to that of the action of the generator on  $Y^{N}(f)$  and to (3.11) are carried out to prove Proposition 3.2, we omit them here and conclude the proof of tightness.

#### 3.1.2 Martingale problem

We now sketch the proof that limit points of  $(\mathbb{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}})$  satisfy the martingale problem of Theorem 2.5.

The key argument is the following:

$$Y_t^N(f_t) = Y_0^N(f_0) + \int_0^t Y_s^N(\partial_s f_s + \mathcal{L}_{\tilde{h}} f_s) \, ds + M_t^N(f) + \int_0^t \varepsilon_s^N(f) \, ds, \qquad (3.12)$$

where  $f \in C^1(\mathbb{R}_+, C^{\infty}(\mathbb{T}))$ ,  $\mathcal{L}_{\tilde{h}}$  is the operator appearing in Theorem 2.5 and  $\varepsilon^N(f)$  is an error term in the sense:

$$\lim_{N \to \infty} \mathbb{E}_{\tilde{h}, \tilde{\lambda}}^{\tilde{\nu}_{h}^{N}} \left[ \left| \int_{0}^{t} \varepsilon_{s}^{N}(f) \, ds \right| \right] = 0.$$
(3.13)

The above estimate is an instance of the so-called Boltzmann-Gibbs principle. The expression of  $\varepsilon^N(f)$  is obtained through computations similar to those carried out in the proof of Proposition 3.2, so we give no more detail. Estimates such as (3.13) are proven in Corollary 3.7 below, see also [36] for a different presentation.

Equation (3.12) defines a limiting process  $(M_t(f))_{t\geq 0}$  for each limit point  $(Y_t)_{t\geq 0}$  of  $(Y_t^N)_{t\geq 0}$ . This process is  $(Y_s)_{s\leq t}$ -measurable, independent from  $Y_0$  and starts at 0 by construction. It is also continuous as  $(Y_t)_{t\geq 0}$  itself is a continuous process due to  $(Y_t^N(f))_{t\geq 0}$  having jumps of amplitude bounded by  $||f||_{\infty}N^{-1/2}$ .

The continuous process  $(\mathfrak{M}_t(f))_{t\geq 0} = (M_t(f)^2 - \int_0^t \|\partial_x f_s\|_2 ds)_{t\geq 0}$  can be constructed similarly. To show that  $(M_t(f))_{t\geq 0}, (\mathfrak{M}_t(f))_{t\geq 0}$  are martingales, it is enough to show that  $(M_t^N(f)^{2+\varepsilon})_{N\in\mathbb{N}_{\geq 1}}$  is uniformly integrable for some  $\varepsilon > 0$  [35, Propositions IX.1.4 and IX.1.12]. This again follows from explicit computations of the quadratic variation of  $M_t^N(f)$  and  $O_N(1)$ relative entropy bounds, see [36, (6.7) and Corollary 2.3] for a similar argument.

The fact that  $(M_t(f))_{t\geq 0}, (\mathfrak{M}_t(f))_{t\geq 0}$  are martingales characterises  $(M_t(f))_{t\geq 0}$  by a result of Dubins-Schwarz: for each  $t\geq 0$ ,

$$M_t(f) = B^f_{2\sigma \int_0^t \|\partial_x f_s\|_2^2 ds}, \qquad (B^f_t)_{t \ge 0} \text{ a standard Brownian motion.}$$
(3.14)

For  $t \geq 0$ , taking as test function  $f_{s,t} = e^{(t-s)\mathcal{L}_{\tilde{h}}} f \in C^1([0,t], C^{\infty}(\mathbb{T}))$  for  $f \in C^{\infty}(\mathbb{T})$  (the regularity is proven in Appendix C), we find that any limit point  $(Y_t)_{t\geq 0}$  of the  $(Y_t^N)_{t\geq 0}$  satisfies:

$$Y_t(f) = Y_0(f_t) + B^{f_{\cdot,t}}_{2\sigma\int_0^t \|\partial_x f_{s,t}\|_2^2 \, ds}.$$
(3.15)

It follows that the law of  $(Y_t)_{t\geq 0}$  is uniquely determined as soon as the law of the initial condition  $Y_0^N$  converges, which we check next.

#### 3.1.3 Initial condition

**Proposition 3.3** (Law at the initial time). The weak limit  $\nu_{\tilde{h}}^{\infty}$  of the law of  $Y_0^N$  under  $\mathbb{P}_{\tilde{h}}^{\nu_{\tilde{h}}^N}$  is a centred Gaussian field on  $\mathcal{S}'(\mathbb{T})$ , with covariance given for  $f_1, f_2 \in C^{\infty}(\mathbb{T})$  by:

$$\mathbb{E}_{\nu_{\tilde{h}}^{\infty}}\Big[Y_0(f_1)Y_0(f_2)\Big] = \int_{\mathbb{T}} f_1(x) \big(C_{\tilde{h}}f_2\big)(x)dx, \quad C_{\tilde{h}} := \big(\sigma^{-1}id - \tilde{h}\big)^{-1}.$$
(3.16)

Above, the function  $\tilde{h}$  is identified with the kernel operator  $\tilde{h}f_2(\cdot) = \int_{\mathbb{T}} \tilde{h}(\cdot, y) f_2(y) dy$ .

*Proof.* Let us prove that the law  $\nu_{\tilde{h}}^N \circ (Y^N)^{-1}$  of  $Y_0^N$  under  $\mathbb{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^N}$  converges to a Gaussian field with covariance  $C_{\tilde{h}}$ . To do so, we compute moments of all order. Let  $n \in \mathbb{N}$  and  $f^1, \ldots, f^{2n+1} \in C^{\infty}(\mathbb{T})$ . It is an immediate consequence of the invariance of  $\nu_{\tilde{h}}$  under the mapping  $\eta \mapsto (1 - \eta_i)_{i \in \mathbb{T}_N}$  that odd moments of  $Y^N$  vanish:

$$\nu_{\bar{h}}^{N} \left[ \prod_{i=1}^{2n+1} Y^{N}(f^{i}) \right] = 0.$$
(3.17)

Above and in the following, we use  $\nu_{\tilde{h}}^{N}(F)$  or  $\nu_{\tilde{h}}^{N}[F]$  to denote expectation of  $F : \Omega_{N} \to \mathbb{R}$ under  $\nu_{\tilde{h}}^{N}$ . Proposition A.2 provides the following estimate of even moments. Let  $\tilde{h}^{N}$  denote the matrix:

$$\tilde{h}^{N}(i,j) := \tilde{h}_{i,j} = \tilde{h}\left(\frac{i}{N}, \frac{j}{N}\right), \qquad \tilde{h}^{N}(i,i) = 0, \qquad i \neq j \in \mathbb{T}_{N}.$$
(3.18)

The following approximate Wick formula then holds: there is  $e_N^n = O(N^{-n-1})$  such that, with  $P_{2n}$  the set of pairings of 2n elements and  $I_N$  the  $N \times N$  identity matrix:

$$\sup_{J \subset \Lambda_N: |J|=2n} \left| \nu_{\tilde{h}}^N \Big( \prod_{j \in J} \bar{\eta}_j \Big) - \sum_{p \in P_{2n}} \prod_{j=1}^n \left( \sigma^{-1} I_N - N^{-1} \tilde{h}^N \right)^{-1} (j, p(j)) \right| \le e_N^n.$$
(3.19)

The inverse matrix above is well defined for large enough N, see the discussion around (A.31). Using (3.19) for each set J with even cardinal  $|J| \leq 2n$  as well as:

$$\bar{\eta}_i^2 = \sigma \quad (i \in \mathbb{T}_N), \qquad \sup_{j \in \mathbb{T}_N} \left| (\sigma^{-1} I_N - N^{-1} \tilde{h}^N)^{-1} (j, j) - \sigma \right| = O(N^{-1}),$$
(3.20)

we find:

$$\nu_{\tilde{h}}^{N} \Big[ \prod_{i=1}^{2n} Y^{N}(f^{i}) \Big] = \frac{1}{N^{n}} \sum_{j_{1},...,j_{2n} \in \mathbb{T}_{N}} \nu_{\tilde{h}}^{N} \Big[ \prod_{i=1}^{2n} \bar{\eta}_{j_{i}} f_{j_{i}}^{i} \Big]$$

$$= \sum_{p \in P_{2n}} \prod_{i=1}^{n} \left( \frac{1}{N} \sum_{a,b \in \mathbb{T}_{N}} f_{a}^{i} f_{b}^{p(i)} (\sigma^{-1} I_{N} - N^{-1} \tilde{h}^{N})^{-1}(a,b) \right) + e'_{N}(f^{1},...,f^{p}),$$
(3.21)

where  $e'_N(f_1, ..., f_p) = O(N^{-1})$ . To conclude, it remains to prove that, for any  $f^1, f^2 \in C^{\infty}(\mathbb{T})$ :

$$\lim_{N \to \infty} \frac{1}{N} \sum_{a,b \in \mathbb{T}_N} f_a^1 f_b^2 \left( \sigma^{-1} I_N - N^{-1} \tilde{h}^N \right)^{-1} (a,b) = \int_{\mathbb{T}} f^1(x) C_{\tilde{h}} f^2(x) \, dx. \tag{3.22}$$

Let  $f \in C^{\infty}(\mathbb{T})$  and, for an  $N \times N$  matrix M define, define for each  $i \in \mathbb{T}_N$ :

$$Mf(i) := \sum_{j \in \mathbb{T}_N} M(i,j) f_j.$$
(3.23)

From  $\tilde{h} \in C_D^{\infty}(\mathbb{T}^2) \subset C^0(\mathbb{T}^2)$  we get:

$$\sup_{x\in\mathbb{T}} \left| (\sigma^{-1}\mathrm{id} - \tilde{h}) f(x) - (\sigma^{-1}I_N - N^{-1}\tilde{h}^N) f(\lfloor x/N \rfloor/N) \right| \xrightarrow[N \to \infty]{} 0.$$
(3.24)

In addition, we prove in Lemma A.3 that  $(\sigma^{-1}I_N - N^{-1}\tilde{h}^N)^{-1}$  satisfies, for some c > 0:

$$\sup_{N \ge 1} \max_{i \in \mathbb{T}_N} \left| (\sigma^{-1} I_N - N^{-1} \tilde{h}^N)^{-1}(i, i) \right| \le c, \qquad \sup_{N \ge 1} \max_{i \ne j \in \mathbb{T}_N} \left| (\sigma^{-1} I_N - N^{-1} \tilde{h}^N)^{-1}(i, j) \right| \le \frac{c}{N}.$$
(3.25)

Using these bounds with (3.24) applied to  $C_{\tilde{h}}f$ , identified with the vector  $((C_{\tilde{h}}f)_i)_{i\in\mathbb{T}_N}$ , we find that the vector:

$$(\sigma^{-1}I_N - N^{-1}\tilde{h}^N)^{-1} - C_{\tilde{h}} = (\sigma^{-1}I_N - N^{-1}\tilde{h}^N)^{-1} \Big[ f - (\sigma^{-1}I_N - N^{-1}\tilde{h}^N)C_{\tilde{h}}f \Big]$$
(3.26)

has largest entry in absolute value that vanishes with N. This implies (3.22) and concludes the proof.  $\Box$ 

Putting together the results of Sections 3.1.1 to 3.1.3 proves item 3 and part of item 1 of Theorem 2.5. In the next section, we move on to characterise limiting correlations, proving items 2, 4 and 5. The stationarity claim of item 1 is proven in Section 3.3.

### 3.2 The limiting correlation process

Here, we prove that the correlation process  $\Pi^N_{\cdot} \in \mathcal{D}([0, +\infty), \mathcal{S}'_D(\mathbb{T}^2))$  under  $\mathbb{P}_{\tilde{h}, \tilde{\lambda}}^{\nu_{\tilde{h}}^N}$  has a unique weak limit, as written out in Theorem 2.5, and characterise its properties. This is first done by expressing  $\Pi_{\cdot}$  in terms of  $Y_{\cdot}$  for smooth test functions. An approximation argument then gives a characterisation of the correlation process also including test functions that are not smooth on the diagonal. Let  $(e_{\ell})_{\ell \geq 0}$  be an orthonormal basis of eigenvectors of the Laplacian on the torus:  $e_0 = 1$ , and:

$$\forall \ell \in \mathbb{N}_{\geq 1}, \forall x \in \mathbb{T}, \qquad e_{2\ell-1}(x) = \sqrt{2}\sin(2\pi\ell x), \quad e_{2\ell}(x) = \sqrt{2}\cos(2\pi\ell x). \tag{3.27}$$

**Proposition 3.4.** 1. (Smooth test functions). Consider  $\Pi_{\cdot}^{N}$  as an element of the set  $\mathcal{D}([0, +\infty), \mathcal{S}'(\mathbb{T}^{2}))$  of distributions tested against smooth test functions only. Then the couple  $(Y_{\cdot}^{N}, \Pi_{\cdot}^{N})$  converges in law to a measure  $\mathbf{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{\infty}}$  on  $\mathcal{D}([0, +\infty), \mathcal{S}'(\mathbb{T}) \times \mathcal{S}'(\mathbb{T}^{2}))$ . Under this measure, for each  $(f_{1}, f_{2}) \in C^{\infty}(\mathbb{T})^{2}$ , with probability 1:

$$\Pi_t(f_1 \otimes f_2) = \frac{1}{4} Y_t(f_1) Y_t(f_2) - \frac{\sigma}{4} \int_{\mathbb{T}} f_1(x) f_2(x) \, dx, \qquad t \ge 0. \tag{3.28}$$

2. (Non-smooth test functions). For each  $t \ge 0$ ,  $\Pi_t$  admits a unique continuous extension to  $\mathcal{S}'_D(\mathbb{T}^2)$  and each vector  $(\Pi^N_{t_1}(\phi_1), ..., \Pi^N_{t_p}(\phi_p))$  converges in law to  $(\Pi_{t_1}(\phi_1), ..., \Pi_{t_p}(\phi_p))$ for  $p \in \mathbb{N}$  and  $t_1, ..., t_p \ge 0$ ,  $\phi_1, ..., \phi_p \in C^{\infty}_D(\mathbb{T}^2)$ . Moreover, there is  $C(\tilde{h}, \tilde{\lambda}) > 0$  such that, for each  $\varepsilon > 0$ :

$$\mathbf{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{\infty}}\left(|\Pi_{t}(\phi_{1})| > \varepsilon\right) \leq \frac{C(\tilde{h},\tilde{\lambda})\|\phi_{1}\|_{\infty}}{\varepsilon}, \qquad t \geq 0.$$
(3.29)

*Proof.* 1. Take  $f_1, f_2 \in C^{\infty}(\mathbb{T})$  and  $N \in \mathbb{N}_{\geq 1}$ . From the definition of  $\Pi^N$  in (2.13), one has with probability one:

$$\forall t \ge 0, \qquad \Pi_t^N(f_1 \otimes f_2) = \frac{1}{4} Y_t^N(f_1) Y_t^N(f_2) - \frac{\sigma}{4N} \sum_{i \in \mathbb{T}_N} (f_1)_i (f_2)_i. \tag{3.30}$$

In view of the characterisation of tightness of a process in Section 3.1.1 and since all test functions in  $C^{\infty}(\mathbb{T}^2)$  can be approximated uniformly by linear combinations of tensor products of functions of one variable, it follows that the sequence of laws of  $(Y_{\cdot}^{N}, \Pi_{\cdot}^{N})$  is tight as probability measures on  $\mathcal{D}([0, +\infty), \mathcal{S}'(\mathbb{T}) \times \mathcal{S}'(\mathbb{T}^2))$ . As the sequence of laws of  $Y_{\cdot}^{N}$  converges and as  $\Pi_{t}^{N}(f_1 \otimes f_2)$  is a continuous function of  $Y_{t}^{N}(f_1), Y_{t}^{N}(f_2)$  for each  $t \geq 0$ , the sequence of laws of  $(Y_{\cdot}^{N}, \Pi_{\cdot}^{N})$  converges to a measure  $\mathbf{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}}$ . The validity of (3.30) for each N and the fact that the second term in the right-hand side of (3.30) is a (non-random) Riemann sum for  $\sigma \int_{\mathbb{T}} f_1(x) f_2(x) dx$  conclude the proof of 1.

2. Let  $t \ge 0$ . As  $\Pi_t$  is a bounded linear form, it is a uniformly continuous function on the complete metric space  $C^{\infty}(\mathbb{T}^2)$  that is dense in  $C^0(\mathbb{T}^2) \supset C_D^{\infty}(\mathbb{T}^2)$  for the topology of uniform convergence. It therefore admits a unique bounded extension on  $C^0(\mathbb{T}^2)' \supset \mathcal{S}'_D(\mathbb{T}^2)$ . Fix now  $p \in \mathbb{N}$ ,  $t_1, ..., t_p \ge 0$ ,  $\phi_1, ..., \phi_p \in C_D^{\infty}(\mathbb{T}^2)$  and sequences  $(\phi_i^n)_n \in C^{\infty}(\mathbb{T}^2)^{\mathbb{N}}$  converging uniformly to  $\phi_i$   $(1 \le i \le p)$ . We do the proof for p = 2, the general case being the same. Let  $F : \mathbb{R}^2 \to \mathbb{R}$  be a bounded Lipschitz function and write:

$$F(\Pi_{t_1}(\phi_1), \Pi_{t_2}(\phi_2)) - F(\Pi_{t_1}^N(\phi_1), \Pi_{t_2}^N(\phi_2))$$
  
=  $F(\Pi_{t_1}(\phi_1), \Pi_{t_2}(\phi_2)) - F(\Pi_{t_1}(\phi_1^n), \Pi_{t_2}(\phi_2^n))$   
+  $F(\Pi_{t_1}(\phi_1^n), \Pi_{t_2}(\phi_2^n)) - F(\Pi_{t_1}^N(\phi_1^n), \Pi_{t_2}^N(\phi_2^n))$   
+  $F(\Pi_{t_1}^N(\phi_1^n), \Pi_{t_2}^N(\phi_2^n)) - F(\Pi_{t_1}^N(\phi_1), \Pi_{t_2}^N(\phi_2)).$  (3.31)

The middle line vanishes with N at fixed n when integrating on the laws of  $\Pi_{.}, \Pi_{.}^{N}$  by weak convergence for smooth test functions. The average of the first line under  $\mathbf{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{\infty}}$  vanishes with

*n* by definition of the extension of  $\Pi$  and the dominated convergence theorem. Lastly, the average of the third line vanishes with *n* uniformly in *N*. Indeed, as *F* is Lipschitz (say with respect to the norm  $||(x, y)||_1 = |x| + |y|$ ), it is enough to control times  $t_1, t_2$  separately. E.g. for  $t_1$ , the entropy inequality gives, for each  $\gamma > 0$ :

$$\mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \Big[ \big| \Pi_{t_{1}}^{N}(\phi_{1}^{n}-\phi_{1}) \big| \Big] \leq \frac{\|\phi_{1}^{n}-\phi_{1}\|_{\infty}}{\gamma} \Big[ \mathbf{H}^{N}(f_{t_{1}}) + \log \nu_{\tilde{h}}^{N} \Big( \exp \Big| \frac{\gamma \Pi^{N}(\phi_{1}^{n}-\phi_{1})}{\|\phi_{1}^{n}-\phi_{1}\|_{\infty}} \Big| \Big) \Big].$$
(3.32)

The entropy is bounded uniformly in N by Proposition 3.2 and Proposition A.4 yields the existence of  $\gamma, C > 0$  independent of  $N, \phi_1^n, \phi_1$  such that the exponential moment is bounded by C. This concludes the proof of 2. except for (3.29) that we now prove. Let  $t \ge 0$  and  $\varepsilon > 0$ . Since  $(\prod_t^N(\phi_1))_N$  converges weakly to  $\prod_t(\phi_1)$ ,

$$\mathbf{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{\infty}}(|\Pi_{t}(\phi_{1})| > \varepsilon) \leq \liminf_{N \to \infty} \mathbb{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}}(|\Pi_{t}^{N}(\phi_{1})| > \varepsilon).$$
(3.33)

Markov- and the entropy inequality give the claim as in (3.32).

### 3.3 Stationarity

The last unproven claim of Theorem 2.5 is the stationarity of the processes  $(Y_t)_{t\geq 0}$ ,  $(\Pi_t)_{t\geq 0}$ .

In view of Proposition 3.4, it is enough to prove that  $(Y_t)_{t\geq 0}$  is stationary, i.e. that the Gaussian field  $\nu_{\tilde{h}}^{\infty}$  on  $\mathcal{S}'(\mathbb{T})$  with covariance  $C_{\tilde{h}} = (\sigma^{-1}\mathrm{id} - \tilde{h})^{-1}$  is invariant. It is enough to check that the law of  $(Y_t(f_1), ..., Y_t(f_n))$   $(t \geq 0)$  does not depend on time for each  $f_1, ..., f_n \in C^{\infty}(\mathbb{T})$  and  $n \in \mathbb{N}_{\geq 1}$ . By Lévy's characteristic function theorem and the linearity of  $Y_t$ , establishing the n = 1 case is sufficient.

Fix  $f \in C^{\infty}(\mathbb{T})$ . Equations (3.14)–(3.15), the fact that  $Y_0$  is a centred Gaussian field with covariance  $C_{\tilde{h}} := (\sigma^{-1} \mathrm{id} - h)^{-1}$  and the independence of  $Y_0$  and  $M_t(f)$  imply that  $Y_t(f)$  is a centred Gaussian random variable with variance:

$$\mathbf{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{\infty}} \left[ Y_t(f)^2 \right] = \int_{\mathbb{T}} f_t(x) (C_{\tilde{h}} f_t)(x) \, dx + 2\sigma \int_0^t \|\partial_x f_s\|_2^2 \, ds.$$
(3.34)

Let us prove that this variance is just  $\int_{\mathbb{T}} f(x)(C_{\tilde{h}}f)(x) dx$ , which will suffice. Recall the expression (2.33) of  $\mathcal{L}_{\tilde{h}}$  and let  $t \mapsto f_t = e^{t\mathcal{L}_{\tilde{h}}} f \in C^{\infty}(\mathbb{R}_+ \times \mathbb{T})$  (see Appendix C for the regularity). Then:

$$\partial_t f_t = \mathcal{L}_{\tilde{h}} f_t = (\mathrm{id} - \sigma \tilde{h}) \Delta f_t = \sigma C_{\tilde{h}}^{-1} \Delta f_t, \qquad (3.35)$$

where we used  $C_{\tilde{h}} = \sigma(\mathrm{id} - \sigma \tilde{h})^{-1}$  in the last line. Integrating the second term in the righthand side of (3.34) by parts and writing the first term as its t = 0 value plus the integral of its time derivative, we find that the variance of  $Y_t(f)$  is indeed  $\int_{\mathbb{T}} f(x)(C_{\tilde{h}}f)(x) dx$ , concluding the proof:

$$\mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{\infty}} \left[ Y_t(f)^2 \right] = \int_{\mathbb{T}} f(x) (C_{\tilde{h}} f)(x) \, dx + 2 \int_0^t \int_{\mathbb{T}} \left[ f_s C_{\tilde{h}} \partial_s f_s - \sigma f_s \Delta f_s \right](x) \, dx \, ds$$
$$= \int_{\mathbb{T}} f(x) (C_{\tilde{h}} f)(x) \, dx. \tag{3.36}$$

#### 3.4 The relative entropy estimate

In this section, we prove the  $o_N(1)$  relative entropy estimate of Proposition 3.2, which includes Theorem 2.7 as a special case.

The main idea is the following: the more information we put on the measure approximating the law of the dynamics  $\mathbb{P}^{N}_{\tilde{h},\tilde{\lambda}}$  at each time, the better the resulting estimate. To obtain information on the two-point correlation structure of the law of the dynamics, one needs to tune the density profile as well as two-point correlations of the approximation measure. One way to do so is to compare the dynamics with a measure of the form  $\nu_g^N$  defined as in (2.29) for a symmetric  $g \in C_D^{\infty}(\mathbb{T}^2)$ , and optimise on g to get the correct correlation structure of the dynamics.

For the dynamics we study here the optimal choice  $g = \tilde{h}$ . This is not true in general and only valid here due to working at density 1/2. We choose to nonetheless work with a general  $g \in C_D^{\infty}(\mathbb{T}^2)$  in the following. This is to showcase that the optimal g is in general the solution of a partial differential equation involving  $\tilde{h}$  (see [13]). Moreover, the computations for general g are very similar to those needed later in Section 4.1.

**Notations:** A function  $g \in C_D^{\infty}(\mathbb{T}^2)$  is henceforth fixed such that  $\sigma g$  has leading eigenvalue eigenvalue strictly below 1. We write  $\nu_g^N$  for the associated measure as in (2.29). If f is a density for  $\nu_g^N$ , the relative entropy of  $f\nu_g^N$  with respect to  $\nu_g^N$  is denoted by:

$$\mathbf{H}_{g}^{N}(f) := \nu_{g}^{N}(f\log f), \tag{3.37}$$

where  $\nu_g^N(\cdot)$  (or  $\nu_g^N[\cdot]$ ) denotes expectation under  $\nu_g^N$ . In the special case  $g = \tilde{h}$  we simply write  $\mathbf{H}_{\tilde{h}}^N(f) = \mathbf{H}^N(f)$ .

To prove Proposition 3.2, the following lemma is the key starting point.

**Lemma 3.5** (Lemmas A.1-A.2 in [36]). For  $t \ge 0$ , recall that  $\mathbf{H}_g^N(f_t) = \nu_g^N(f_t \log f_t)$  denotes the relative entropy of the law of the dynamics  $f_t \nu_g^N$  at time t with respect to  $\nu_g^N$ . Then:

$$\partial_t \mathbf{H}_g^N(f_t) \le -N^2 \nu_g^N \big( \Gamma_{\tilde{h}, \tilde{\lambda}}(\sqrt{f_t}) \big) + N^2 \nu_g^N \big( f_t L_{g, \tilde{h}, \tilde{\lambda}}^* \mathbf{1} \big).$$
(3.38)

Above,  $\Gamma_{\tilde{h},\tilde{\lambda}}$  is the carre du champ associated with the dynamics (recall the definition (2.17)–(2.18) of the jump rates): for each function  $F: \Omega_N \to \mathbb{R}$ ,

$$\Gamma_{\tilde{h},\tilde{\lambda}}(F)(\eta) := \sum_{i \in \mathbb{R}_N} c_{\tilde{h},\tilde{\lambda}}(\eta,\eta^{i,i+1}) \left[ F(\eta^{i,i+1}) - F(\eta) \right]^2.$$
(3.39)

The operator  $N^2 L^*_{g,\tilde{h},\tilde{\lambda}}$  is the adjoint of the generator  $N^2 L_{\tilde{h},\tilde{\lambda}}$  of the dynamics in  $\mathbb{L}^2(d\nu^N_{\tilde{h}})$ :

$$N^{2}L_{g,\tilde{h},\tilde{\lambda}}^{*}F(\eta) := N^{2}\sum_{i\in\mathbb{T}_{N}} \left[ c_{\tilde{h},\tilde{\lambda}}(\eta,\eta^{i,i+1})F(\eta^{i,i+1}) \frac{\nu_{g}^{N}(\eta^{i,i+1})}{\nu_{g}^{N}(\eta)} - c_{\tilde{h},\tilde{\lambda}}(\eta,\eta^{i,i+1})F(\eta) \right].$$
(3.40)

A similar expression arises in the bound of exponential moments of time-integrated observables using the Feynman-Kac inequality:

$$\frac{1}{T}\log\mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{g}^{N}}\left[\exp\left[\int_{0}^{T}F(\eta_{t})\,dt\right]\right] \leq \sup_{f\geq0:\nu_{g}^{N}(f)=1}\left\{\nu_{g}^{N}(fF)+\frac{N^{2}}{2}\nu_{g}^{N}(fL_{g,\tilde{h},\tilde{\lambda}}^{*}\mathbf{1})-\frac{N^{2}}{2}\nu_{g}^{N}(\Gamma_{\tilde{h},\tilde{\lambda}}(\sqrt{f}))\right\}.$$
(3.41)

Using Lemma 3.5, the entropy estimate (3.2) in Proposition 3.2 follows from the following claim as proven next.

**Lemma 3.6.** Take  $g = \tilde{h}$  and write  $\mathbf{H}^N := \mathbf{H}^N_{\tilde{h}}$  for the associated relative entropy. For each  $N \in \mathbb{N}_{\geq 1}$ , there is a function  $\mathcal{R}_{\tilde{h},\tilde{\lambda}} : \Omega_N \to \mathbb{R}$  such that, for any density f for  $\nu_{\tilde{h}}^N$ :

$$\nu_{\tilde{h}}^{N}\left(fN^{2}L_{\tilde{h},\tilde{h},\tilde{\lambda}}^{*}\mathbf{1}\right) \leq \frac{N^{2}}{2}\nu_{\tilde{h}}^{N}\left(\Gamma_{\tilde{h},\tilde{\lambda}}(\sqrt{f})\right) + \nu_{\tilde{h}}^{N}\left(f\mathcal{R}_{\tilde{h},\tilde{\lambda}}\right).$$
(3.42)

Moreover, there are  $\gamma, C > 0$  such that:

$$\forall N \in \mathbb{N}_{\geq 1}, \qquad \frac{1}{\gamma} \log \nu_{\tilde{h}}^{N} \Big( \exp \left[ \gamma | \mathcal{R}_{\tilde{h}, \tilde{\lambda}} | \right] \Big) \leq \frac{C}{N^{1/2}}.$$
(3.43)

Let us prove the entropy estimate in Proposition 3.2 assuming Lemma 3.6. The remaining part of Proposition 3.2, the moment bound on  $\Pi_t^N$ , is proven in Corollary 3.9 below. Recall that the entropy inequality (see Appendix A.8 in [38]) states: for any density f for  $\nu_{\tilde{h}}^N$ , any  $F: \Omega_N \to \mathbb{R}$  and  $\gamma > 0$ ,

$$\nu_{\tilde{h}}^{N}(fF) \leq \frac{\mathbf{H}^{N}(f)}{\gamma} + \frac{1}{\gamma} \log \nu_{\tilde{h}}^{N}\left(e^{\gamma F}\right).$$
(3.44)

Using Lemma 3.6 and the entropy inequality, the bound (3.38) on  $\partial_t \mathbf{H}^N(f_t)$  becomes:

$$\partial_{t} \mathbf{H}^{N}(f_{t}) \leq -\frac{N^{2}}{2} \nu_{\tilde{h}}^{N} \left( \Gamma_{\tilde{h},\tilde{\lambda}}(\sqrt{f_{t}}) \right) + \nu_{\tilde{h}}^{N} \left( \mathcal{R}_{\tilde{h},\tilde{\lambda}} \right)$$
$$\leq \frac{\mathbf{H}^{N}(f_{t})}{\gamma} + \frac{1}{\gamma} \log \nu_{\tilde{h}}^{N} \left[ e^{\gamma \mathcal{R}_{\tilde{h},\tilde{\lambda}}} \right] \leq \frac{\mathbf{H}^{N}(f_{t})}{\gamma} + \frac{C}{N^{1/2}}.$$
(3.45)

Gronwall inequality and  $\mathbf{H}^{N}(f_{0}) = 0$  then give  $\mathbf{H}^{N}(f_{t}) \leq \gamma C e^{t/\gamma} N^{-1/2}$  as claimed.

The proof of Lemma 3.6 spans the next two sections, with useful concentration estimates (such as (3.43)) established in Appendix A.

Before we start proving Lemma 3.6, we give a useful corollary. It states that the relative entropy estimate implies the so-called Boltzmann-Gibbs principle, or replacement lemma at the level of fluctuations, as already observed in [36]. Corollary 3.7 was already used in Section 3.1.2 to identify the limiting martingale process. **Corollary 3.7** (Bounding error terms). Assume  $\theta^N : \Omega_N \to \mathbb{R}$ ,  $N \in \mathbb{N}_{\geq 1}$  satisfies the following conditions: there are functions  $\zeta_{\delta}^N$  ( $N \in \mathbb{N}_{\geq 1}$ ,  $\delta \in (0, 1/2)$ ) such that, for any f for  $\nu_{\tilde{h}}^N$ :

$$\nu_{\tilde{h}}^{N} \left( f(\pm \theta^{N}) \right) \leq \delta N^{2} \nu_{\tilde{h}}^{N} \left( \Gamma(\sqrt{f}) \right) + \nu_{\tilde{h}}^{N} \left( f \zeta_{\delta}^{N} \right), \tag{3.46}$$

and for some  $\gamma > 0$  depending on  $\delta$ :

$$\lim_{N \to \infty} \sup \log \nu_{\tilde{h}}^{N} \Big[ \exp \left( \gamma |\zeta_{\delta}^{N}| \right) \Big] = 0.$$
(3.47)

Then:

$$\limsup_{N \to \infty} \mathbb{E}_{\tilde{h}, \tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \left[ \left| \int_{0}^{T} \theta^{N}(\eta_{t}) dt \right| \right] = 0.$$
(3.48)

If  $\theta^N = \zeta_{\delta}^N$  for some  $\delta \in (0, 1/2)$ , the assumptions in the corollary are just saying that  $\gamma |\theta^N|$  has exponential moments under  $\nu_{\tilde{h}}^N$  equal to  $1 + o_N(1)$ . The result is then a direct consequence of the entropy estimate of Proposition 3.2 applied at each time  $t \in [0, T]$ . The more subtle  $\theta^N \neq \zeta_{\delta}^N$  case will be proven at the end of Section 3.4.2.

#### 3.4.1 Computation of the adjoint

Here, we compute the adjoint  $N^2 L^*_{g,\tilde{h},\tilde{\lambda}} \mathbf{1}(\eta)$  of the generator in  $\mathbb{L}^2(\nu_g^N)$  for a fixed configuration  $\eta \in \Omega_N$ . For any  $f : \mathbb{Z}^2 \to \mathbb{R}$  and any  $(i, j) \in \mathbb{Z}^2$ , write for short:

$$\partial_1^N f(i,j) := N \big[ f(i+1,j) - f(i,j) \big], \Delta_1^N f(i,j) := N^2 \big[ f(i+1,j) + f(i-1,j) - 2f(i,j) \big].$$
(3.49)

Recall that  $\bar{\eta}_i = \eta_i - 1/2$  for  $i \in \mathbb{T}_N$ . Elementary computations yield, for each  $i \in \mathbb{T}_N$ :

$$\frac{\nu_g^N(\eta^{i,i+1})}{\nu_g^N(\eta)} = \exp\left[-2(\eta_{i+1} - \eta_i)B_i^g/N\right],\tag{3.50}$$

with:

$$B_{i}^{g}(\eta) =: \frac{1}{2N} \sum_{j \notin \{i, i+1\}} \bar{\eta}_{j} \partial_{1}^{N} g_{i, j}.$$
(3.51)

Similarly,

$$c_{\tilde{h},\tilde{\lambda}}(\eta,\eta^{i,i+1}) = c_{\tilde{\lambda}}(\eta,\eta^{i,i+1}) \exp\left[-(\eta_{i+1}-\eta_i)B_i^{\tilde{h}}/N\right] \\ = c(\eta,\eta^{i,i+1}) \exp\left[-(\eta_{i+1}-\eta_i)\frac{(\tilde{\lambda}+B_i^{\tilde{h}})}{N}\right].$$
(3.52)

With these relations, one has for the adjoint  $N^2L^*1$ :

$$N^{2}L_{g,\tilde{h},\tilde{\lambda}}^{*}\mathbf{1}(\eta) = N^{2}\sum_{i\in\mathbb{T}_{N}} \left[ c_{\tilde{h},\tilde{\lambda}}(\eta^{i,i+1},\eta) \frac{\nu_{g}^{N}(\eta^{i,i+1})}{\nu_{g}(\eta)} - c_{\tilde{h},\tilde{\lambda}}(\eta,\eta^{i,i+1}) \right]$$
$$= N^{2}\sum_{i\in\mathbb{T}_{N}} \eta_{i}(1-\eta_{i+1}) \left( \exp\left[\frac{-\tilde{\lambda}+B_{i}^{2g-\tilde{h}}}{N}\right] - \exp\left[\frac{\tilde{\lambda}+B_{i}^{\tilde{h}}}{N}\right] \right)$$
$$+ N^{2}\sum_{i\in\mathbb{T}_{N}} \eta_{i+1}(1-\eta_{i}) \left( \exp\left[\frac{\tilde{\lambda}-B_{i}^{2g-\tilde{h}}}{N}\right] - \exp\left[-\frac{\tilde{\lambda}+B_{i}^{\tilde{h}}}{N}\right] \right). \quad (3.53)$$

Developing the exponentials using  $e^x = 1 + x + \frac{x^2}{2} + \frac{x^3}{6} + \int_0^1 \frac{(1-t)^3}{6} x^4 e^{tx} dt$  yields:

$$N^{2}L_{g,\tilde{h},\tilde{\lambda}}^{*}\mathbf{1}(\eta) = 2N\sum_{i\in\mathbb{T}_{N}} \left(\eta_{i+1} - \eta_{i}\right) \left(\tilde{\lambda} - B_{i}^{g-\tilde{h}}\right) + 2\sum_{i\in\mathbb{T}_{N}} c(\eta,\eta^{i,i+1}) \left[ \left(B_{i}^{g-\tilde{h}} - \tilde{\lambda}\right) B_{i}^{g} \right] + \frac{1}{6N}\sum_{i\in\mathbb{T}_{N}} \left(\eta_{i+1} - \eta_{i}\right) \left[ \left(\tilde{\lambda} - B_{i}^{2g-\tilde{h}}\right)^{3} - \left(\tilde{\lambda} + B_{i}^{\tilde{h}}\right)^{3} \right] + \delta_{\mathrm{order}\geq4}^{N}(\eta) \qquad (3.54)$$
$$=: \sum_{k=1}^{3} \left( N^{2}L_{g,\tilde{h},\tilde{\lambda}}^{*}\mathbf{1}(\eta) \right)_{|\mathrm{order } k} + \delta_{\mathrm{order}\geq4}^{N}(\eta),$$

with:

$$\sup_{\eta \in \Omega_N} \left| \delta^N_{\text{order} \ge 4}(\eta) \right| = O(N^{-1}).$$
(3.55)

The bound on  $\delta^N_{\text{order}\geq 4}$  comes from the fact that  $B^g_i, B^{\tilde{h}}_i$  are bounded uniformly in  $i, N, \eta$ . We now focus on computing  $\left(N^2 L^*_{g,\tilde{h},\tilde{\lambda}} \mathbf{1}(\eta)\right)_{|\text{order } k}$  for  $k \in \{1, 2, 3\}$ .

**Computing**  $\left(N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1}(\eta)\right)_{|\text{order }1}$ . Since  $(\eta_{i+1} - \eta_i) = (\bar{\eta}_{i+1} - \bar{\eta}_i)$ , upon integrating  $\bar{\eta}_i$  by parts in the first sum in (3.54) and re-indexing, one finds:

$$\left( N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1}(\eta) \right)_{|\text{order 1}} = \frac{1}{N} \sum_{i \in \mathbb{T}_N} \sum_{j:|j-i|>1} \bar{\eta}_j \bar{\lambda}_1^N (g-h)_{i,j} + \sum_{i \in \mathbb{T}_N} \bar{\eta}_i \bar{\eta}_{i+1} \left[ \partial_1^N (g-\tilde{h})_{i+1,i} - \partial_1^N (g-\tilde{h})_{i-1,i+1} \right].$$
(3.56)

**Computing**  $\left(N^2 L^*_{g,\tilde{h},\tilde{\lambda}} \mathbf{1}(\eta)\right)_{|\text{order } 2}$ . Recall definition (3.51) of  $B^g$ . One has:

$$\left( N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1}(\eta) \right)_{|\text{order } 2} = \frac{1}{2N^2} \sum_{j,\ell \in \mathbb{T}_N} \bar{\eta}_j \bar{\eta}_\ell \left( \frac{1}{N} \sum_{\substack{i \in \mathbb{T}_N \\ i \notin \{j-1,j,\ell-1,\ell\}}} c(\eta,\eta^{i,i+1}) \partial_1^N (g-\tilde{h})_{i,j} \partial_1^N g_{i,\ell} \right)$$
$$- \frac{\tilde{\lambda}}{N} \sum_{i \in \mathbb{T}_N} c(\eta,\eta^{i,i+1}) \sum_{j \notin \{i,i+1\}} \bar{\eta}_j \partial_1^N g_{i,j}.$$
(3.57)

To proceed, rewrite the jump rates  $c(\eta, \eta^{i,i+1})$  in terms of  $\bar{\eta}_i, \bar{\eta}_{i+1}$  as follows:

$$c(\eta, \eta^{i,i+1}) = \eta_i(1 - \eta_{i+1}) + \eta_{i+1}(1 - \eta_i) = \frac{1}{2} - 2\bar{\eta}_i\bar{\eta}_{i+1} = 2(\sigma - \bar{\eta}_i\bar{\eta}_{i+1}).$$
(3.58)

Splitting also the first sum in (3.57) between  $j = \ell$  diagonal terms and non-diagonal ones and recalling  $\bar{\eta}_i^2 = \sigma$ , one finds:

$$\left( N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1}(\eta) \right)_{|\text{order } 2} = \frac{\sigma}{N} \sum_{\substack{j,\ell \in \mathbb{T}_N \\ j \neq \ell}} \bar{\eta}_j \bar{\eta}_\ell \left( \frac{1}{N} \sum_{\substack{i \in \mathbb{T}_N \\ i \notin \{j-1,j,\ell-1,\ell\}}} \partial_1^N (g - \tilde{h})_{i,j} \partial_1^N g_{i,\ell} \right)$$
$$+ \frac{\sigma^2}{N^2} \sum_{j \in \mathbb{T}_N} \sum_{\substack{i \in \mathbb{T}_N \\ i \notin \{j-1,j\}}} \partial_1^N (g - \tilde{h})_{i,j} \partial_1^N g_{i,j} + \varepsilon_{g,\tilde{h},\tilde{\lambda}} + \mathcal{E}_g.$$
(3.59)

The term  $\mathcal{E}_g$  contains third-order correlations that will require more work to estimate:

$$\mathcal{E}_g(\eta) = \frac{2\lambda}{N} \sum_{i \in \mathbb{T}_N} \bar{\eta}_i \bar{\eta}_{i+1} \sum_{j \notin \{i,i+1\}} \bar{\eta}_j \partial_1^N g_{i,j}.$$
(3.60)

The term  $\varepsilon_{a,\tilde{h}}$  in (3.59) contains all other terms that will be shown to be small:

$$\varepsilon_{g,\tilde{h},\tilde{\lambda}} = -\frac{1}{N} \sum_{i \in \mathbb{T}_N} \bar{\eta}_i \bar{\eta}_{i+1} \left( \frac{1}{N} \sum_{j \notin \{i,i+1\}} \sigma \partial_1^N (g - \tilde{h})_{i,j} \partial_1^N g_{i,j} \right) - \frac{1}{N} \sum_{i \in \mathbb{T}_N} \bar{\eta}_i \bar{\eta}_{i+1} \left( \frac{1}{N} \sum_{j \neq \ell \notin \{i,i+1\}} \bar{\eta}_j \bar{\eta}_\ell \partial_1^N g_{i,j} \partial_1^N g_{i,\ell} \right) - \frac{2\tilde{\lambda}\sigma}{N} \sum_{i \in \mathbb{T}_N} \sum_{j \notin \{i,i+1\}} \bar{\eta}_j \partial_1^N g_{i,j}.$$
(3.61)

Putting together the initial expression (3.54) of the adjoint and the first and second order estimates (3.56)-(3.59), we have so far obtained:

$$N^{2}L_{g,\tilde{h},\tilde{\lambda}}^{*}\mathbf{1}(\eta) = \frac{1}{N} \sum_{i \in \mathbb{T}_{N}} \sum_{j:|j-i|>1} \bar{\eta}_{i}\bar{\eta}_{j}\Delta_{1}^{N}(g-\tilde{h})_{i,j} + \sum_{i \in \mathbb{T}_{N}} \bar{\eta}_{i}\bar{\eta}_{i+1} \left[\partial_{1}^{N}(g-\tilde{h})_{i+1,i} - \partial_{1}^{N}(g-\tilde{h})_{i-1,i+1}\right] + \frac{\sigma}{N} \sum_{\substack{j,\ell \in \mathbb{T}_{N} \\ j \neq \ell}} \bar{\eta}_{j}\bar{\eta}_{\ell} \left(\frac{1}{N} \sum_{\substack{i \in \mathbb{T}_{N} \\ i \notin \{j-1,j,\ell-1,\ell\}}} \partial_{1}^{N}(g-\tilde{h})_{i,j}\partial_{1}^{N}g_{i,\ell}\right) + \frac{\sigma^{2}}{N^{2}} \sum_{j \in \mathbb{T}_{N}} \sum_{\substack{i \in \mathbb{T}_{N} \\ i \notin \{j-1,j\}}} \partial_{1}^{N}(g-\tilde{h})_{i,j}\partial_{1}^{N}g_{i,j} + \delta_{g,\tilde{h},\tilde{\lambda}}(\eta),$$
(3.62)

where  $\delta_{q,\tilde{h}}$  is given by:

$$\delta_{g,\tilde{h},\tilde{\lambda}} = \varepsilon_{g,\tilde{h},\tilde{\lambda}} + \mathcal{E}_g + \left(N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1}(\eta)\right)_{|\text{order }3} + O(N^{-1}), \tag{3.63}$$

with the  $O(N^{-1})$  uniform on the configuration.

Postponing the precise estimate of  $\delta_{g,\tilde{h},\tilde{\lambda}}$  to Section 3.4.2, let us informally explain why the choice  $g = \tilde{h}$  is optimal. The contribution of two-point correlations i.e. the first four lines of (3.62), are expected to typically be of order 1 in N since  $\Pi^N$  is, informally speaking, the square of the approximately Gaussian random variable  $Y^N$ . On the other hand  $\delta_{g,\tilde{h},\tilde{\lambda}}$  involves higher order correlations or other terms that we expect to typically be of order  $o_N(1)$ . Thus, in order to make the adjoint as small as possible as stated in Lemma 3.5, we need to choose g so that all two-point correlations vanish. This is the case when  $g = \tilde{h}$  and we obtain:

$$N^{2}L^{*}_{\tilde{h},\tilde{h},\tilde{\lambda}}\mathbf{1}(\eta) = \delta_{\tilde{h},\tilde{h},\tilde{\lambda}}.$$
(3.64)

Therefore bounding  $\delta_{\tilde{h},\tilde{h},\tilde{\lambda}}$  is the same as bounding the adjoint.

In Section 4 we will make use of a bound of  $\delta_{g,\tilde{h},\tilde{\lambda}}$  for  $g \neq \tilde{h}$ , so we keep a general g for now. In the next section, we conclude the proof of Lemma 3.6 by constructing, for each  $\alpha \in (0, 1/2]$ , a function  $\mathcal{R}_{g,\tilde{h},\tilde{\lambda},\alpha}: \Omega_N \to \mathbb{R}$  such that, for each density f for  $\nu_{\tilde{h}}^N$ :

$$\nu_g^N \left( f \delta_{g,\tilde{h},\tilde{\lambda}} \right) \le \alpha N^2 \nu_g^N \left( \Gamma(\sqrt{f}) \right) + \nu_g^N \left( f \mathcal{R}_{g,\tilde{h},\tilde{\lambda},\alpha} \right), \tag{3.65}$$

with, for some  $\gamma_{\alpha}, C_{\alpha} > 0$  independent of N and all  $N \in \mathbb{N}_{\geq 1}$ :

$$\frac{1}{\gamma_{\alpha}}\log\nu_{g}^{N}\left(\exp\left|\gamma_{\alpha}\mathcal{R}_{g,\tilde{h},\tilde{\lambda},\alpha}\right|\right) \leq \frac{C_{\alpha}}{N^{1/2}}.$$
(3.66)

The function  $\mathcal{R}_{\tilde{h},\tilde{\lambda}}$  appearing in Lemma 3.5 is simply  $\mathcal{R}_{q=\tilde{h},\tilde{h},\tilde{\lambda},\alpha=1/2}$ .

#### **3.4.2** Proof of (3.65) and estimate of error terms

Here, we complete the proof of Lemma 3.6 by proving (3.65). At the end of the section, we also establish the moment bound in Proposition 3.2 and prove Corollary 3.7, which in particular implies the estimate (3.13). Recall that we still work with a general  $g \in C_D^{\infty}(\mathbb{T}^2)$  such that  $\sigma g$  has leading eigenvalue strictly below 1.

The proof of (3.65) is split into two parts: the estimate of  $\mathcal{E}_{\tilde{h}}$  (recall (3.60)), and the estimate of the rest, i.e.  $\varepsilon_{g,\tilde{h},\tilde{\lambda}} + (N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1})_{|\text{order 3}}$  (recall that  $\varepsilon_{g,\tilde{h},\tilde{\lambda}}$  is defined in (3.61)).

In Proposition A.4, the following concentration estimates under  $\nu_{\tilde{h}}^N$  are established. Let  $n \geq 1$  be an integer and  $\phi_n : \mathbb{T}_N^n \to \mathbb{R}$   $(N \in \mathbb{N}_{\geq} 1)$  be a sequence of functions satisfying  $\sup_N \|\phi_n\|_{\infty} < \infty$ . For  $J \subset \mathbb{T}_N$  and  $n \geq 2$ , define:

$$W_{\phi_n}^{n,J}(\eta) := \frac{1}{N^{n-1}} \sum_{i_1,\dots,i_n \in \mathbb{T}_N} \phi_n(i_1,\dots,i_n) \Big(\prod_{j \in J} \bar{\eta}_{i_1+j}\Big) \bar{\eta}_{i_2}\dots\bar{\eta}_{i_n}, \qquad \eta \in \Omega_N.$$
(3.67)

If n = 1, define instead:

$$W_{\phi_1}^{1,J}(\eta) := \frac{1}{N^{1/2}} \sum_{i \in \mathbb{T}_N} \phi_1(i) \Big(\prod_{j \in J} \bar{\eta}_{i+j}\Big), \qquad \eta \in \Omega_N.$$
(3.68)

There is then  $\gamma_n, C_n > 0$  depending only  $\tilde{h}, J$  but not on  $\phi_n$  such that, for each  $N \in \mathbb{N}$ :

$$\frac{1}{\gamma_n} \log \nu_g^N \Big( \exp\left[\frac{\gamma_n |W_{\phi_n}^{n,K}|}{\sup_N \|\phi_n\|_{\infty}}\right] \Big) \le \begin{cases} C_1 & \text{if } n \in \{1,2\},\\ \frac{C_n}{N^{\frac{n-2}{2}}} & \text{if } n \ge 3. \end{cases}$$
(3.69)

Let us use (3.69) to bound  $\varepsilon_{g,\tilde{h},\tilde{\lambda}}$  and  $\left(N^2 L^*_{g,\tilde{h},\tilde{\lambda}} \mathbf{1}(\eta)\right)_{|\text{order 3}}$ . The first line of  $\varepsilon_{g,\tilde{h},\tilde{\lambda}}$  in (3.61) is of the form  $N^{-1/2} W^{1,\{0,1\}}_{\phi_1}$  for a bounded  $\phi_1$  depending on  $g, \tilde{h}$  and given by:

$$\phi_1(i) := -\frac{1}{N} \sum_{j \notin \{i, i+1\}} \sigma \partial_1^N (g - \tilde{h})_{i,j} \partial_1^N g_{i,j}, \qquad i \in \mathbb{T}_N.$$
(3.70)

On the other hand, the second line of  $\varepsilon_{g,\tilde{h},\tilde{\lambda}}$  is of the form  $W^{3,\{0,1\}}_{\phi_3}$  with:

$$\phi_3(i,j,\ell) = -\mathbf{1}_{j \neq \ell \notin \{i,i+1\}} \partial_1^N g_{i,j} \partial_1^N g_{i,\ell}, \qquad (i,j,\ell) \in \mathbb{T}_N^3.$$
(3.71)

Lastly, the last line of  $\varepsilon_{g,\tilde{h},\tilde{\lambda}}$  reads, exchanging sums and integrating by parts:

$$-\frac{2\tilde{\lambda}\sigma}{N}\sum_{i\in\mathbb{T}_N}\sum_{j\notin\{i,i+1\}}\bar{\eta}_j\partial_1^N g_{i,j} = -2\tilde{\lambda}\sigma\sum_{j\in\mathbb{T}_N}\bar{\eta}_j\sum_{i\notin\{j-1,j\}}\left[g_{i+1,j} - g_{i,j}\right]$$
$$=\frac{2\tilde{\lambda}\sigma}{N}\sum_{j\in\mathbb{T}_N}\bar{\eta}_j N\left[g_{j+1,j} - g_{j-1,j}\right].$$
(3.72)

Since  $g \in C_D^{\infty}(\mathbb{T}^2)$ , N times the bracket is bounded uniformly in N, j:

$$\psi_{2}(j) := N [g_{j+1,j} - g_{j-1,j}] = N [g_{j+1,j} - g_{j+,j}] + N [g_{j-,j} - g_{j-1,j}]$$
  
=  $\partial_{1}g_{j+,j} - \partial_{1}g_{j-,j} + o_{N}(1),$  (3.73)

with the  $o_N(1)$  uniform in j and where we used the continuity of g on the diagonal. The last line of (3.72) is thus of the form  $N^{-1/2}W^{1,\{0\}}_{\psi_1}$ . From (3.69) and the entropy inequality applied to each three terms in  $\varepsilon_{g,\tilde{h},\tilde{\lambda}}$ , it follows that there are constants  $\gamma_1, \gamma'_1, \gamma_3 > 0$  and  $C_1, C'_1, C_3 > 0$  such that, for any density f for  $\nu_g^N$  (recall  $\mathbf{H}_g^N(f) := \nu_g^N(f \log f)$ ):

$$\nu_{g}^{N}(f\varepsilon_{g,\tilde{h},\tilde{\lambda}}) \leq \frac{\mathbf{H}_{g}^{N}(f)}{\gamma_{1}N^{1/2}} + \frac{1}{\gamma_{1}}\log\nu_{g}^{N}\left(\exp\left[\gamma_{1}W_{\phi_{1}}^{1,\{0,1\}}\right]\right) + \frac{\mathbf{H}_{g}^{N}(f)}{\gamma_{3}N^{1/2}} \\ + \frac{1}{\gamma_{3}}\log\nu_{g}^{N}\left(\exp\left[\gamma_{3}W_{\phi_{3}}^{3,\{0,1\}}\right]\right) + \frac{\mathbf{H}_{g}^{N}(f)}{\gamma_{1}'N^{1/2}} + \frac{1}{\gamma_{1}'N^{1/2}}\log\nu_{g}^{N}\left(\exp\left[\gamma_{1}'W_{\psi_{1}}^{1,\{0\}}\right]\right) \\ \leq \mathbf{H}_{g}^{N}(f)\left[\frac{1}{\gamma_{3}} + \frac{1}{\gamma_{1}N^{1/2}} + \frac{1}{\gamma_{1}'N^{1/2}}\right] + \frac{C_{1} + C_{1}' + C_{3}}{N^{1/2}}.$$
(3.74)

Consider now  $\left(N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1}(\eta)\right)_{|\text{order }3}$ . Recalling the expression (3.54) of the third order term,

$$\left( N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1}(\eta) \right)_{|\text{order } 3} = \frac{1}{6N} \sum_{i \in \mathbb{T}_N} \left( \eta_{i+1} - \eta_i \right) \left[ -6\tilde{\lambda}^2 B_i^g + 12\tilde{\lambda} B_i^g B_i^{g-\tilde{h}} - \left( B_i^{2g-\tilde{h}} \right)^3 - \left( B_i^{\tilde{h}} \right)^3 \right].$$

$$(3.75)$$

Since  $B_i^g, B_i^{\tilde{h}}$  are bounded uniformly in i, N and  $\eta_{i+1} - \eta_i = \bar{\eta}_{i+1} - \bar{\eta}_i$ , (3.75) can be bounded as follows for some  $C(g, \tilde{h}, \tilde{\lambda}) > 0$ :

$$\left(N^{2}L_{g,\tilde{h},\tilde{\lambda}}^{*}\mathbf{1}(\eta)\right)_{|\text{order }3} \leq -\frac{\tilde{\lambda}^{2}}{N} \sum_{i \in \mathbb{T}_{N}} (\bar{\eta}_{i+1} - \bar{\eta}_{i})B_{i}^{g} + \frac{C(g,\tilde{h},\tilde{\lambda})}{N} \sum_{i \in \mathbb{T}_{N}} \left[ (B_{i}^{g})^{2} + (B_{i}^{\tilde{h}})^{2} \right].$$
(3.76)

The first term in the right-hand side of (3.76) can be integrated by parts, after which it becomes bounded by  $O(N^{-1})$  uniformly in the configuration. On the other hand, for each  $i \in \mathbb{T}_N$ ,  $(B_i^g)^2$ ,  $(B_i^{\tilde{h}})^2$  are of the form  $N^{-1}W_{\psi_2^i}^{2,\{0\}}$  for a family  $\psi_2^i : \mathbb{T}_N^2 \to \mathbb{R}$  satisfying  $\sup_{N,i,\mathbb{T}_N^2} |\psi_2^i| < \infty$ . Equation (3.69) and the entropy inequality thus yield similarly to (3.74), for some  $C_2, \gamma_2 > 0$  depending only on  $g, \tilde{h}, \tilde{\lambda}$ :

$$\nu_g^N \left( f \left( N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1}(\eta) \right)_{|\text{order } 3} \right) \le \frac{\mathbf{H}_g^N(f)}{\gamma_2 N} + \frac{C_2}{N}.$$
(3.77)

Putting together (3.74)–(3.77), we have shown the existence of  $\gamma, C > 0$ , depending only on  $g, \tilde{h}, \tilde{\lambda}$ , such that (recall the definition (3.60) of  $\mathcal{E}_g$  and (3.63) of  $\delta_{g,\tilde{h},\tilde{\lambda}}$ ):

$$\nu_g^N(f\delta_{g,\tilde{h},\tilde{\lambda}}) \le \nu_g^N(f\mathcal{E}_g) + \frac{\mathbf{H}_g^N(f)}{\gamma} + \frac{C}{N^{1/2}}.$$
(3.78)

Lemma 3.8 below provides an estimate of  $\mathcal{E}_g$ , concluding the proof of (3.65) with the function  $\mathcal{R}_{q,\tilde{h},\tilde{\lambda},\alpha}$  ( $\alpha \in (0, 1/2]$ ) appearing there given by:

$$\mathcal{R}_{g,\tilde{h},\tilde{\lambda},\alpha} := \left( N^2 L_{g,\tilde{h},\tilde{\lambda}}^* \mathbf{1} \right)_{\text{order } 3} + \varepsilon_{g,\tilde{h},\tilde{\lambda}} + \mathcal{W}_{g,\tilde{h},\tilde{\lambda},\alpha}, \tag{3.79}$$

and  $\mathcal{W}_{a,\tilde{h},\tilde{\lambda},\alpha}$  defined in the next lemma.

**Lemma 3.8.** Let  $\mathcal{E}_g$  be given by (3.60) and  $\alpha \in (0, 1/2]$ . There is  $\mathcal{W}_{g,\tilde{h},\tilde{\lambda},\alpha} : \Omega_N \to \mathbb{R}$  such that, for any density f for  $\nu_g^N$ :

$$\nu_g^N(f\mathcal{E}_g) \le \alpha N^2 \nu_g^N(\Gamma_{\tilde{h},\tilde{\lambda}}(\sqrt{f})) + \nu_g^N(f\mathcal{W}_{\tilde{h},\tilde{\lambda}}), \qquad (3.80)$$

and, for some  $\gamma, C > 0$  depending on  $g, \tilde{h}, \tilde{\lambda}, \alpha$  but not on N:

$$\frac{1}{\gamma} \log \nu_g^N \Big( \exp\left[\gamma |\mathcal{W}_{\tilde{h},\tilde{\lambda},\alpha}|\right] \Big) \le \frac{C}{N^{1/2}}.$$
(3.81)

Proof of Lemma 3.8. Fix a density f for  $\nu_g^N$  throughout the proof. Consider more generally a family of functions  $\phi_2 : \mathbb{T}_N^2 \to \mathbb{R}$  that is bounded uniformly in N and let us prove Lemma 3.8 for  $W_{\phi_2}^{2,\{0,1\}}$  (recall (3.67)) rather than  $\mathcal{E}_{\tilde{h}}$  (defined in (3.60)). A direct application of the entropy inequality only yields a bound  $O_N(1)$  on the exponential moment. To improve this bound to  $O(N^{-1/2})$ , we make use of the long, diffusive time-scale, which appears in the bound (3.38) on the entropy production  $\partial_t \mathbf{H}^N(f_t)$  of the relative entropy through the carrédu champ. Following the technique of [36], we estimate the cost of replacing the sum on  $\bar{\eta}_i \bar{\eta}_{i+1}$  by a sum on local averages of  $\bar{\eta}$ 's. To do so, let  $a \in (0, 1/2)$ . For ease of notation, we write aN instead of  $\lfloor aN \rfloor$ . Split  $W_{\phi_2}^{2,\{0,1\}}$  as follows:

$$W^{2,\{0,1\}}_{\phi_2} = W^+ + W^- + W^=, \qquad (3.82)$$

with, for  $\eta \in \Omega_N$ :

$$W^{+}(\eta) := \frac{1}{N} \sum_{i \in \mathbb{T}_{N}} \sum_{j \notin \{i, \dots, i+aN\}} \bar{\eta}_{i} \bar{\eta}_{i+1} \bar{\eta}_{j} \phi_{2}(i, j),$$

$$W^{-}(\eta) := \frac{1}{N} \sum_{i \in \mathbb{T}_{N}} \sum_{j \in \{i+2, \dots, i+aN\}} \bar{\eta}_{i} \bar{\eta}_{i+1} \bar{\eta}_{j} \phi_{2}(i, j)$$

$$W^{=}(\eta) := \frac{1}{N} \sum_{i \in \mathbb{T}_{N}} \sum_{j \in \{i, i+1\}} \bar{\eta}_{i} \bar{\eta}_{i+1} \bar{\eta}_{j} \phi_{2}(i, j).$$
(3.83)

Note that, recalling  $(\bar{\eta}_i)^2 = \sigma$   $(i \in \mathbb{T}_N)$ ,  $W^=$  is of the form  $N^{-1/2}W^{1,\{0,1\}}_{\psi_1}$  for  $\psi_1 = \sigma[\phi_2(i,i) + \phi_2(i,i+1)]$ . One thus directly has by (3.69) and the entropy inequality, for some  $\gamma, C > 0$ :

$$\nu_{\tilde{h}}^{N}(fW^{=}) \leq \frac{\mathbf{H}_{g}^{N}(f)}{\gamma N^{1/2}} + \frac{1}{\gamma N^{1/2}} \log \nu_{g}^{N} \Big( \exp\left[\gamma W^{=}\right] \Big) \leq \frac{\mathbf{H}_{g}^{N}(f)}{\gamma N^{1/2}} + \frac{C}{N^{1/2}}.$$
 (3.84)

We now estimate  $W^+$  and  $W^-$ . For  $i \in \mathbb{T}_N$ , define the local averages:

$$\bar{\eta}_i^{+,aN} := \frac{1}{aN} \sum_{\ell=i}^{i+aN-1} \bar{\eta}_\ell, \qquad \bar{\eta}_i^{-,aN} := \frac{1}{aN} \sum_{\ell=i-aN+1}^{i} \bar{\eta}_\ell, \qquad (3.85)$$

with  $i \pm (aN - 1)$  understood modulo N. The only difference in the estimate of  $W^+, W^-$  is that we replace each  $\eta_{i+1}$  (respectively  $\eta_i$ ) by  $\eta_{i+1}^{+,aN}$  (respectively  $\eta^{-,aN}$ )  $(i \in \mathbb{T}_N)$ , so we only treat  $W^+$ . Introduce  $W^{+,aN}$ , defined for  $\eta \in \Omega_N$  by:

$$W^{+,aN}(\eta) := \frac{1}{N} \sum_{i \in \mathbb{T}_N} \sum_{j \notin \{i,...,i+aN\}} \bar{\eta}_i \bar{\eta}_{i+1}^{+,aN} \bar{\eta}_j \phi_2(i,j)$$
$$= \frac{1}{aN^2} \sum_{i \in \mathbb{T}_N} \sum_{j \notin \{i,...,i+aN\}} \sum_{\ell \in \{i+1,...,i+aN\}} \bar{\eta}_i \bar{\eta}_\ell \bar{\eta}_j \phi_2(i,j).$$
(3.86)

As a < 1/2 implies that the sums on  $j, \ell$  do not overlap,  $W^{+,aN}$  is of the form  $W^{3,\{0\}}_{\phi_3}$  for a bounded  $\phi_3$ , so its log-exponential moments are bounded by  $O(N^{-1/2})$  according to (3.69). We now show that the difference  $W^+ - W^{+,aN}$  can be estimated in terms of the carré du champ. To do so, notice first the following elementary identity:

$$\bar{\eta}_{i+1} - \bar{\eta}_{i+1}^{+,aN} = \sum_{p \in \{i+1,\dots,i+aN-1\}} \varphi_a(p-i) \left(\bar{\eta}_p - \bar{\eta}_{p+1}\right), \qquad \varphi_a(n) = \mathbf{1}_{n \in \{1,\dots,aN-1\}} \left(1 - \frac{n}{aN}\right).$$
(3.87)

Define then, for each  $p \in \mathbb{T}_N$ :

$$u_{p}(\eta) := \frac{1}{N} \sum_{\substack{i \in \mathbb{T}_{N} \\ i \in \{p - (aN - 1), \dots, p - 1\}}} \sum_{j \notin \{i, \dots, i + aN\}} \varphi_{a}(|p - i|_{\mathbb{T}_{N}}) \bar{\eta}_{i} \bar{\eta}_{j} \phi_{2}(i, j) \qquad \eta \in \Omega_{N},$$
(3.88)

with  $|p - i|_{\mathbb{T}_N}$  the torus distance between i, p. The earlier choice of  $a \in (0, 1/2)$  removes any ambiguity in the definition of this distance and:

$$W^{+}(\eta) - W^{+,aN}(\eta) = \sum_{p \in \mathbb{T}_{N}} (\bar{\eta}_{p} - \bar{\eta}_{p+1}) u_{p}(\eta), \qquad \eta \in \Omega_{N}.$$
(3.89)

Applying the integration by parts formula of Lemma B.1 for each  $p \in \mathbb{T}_N$  to  $u = u_p$ , then summing over p gives the existence of  $C = C(q, \tilde{h}, \lambda) > 0$  such that:

$$\nu_{g}^{N} \left( f \left[ W^{+} - W^{+,aN} \right] \right) \leq \alpha N^{2} \nu_{g}^{N} \left( \Gamma_{\tilde{h},\tilde{\lambda}}(\sqrt{f}) \right) + \frac{C}{\alpha N^{2}} \sum_{p \in \mathbb{T}_{N}} \nu_{g}^{N} \left( f |u_{p}|^{2} \right) \\ + \sum_{p \in \mathbb{T}_{N}} \nu_{g}^{N} \left( f u_{p}(\bar{\eta}_{p} - \bar{\eta}_{p+1}) \left( \exp \left[ -\frac{2(\bar{\eta}_{p+1} - \bar{\eta}_{p})}{N} B_{p}^{\tilde{h}} \right] - 1 \right) \right).$$
(3.90)

Let us estimate the different terms in the right-hand side of (3.90). The quantity:

$$\mathcal{R}^+_{\alpha} := \frac{C}{\alpha N} \sum_{p \in \mathbb{T}_N} |u_p|^2 \tag{3.91}$$

is of the form  $W_{\phi_4}^{4,\{0\}}$  for some bounded  $\phi_4$ . By (3.69), it thus satisfies, for some  $\gamma_4, C_4 > 0$  (depending on  $\alpha$ ):

$$\nu_g^N(f\mathcal{R}^+_\alpha) \le \frac{\mathbf{H}_g^N(f)}{\gamma_4} + \frac{C}{\gamma_4 N}.$$
(3.92)

On the other hand, using  $|e^x - 1 - x| \leq cx^2$  for some c > 0 and all x smaller than  $2\sup_{p,N} N^{-1}|B_p^{\tilde{h}}|$ , we find that the second line in (3.90) is bounded by:

$$\nu_g^N(f\mathcal{R}^+) := -\frac{2}{N} \sum_{p \in \mathbb{T}_N} \nu_g^N(fu_p B_p^{\tilde{h}}) + \frac{c}{N} \sum_{p \in \mathbb{T}_N} \left( \sup_{p,N} \frac{|u_p|}{N} \right) \nu_g^N(f(B_p^{\tilde{h}})^2).$$
(3.93)

The first term is an average of functions of the form  $W^{3,\{0\}}_{\phi_3^p}$ , the second one of the form  $N^{-1}W^{2,\{0\}}_{\phi_2^p}$  for functions  $\phi_2^p, \phi_3^p$  bounded uniformly in  $p \in \mathbb{T}_N$  and N. The log exponential moments under  $\nu_g^N$  of both terms in the right-hand side of (3.93) are thus bounded by  $O(N^{-1/2})$  in a neighbourhood of 0 by (3.69). This concludes the estimate of the cost of replacing  $W^+$  by  $W^{+,aN}$ : for some  $\gamma, C > 0$ ,

$$\nu_{g}^{N}\left(f\left[W^{+}-W^{+,aN}\right]\right) \leq \alpha N^{2}\nu_{g}^{N}\left(\Gamma_{\tilde{h},\tilde{\lambda}}(\sqrt{f})\right) + \nu_{g}^{N}\left(f\mathcal{R}_{\alpha}^{+}\right) + \nu_{g}^{N}\left(f\mathcal{R}^{+}\right)$$
$$\leq \alpha N^{2}\nu_{g}^{N}\left(\Gamma_{\tilde{h},\tilde{\lambda}}(\sqrt{f})\right) + \frac{\mathbf{H}_{g}^{N}(f)}{\gamma} + \frac{C}{N^{1/2}}.$$
(3.94)

Similarly building  $\mathcal{R}^-, \mathcal{R}^-_{\alpha}$  and defining  $\mathcal{W}_{q,\tilde{h},\tilde{\lambda},\alpha}$  as follows concludes the proof:

$$\mathcal{W}_{g,\tilde{h},\tilde{\lambda},\alpha} := W^{+,aN} + \mathcal{R}^+ + \mathcal{R}^+_{\alpha} + W^{-,aN} + \mathcal{R}^- + \mathcal{R}^-_{\alpha} + W^=.$$
(3.95)

Lemma 3.5 provides the entropy bound in Proposition 3.2. We now prove the moment bound on the correlation field at each time.

**Corollary 3.9** (Moment bound for the correlation field). Let  $\phi : \mathbb{T}^2 \to \mathbb{R}$  be bounded. Then, for any  $T \ge 0$ :

$$\sup_{N} \sup_{t \le T} \mathbb{E}_{\tilde{h}, \tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \left[ \left| \Pi_{t}^{N}(\phi) \right|^{3/2} \right] \le C(T, \phi).$$

$$(3.96)$$

*Proof.* Let  $t \ge 0$ . Recall the identity  $\mathbb{E}[X^{3/2}] \le 1 + \frac{3}{2} \int_1^\infty r^{1/2} \mathbb{P}(X > r) dr$  valid for any random variable  $X \ge 0$ . This together with the bound  $|\Pi^N(\phi)| \le C(\phi)N$  for each N gives:

$$\mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \Big[ \big| \Pi_{t}^{N}(\phi) \big|^{3/2} \Big] \leq \frac{3}{2} \int_{0}^{C(\phi)N} r^{1/2} \mathbb{P}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \Big( \big| \Pi_{t}^{N}(\phi) \big| > r \Big) \, dr.$$
(3.97)

Using the entropy inequality (3.7) with a constant  $\gamma = cr$  for c > 0 to be chosen later and the test function  $\mathbf{1}_{|\Pi^N(\phi)|>r}$  (r > 0) yields:

$$\mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \Big[ \big| \Pi_{t}^{N}(\phi) \big|^{3/2} \Big] \leq \frac{3}{2} \int_{0}^{C(\phi)N} \frac{1}{c\sqrt{r}} \Big( \mathbf{H}^{N}(f_{t}) + \log \nu_{g}^{N} \Big[ e^{cr \mathbf{1}_{|\Pi^{N}(\phi)| > r}} \Big] \Big) dr \\
= \frac{3}{2} \int_{0}^{C(\phi)N} \frac{1}{c\sqrt{r}} \Big( \mathbf{H}^{N}(f_{t}) + \log \Big( 1 + (e^{cr} - 1)\nu_{\tilde{h}}^{N} \big( |\Pi^{N}(\phi)| > r \big) \Big) \Big) dr.$$
(3.98)
Proposition A.4 ensures that one can choose  $c = c(\phi) > 0$  such that:

$$(e^{cr}-1)\nu_{\tilde{h}}^{N}(|\Pi^{N}(\phi)|>r) \le e^{-cr/2}, \qquad r>0.$$
 (3.99)

Using the entropy bound of Proposition 3.2, the bound  $\log(1+x) \leq x$  for  $x \geq 0$  and the fact that  $r \mapsto r^{-1/2}e^{-cr/2}$  is integrable on  $\mathbb{R}_+$  concludes the proof.

We now prove Corollary 3.7 on time averages of error terms.

Proof of Corollary 3.7. Only the case  $\theta^N \neq \zeta_{\alpha}^N$  ( $\alpha \in (0, 1/2]$ ) requires a proof as explained below the statement of the corollary. This case corresponds to situations where a renormalisation procedure such as the one performed for  $\mathcal{E}_{\tilde{h}}$  in the proof of Lemma 3.8 is necessary. The proof below is basically the one in [36, Theorem 5.1] adapted to get bounds on expectations of time-integrated observables rather than just tail probabilities. Recall the definition of the function  $\mathcal{R}_{\tilde{h},\tilde{\lambda},\alpha}$  of Lemma 3.6, let a > 1 be a parameter that will eventually become large, define  $\alpha_a := (4a)^{-1}$  and write:

$$\left|\int_{0}^{T} \theta^{N}(\eta_{t}) dt\right| \leq \left|\int_{0}^{T} \left[\theta^{N}(\eta_{t}) - \zeta_{\alpha_{a}}^{N}(\eta_{t})\right] dt\right| - \frac{1}{2a} \int_{0}^{T} \mathcal{R}_{\tilde{h},\tilde{\lambda}}(\eta_{t}) dt + \int_{0}^{T} \left|\zeta_{\alpha_{a}}^{N}(\eta_{t})\right| dt + \frac{1}{2a} \int_{0}^{T} \mathcal{R}_{\tilde{h},\tilde{\lambda}}(\eta_{t}) dt.$$

$$(3.100)$$

By assumption on  $\zeta_{\alpha_a}^N$  and by Lemma 3.6 for  $\mathcal{R}_{\tilde{h},\tilde{\lambda}}$ , the second line is controlled by the entropy inequality:

$$\limsup_{N \to \infty} \mathbb{E}_{\tilde{h}, \tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \left[ \int_{0}^{T} \left| \zeta_{\alpha_{a}}^{N}(\eta_{t}) + \mathcal{R}_{\tilde{h}, \tilde{\lambda}}(\eta_{t}) \right| dt \right] = 0.$$
(3.101)

On the other hand, subtracting  $\zeta_{\alpha_a}^N$  and  $\mathcal{R}_{\tilde{h},\tilde{\lambda}}$  gives us good control on the expectation of the first line in (3.100) as we show below:

$$\mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \left[ \left| \int_{0}^{T} \left[ \theta^{N}(\eta_{t}) - \zeta_{\alpha_{a}}^{N}(\eta_{t}) \right] dt \right| - \frac{1}{2a} \int_{0}^{T} \mathcal{R}_{\tilde{h},\tilde{\lambda}}(\eta_{t}) dt \right] \qquad (3.102)$$

$$\leq \frac{1}{a} \log \mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \left[ \exp \left[ a \right| \int_{0}^{T} \left[ \theta^{N}(\eta_{t}) - \zeta_{\alpha_{a}}^{N}(\eta_{t}) \right] dt \right| - \frac{1}{2} \int_{0}^{T} \mathcal{R}_{\tilde{h},\tilde{\lambda}}(\eta_{t}) dt \right].$$

Inside the exponential, the absolute value can be removed using  $e^{|x|} \leq e^x + e^{-x}$   $(x \in \mathbb{R})$ . It is thus enough to separately bound, uniformly in N, a, the exponential moments of  $\pm a[\theta^N - \zeta_{\alpha_a}^N]$ . Feynman-Kac inequality (see Lemma 3.5) and the bound of Lemma 3.6 on the adjoint give, e.g. for  $+a[\theta^N - \zeta_{\alpha_a}^N]$ :

$$\log \mathbb{E}_{\tilde{h},\tilde{\lambda}}^{\nu_{\tilde{h}}^{N}} \left[ \exp \left[ a \int_{0}^{T} \left[ \theta^{N}(\eta_{t}) - \zeta_{\alpha_{a}}^{N}(\eta_{t}) \right] dt - \frac{1}{2} \int_{0}^{T} \mathcal{R}_{\tilde{h},\tilde{\lambda}}(\eta_{t}) dt \right] \right]$$

$$\leq T \sup_{f \geq 0: \nu_{h}^{N}(f)=1} \left\{ a \nu_{\tilde{h}}^{N} \left( f \left[ \theta^{N} - \zeta_{\alpha_{a}}^{N} \right] \right) - \frac{1}{2} \nu_{\tilde{h}}^{N} \left( f \mathcal{R}_{\tilde{h},\tilde{\lambda}} \right) - \frac{N^{2}}{2} \nu_{\tilde{h}}^{N} \left( \Gamma_{\tilde{h},\tilde{\lambda}}(\sqrt{f}) \right) + \frac{N^{2}}{2} \nu_{\tilde{h}}^{N} \left( N^{2} L_{\tilde{h},\tilde{h},\tilde{\lambda}}^{*} \mathbf{1} f \right) \right\}$$

$$\leq T \sup_{f \geq 0: \nu_{h}^{N}(f)=1} \left\{ a \nu_{\tilde{h}}^{N} \left( f \left[ \theta^{N} - \zeta_{\alpha_{a}}^{N} \right] \right) - \frac{N^{2}}{4} \nu_{\tilde{h}}^{N} \left( \Gamma_{\tilde{h},\tilde{\lambda}}(\sqrt{f}) \right) \right\}. \tag{3.103}$$

Recalling the definition (3.46) of  $\zeta_{\alpha_a}^N$  and the choice  $\alpha_a = (4a)^{-1}$ , the last supremum is bounded by 0. This bound and (3.101) conclude the proof.

## 4 A candidate for the approximate driven process

In this section, we start the proof of Theorem 2.4 by carrying out the first step in the program expounded in Section 2.4.

This involves computing the Radon-Nikodym derivative  $D_{\tilde{h},\lambda,E}^N := d\mathbb{P}_E^N/d\mathbb{P}_{\tilde{h},\lambda+E}^N|_T$  up to time T for  $\lambda, E \in \mathbb{R}$  and some  $\tilde{h} : \mathbb{T}^2 \to \mathbb{R}$  (these two dynamics are defined in (2.2)–(2.18)), then finding conditions on  $\tilde{h}$  that make  $\mathbb{P}_{\tilde{h},\lambda+E}^N$  a good candidate for the approximate driven process as defined in (2.15). This is carried out in Section 4.1.

In Section 4.2, we prove domination results between the current-biased dynamics and the dynamics  $\mathbb{P}^{N}_{\lambda+E}$ ,  $\mathbb{P}^{N}_{h,\lambda+E}$ , with *h* the bias identified in Section 4.1. These domination results ensure that the current-biased dynamics and  $\mathbb{P}^{N}_{h,\lambda+E}$  are indeed suitably close and reduce the proof of Theorem 2.4 to long-time decorrelation estimates at the macroscopic level, established in Section 5.

## 4.1 The Radon-Nikodym derivatives

Let  $\tilde{h} \in C^0(\mathbb{T}) \cap C^\infty([0,1])$  be fixed throughout. The function  $\tilde{h}$  is identified with  $(x,y) \mapsto \tilde{h}(x-y) \in C_D^\infty(\mathbb{T}^2)$  so that  $\Pi^N(\tilde{h})$  is a meaningful object. We always assume that  $\sigma \tilde{h}$  has leading eigenvalue strictly smaller than 1, so that the results of Section 3 on  $\mathbb{P}^N_{\tilde{h},\lambda+E}$  apply.

Here, we compute the Radon-Nikodym derivative  $D_{\tilde{h},\lambda+E}^N := d\mathbb{P}^N/d\mathbb{P}_{\tilde{h},\lambda+E}^N \Big|_T^{N,\lambda+E}$  for trajectories up to time T > 0, starting with  $D_{\lambda+E}^N := D_{0,\lambda+E}^N$ .

**Lemma 4.1.** Let  $\mu^N$  be a probability measure on  $\Omega_N$   $(N \in \mathbb{N}_{\geq 1})$ . Let  $\lambda, E \in \mathbb{R}$ , T > 0 and let  $A_{\lambda,E}$  be given by:

$$A_{\lambda,E}(\eta) = -\lambda(\lambda + 2E) \sum_{i \in \mathbb{T}_N} \bar{\eta}_i \bar{\eta}_{i+1}, \qquad \eta \in \Omega_N.$$
(4.1)

There is then  $\varepsilon_{\lambda,E}^N : \Omega_N \to \mathbb{R}$  with  $\sup_{\eta} |\varepsilon_{\lambda,E}^N| \leq C(\lambda,E)/N$  for each  $N \in \mathbb{N}_{\geq 1}$ , such that:

$$\mathbb{P}_{\lambda,E,T}^{curr,\mu^{N}}(\cdot) = \frac{\mathbb{E}_{\lambda+E}^{\mu^{N}} \Big[ \mathbf{1} \cdot \exp\left(\int_{0}^{T} A_{\lambda,E}(\eta_{t}) \, dt + \int_{0}^{T} \varepsilon_{\lambda,E}^{N}(\eta_{t}) \, dt\right) \Big]}{\mathbb{E}_{\lambda+E}^{\mu^{N}} \Big[ \exp\left(\int_{0}^{T} A_{\lambda,E}(\eta_{t}) \, dt + \int_{0}^{T} \varepsilon_{\lambda,E}^{N}(\eta_{t}) \, dt\right) \Big]}.$$
(4.2)

*Proof.* Let T > 0. The two dynamics  $\mathbb{P}_E^N$  and  $\mathbb{P}_{\lambda+E}^N$  are absolutely continuous with respect to one another on the space of trajectories on [0, T]. Fix  $\eta(\cdot) \in \mathcal{D}(\mathbb{R}_+, \Omega_N)$ . We write  $\eta_t$  rather than  $\eta(t)$  for the configuration at time t when there is no ambiguity with the occupation number  $\eta_i$  at a site  $i \in \mathbb{T}_N$ . The Radon-Nikodym derivative  $D_{\lambda,E}^N$  until time T reads (see the

proof of Proposition A.2.6 in [38]):

$$D_{\lambda,E}^{N}((\eta_{t})_{t\leq T}) = \frac{\mathrm{d}\mathbb{P}_{E}^{N}}{\mathrm{d}\mathbb{P}_{\lambda+E}^{N}} \Big|_{T}((\eta_{t})_{t\leq T})$$
  
= exp  $\Big[ -\lambda NQ_{T} + N^{2} \int_{0}^{T} \Big( \sum_{i\in\mathbb{T}_{N}} \eta_{i}(t)(1-\eta_{i+1}(t))e^{E/N}(e^{\lambda/N}-1) + \eta_{i+1}(t)(1-\eta_{i}(t))e^{-E/N}(e^{-\lambda/N}-1) \Big) dt \Big].$  (4.3)

Developing the exponential in the second member, we find that there is C > 0 and a function  $\varepsilon_{\lambda,E}^N : \Omega_N \to \mathbb{R}$  with  $\sup_{\eta \in \Omega_N} |\varepsilon_{\lambda,E}^N| \leq C/N$  such that:

$$D_{\lambda,E}^{N}\left((\eta_{t})_{t\leq T}\right) = \exp\left[-\lambda NQ_{T} + \frac{\lambda(\lambda+2E)}{2} \int_{0}^{T} \sum_{i\in\mathbb{T}_{N}} c(\eta_{t},\eta_{t}^{i,i+1}) dt + \int_{0}^{t} \varepsilon_{\lambda,E}^{N}(\eta_{t}) dt\right].$$
(4.4)

The jump rates  $c(\eta, \eta^{i,i+1})$  can be rewritten in terms of the variables  $\bar{\eta}_i = \eta_i - 1/2$   $(i \in \mathbb{T}_N)$ :

$$c(\eta, \eta^{i,i+1}) = 2\sigma - 2\bar{\eta}_i \bar{\eta}_{i+1}.$$
 (4.5)

Note that the total number of particles  $\sum_{i} \eta_i(t)$  is constant in time. The sum in (4.4) thus reads:

$$\int_{0}^{T} \sum_{i \in \mathbb{T}_{N}} c(\eta_{t}, \eta_{t}^{i, i+1}) dt = 2NT\sigma - 2 \int_{0}^{T} \sum_{i \in \mathbb{T}_{N}} \bar{\eta}_{i}(t) \bar{\eta}_{i+1}(t) dt.$$
(4.6)

Injecting the expression (4.6) in both numerator and denominator in the definition (2.7) of  $\mathbb{P}_{\lambda,E,T}^{curr,\mu^N}$  concludes the proof.

Lemma 4.1 expresses the current-biased dynamics  $\mathbb{P}_{\lambda,E,T}^{curr,\mu^N}$  in terms of the dynamics  $\mathbb{P}_{\lambda+E}^{\mu^N}$  which has the same macroscopic current. We now tune the long-range two-point correlation structure by considering the dynamics  $\mathbb{P}_{\tilde{h},\lambda+E}^{\mu^N}$  and optimising on  $\tilde{h}$  so that this dynamics is, loosely speaking, as close to  $\mathbb{P}_{\lambda,E,T}^{curr,\mu^N}$  as possible when N, then T are large.

**Proposition 4.2.** Let  $\mu^N$  be a probability measure on  $\Omega_N$  ( $N \in \mathbb{N}_{\geq 1}$ ). Let  $\lambda, E \in \mathbb{R}$  be sub-critical in the sense of Proposition 2.1 and  $h = h_{\lambda,E}$  be the associated solution of (2.21). Then, for each T > 0:

$$\mathbb{P}_{\lambda,E,T}^{curr,\mu^{N}}(\cdot) = \frac{\mathbb{E}_{h,\lambda+E}^{\mu^{N}} \Big[ \mathbf{1} \cdot \exp\left(-\Pi_{T}^{N}(h) + \Pi_{0}^{N}(h) + \int_{0}^{T} \varepsilon_{h,\lambda,E}^{N}(\eta_{t}) \, dt\right) \Big]}{\mathbb{E}_{h,\lambda+E}^{\mu^{N}} \Big[ \exp\left(-\Pi_{T}^{N}(h) + \Pi_{0}^{N}(h) + \int_{0}^{T} \varepsilon_{h,\lambda,E}^{N}(\eta_{t}) \, dt\right) \Big]},\tag{4.7}$$

where  $\varepsilon_{h,\lambda,E}^N$  is a function that has small time integral under both  $\mathbb{P}_{\lambda+E}^N$ ,  $\mathbb{P}_{h,\lambda+E}^N$  (started from suitable initial conditions):

$$\forall \delta > 0, \qquad \limsup_{N \to \infty} \mathbb{E}_{\lambda+E}^{\nu_{1/2}^{N}} \left[ \left| \int_{0}^{T} \varepsilon_{h,\lambda,E}^{N}(\eta_{t}) dt \right| \right] = 0, \\ \limsup_{N \to \infty} \mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}} \left[ \left| \int_{0}^{T} \varepsilon_{h,\lambda,E}^{N}(\eta_{t}) dt \right| \right] = 0.$$

$$(4.8)$$

*Proof.* Let T > 0. By Feynman-Kac formula, see Appendix A.7 in [38], the Radon-Nikodym derivative  $D_{\tilde{h},\lambda+E}^N = d\mathbb{P}_{\lambda+E}^N/d\mathbb{P}_{\tilde{h},\lambda+E}^N|_T$  up to time T reads, on each trajectory  $\eta_{\cdot} = (\eta_t)_{t \leq T}$ :

$$\log D^{N}_{\tilde{h},\lambda+E}(\eta) = -\Pi^{N}_{T}(\tilde{h}) + \Pi^{N}_{0}(\tilde{h}) + \int_{0}^{T} e^{-\Pi^{N}_{t}(\tilde{h})} N^{2} L_{\lambda+E} e^{\Pi^{N}_{t}(\tilde{h})} dt.$$
(4.9)

The computation of  $e^{-\Pi^N(\tilde{h})}N^2L_{\lambda+E}e^{\Pi^N(\tilde{h})}$  is very similar to that of  $N^2L_{h,h,\lambda+E}^*\mathbf{1}$  performed in Section 3.4, so we only give the result. For each  $\eta \in \Omega_N$ ,

$$e^{-\Pi^{N}(\tilde{h})} N^{2} L_{\lambda+E} e^{\Pi^{N}(\tilde{h})}(\eta) = \frac{1}{2} \sum_{i \in \mathbb{T}_{N}} \bar{\eta}_{i} \bar{\eta}_{i+1} \left( \partial_{1}^{N} \tilde{h}_{i+1,i} - \partial_{1}^{N} \tilde{h}_{i-1,i+1} \right)$$
(4.10)

$$+\frac{1}{2N}\sum_{j\neq\ell\in\mathbb{T}_N}\bar{\eta}_j\bar{\eta}_\ell \left[\mathbf{1}_{|j-\ell|>1}\Delta_1^N\tilde{h}_{j,\ell} + \frac{\sigma}{2N}\sum_{i\notin\{j-1,j,\ell-1,\ell\}}\partial_1^N\tilde{h}_{i,j}\partial_1^N\tilde{h}_{i,\ell}\right]$$
(4.11)

$$+ \frac{\sigma^2}{4N} \sum_{i \in \mathbb{T}_N} \sum_{j \notin \{i, i+1\}} \left( \partial_1^N \tilde{h}_{i,j} \right)^2 + \delta_{\tilde{h}, \lambda+E}^N(\eta) + O(N^{-1}), \tag{4.12}$$

with the error term  $O(N^{-1})$  uniform on the configuration. The quantity  $\delta^N_{\tilde{h},\lambda+E}(\eta)$  reads:

$$\delta_{\tilde{h},\lambda+E}^{N}(\eta) := \frac{(\lambda+E)\sigma}{N} \sum_{i\in\mathbb{T}_{N}} \bar{\eta}_{i} N \left[\tilde{h}_{i-1,i} - \tilde{h}_{i+1,i}\right] - \frac{1}{4N^{2}} \sum_{i\in\mathbb{T}_{N}} \bar{\eta}_{i} \bar{\eta}_{i+1} \sum_{j,\ell\notin\{i,i+1\}} \bar{\eta}_{j} \bar{\eta}_{\ell} \partial_{1}^{N} \tilde{h}_{i,j} \partial_{1}^{N} \tilde{h}_{i,\ell} - \frac{(\lambda+E)}{N} \sum_{i\in\mathbb{T}_{N}} \bar{\eta}_{i} \bar{\eta}_{i+1} \sum_{j\notin\{i,i+1\}} \bar{\eta}_{j} \partial_{1}^{N} \tilde{h}_{i,j}.$$
(4.13)

Let us show that  $\delta_{\tilde{h},\lambda+E}^N$  satisfies (4.8). The term  $\delta_{\tilde{h},\lambda+E}^N$  is similar to (3.61) estimated in Section 3.4. There we showed that, for any  $\alpha \in (0, 1/2]$  and any  $g \in C_D^{\infty}(\mathbb{T}^2)$  with  $\sigma g$  having leading eigenvalue strictly below 1, there is a function  $\zeta_{g,\tilde{h},\lambda+E,\alpha}^N$  such that, for any density ffor  $\nu_q^N$ :

$$\nu_g^N \left( f \delta_{\tilde{h}, \lambda+E}^N \right) \le \alpha N^2 \nu_g^N \left( \Gamma_{\tilde{h}, \tilde{\lambda}}(\sqrt{f}) \right) + \nu_g^N \left( f \zeta_{g, \tilde{h}, \lambda+E, \alpha}^N \right)$$
(4.14)

with, for some  $\gamma_{\alpha}, C_{\alpha} > 0$ :

$$\frac{1}{\gamma_{\alpha}}\log\nu_{g}^{N}\left(\exp\left[\gamma_{\alpha}\left|\zeta_{g,\tilde{h},\lambda+E,\alpha}^{N}\right|\right]\right) \leq \frac{C_{\alpha}}{N^{1/2}}, \qquad N \in \mathbb{N}_{\geq 1}.$$
(4.15)

Taking g = 0 and  $g = \tilde{h}$  this implies (4.8) for  $\mathbb{P}_{\lambda+E}^{\nu_{1/2}^{N}}, \mathbb{P}_{\tilde{h},\lambda+E}^{\nu_{\tilde{h}}^{N}}$  respectively by Corollary 3.7.

We now choose  $\tilde{h}$ . The bracket in (4.11) is a discretised differential operator acting on  $\tilde{h}$ , while the first line (4.10) plays the role of a condition on the normal derivative of  $\tilde{h}$  on the diagonal  $\{(x,x): x \in \mathbb{T}\}$  of  $\mathbb{T}^2$  (recall that, as  $\tilde{h}$  is symmetric,  $\partial_1 \tilde{h}(x,y) = \partial_2 \tilde{h}(y,x)$  for each  $x \neq y \in \mathbb{T}^2$ ), see (4.20) below. To determine the continuous analogue of the discrete

differential operator, let us replace discrete derivatives by their continuous counterparts using the following error estimates that follow from  $\tilde{h} \in C_D^{\infty}(\mathbb{T}^2)$ :

$$\sup_{N} \sup_{|i-j|>1} N \left| \Delta_1^N \tilde{h}_{i,j} - \Delta_1 \tilde{h}_{i,j} \right| < \infty, \qquad \sup_{N} \sup_{\substack{i,j \in \mathbb{T}_N \\ j \notin \{i,i+1\}}} N \left| \partial_1^N \tilde{h}_{i,j} - \partial_1 \tilde{h}_{i,j} \right| < \infty.$$
(4.16)

Let  $\varepsilon^N_{\tilde{h},\lambda,E}$  be given by:

$$\varepsilon_{\tilde{h},\lambda,E}^{N} = \delta_{\tilde{h},\lambda,E}^{N} + \varepsilon_{disc,\tilde{h}}^{N}, \qquad (4.17)$$

with  $\varepsilon^N_{disc,\tilde{h}}$  the discretisation error from switching to continuous partial derivatives:

$$\varepsilon_{disc,\tilde{h}}^{N}(\eta) = \frac{1}{2N} \sum_{i \in \mathbb{T}_{N}} \bar{\eta}_{i} \bar{\eta}_{i+1} N \left[ (\partial_{1}^{N} \tilde{h}_{i+1,i} - \partial_{1}^{N} \tilde{h}_{i-1,i+1}) - (\partial_{1} \tilde{h}_{i+,i} - \partial_{1} \tilde{h}_{i-,i}) \right] + \frac{1}{2N^{2}} \sum_{j,\ell:|j-\ell|>1} \bar{\eta}_{j} \bar{\eta}_{\ell} N \left[ \Delta_{1}^{N} \tilde{h}_{j,\ell} - \Delta_{1} \tilde{h}_{j,\ell} \right] + \frac{\sigma}{4N^{2}} \sum_{j\neq\ell} \bar{\eta}_{j} \bar{\eta}_{\ell} N \left[ \frac{1}{N} \sum_{i\notin\{j-1,j,\ell-1,\ell\}} \partial_{1}^{N} \tilde{h}_{i,j} \partial_{1}^{N} \tilde{h}_{i,\ell} - \int_{\mathbb{T}} \partial_{1} \tilde{h}(x-j/N) \partial_{1} \tilde{h}(x-\ell/N) \, dx \right].$$

$$(4.18)$$

The exponential moment under  $\nu_{\tilde{h}}^N, \nu_{1/2}^N$  of each term above is controlled through Proposition A.4. Corollary 3.7 thus shows that  $\varepsilon_{\tilde{h},\lambda,E}^N$ , also satisfies (4.8) (replacing *h* there by  $\tilde{h}$ ) and:

$$e^{-\Pi^{N}(\tilde{h})}N^{2}L_{\lambda+E}e^{\Pi^{N}(\tilde{h})}(\eta) = \frac{\sigma^{2}}{4N}\sum_{i\in\mathbb{T}_{N}}\sum_{j\notin\{i,i+1\}}\left(\partial_{1}^{N}\tilde{h}_{i,j}\right)^{2} + \frac{1}{2}\sum_{i\in\mathbb{T}_{N}}\bar{\eta}_{i}\bar{\eta}_{i+1}\left(\partial_{1}\tilde{h}_{i+,i} - \partial_{1}^{N}\tilde{h}_{i-,i}\right) \\ + 2\Pi^{N}\left(\Delta_{1}\tilde{h} + (x,y)\mapsto\frac{\sigma}{2}\int_{\mathbb{T}}\partial_{1}\tilde{h}(z,x)\partial_{1}\tilde{h}(z,y)\,dz\right) + \varepsilon^{N}_{\tilde{h},\lambda,E}(\eta).$$
(4.19)

Our goal is to now choose  $\tilde{h}$  such that the two-point correlation term in (4.19) precisely cancels the term  $A_{\lambda,E} = -\lambda(\lambda + 2E) \sum_i \bar{\eta}_i \bar{\eta}_{i+1}$  appearing in Lemma 4.1. This fixes the partial derivative of  $\tilde{h}$  across the diagonal:

$$\forall i \in \mathbb{T}_N, \qquad \partial_1 \tilde{h}_{i_+,i} - \partial_1 \tilde{h}_{i_-,i} = h'(0_+) - h'(1_-) = 2\lambda(\lambda + 2E). \tag{4.20}$$

The remaining two-point correlation terms in (4.19) also need to vanish for the approximation of the current-biased dynamics by  $\mathbb{P}^{N}_{\tilde{h},\lambda+E}$  to be better than that by  $\mathbb{P}^{N}_{\lambda+E}$ . Thus:

$$\Delta_1 \tilde{h} + (x, y) \mapsto \frac{\sigma}{2} \int_{\mathbb{T}} \partial_1 \tilde{h}(z, x) \partial_1 \tilde{h}(z, y) = 0, \qquad x \neq y \in \mathbb{T}.$$
(4.21)

As  $\tilde{h}(x,y) = \tilde{h}(x-y)$  by assumption,  $\partial_1 \tilde{h}(z,x) = \tilde{h}'(z-x) = -\tilde{h}'(x-z)$   $(x \neq z)$  and we see that (4.21) is precisely the ordinary differential equation appearing in Proposition 2.1:

$$\tilde{h}''(x) - \sigma \int_{\mathbb{T}} \tilde{h}'(x-y)\tilde{h}'(y) \, dy = 0, \qquad x \in (0,1).$$
(4.22)

From this observation and (4.20) it follows that  $\tilde{h}$  is the function h of Proposition 2.1. Note that  $\sigma h$  indeed has leading eigenvalue strictly below 1 as  $\lambda, E$  are sub-critical by assumption. As the last term in the first line of (4.19) is configuration-independent, the proof of Proposition 4.2 is concluded.

## 4.2 A key domination result

The expression (4.7) involves the exponential of terms that we know to have average bounded with N under  $\mathbb{P}^{N}_{h,\lambda+E}$  when N, then T are large. Since these terms appear inside exponentials, we need to make sure that their exponential moments are also bounded uniformly in N, T. Unbounded exponential moments would mean that that  $\mathbb{P}^{N}_{h,\lambda+E}$  is in fact not a good candidate for an approximate driven process. Proposition 4.4 shows that the exponential moments are indeed bounded, at least for parameters  $\lambda, E$  in a small enough subset of the sub-critical region of Proposition 2.1. This corresponds to Step 2 in the program outlined in Section 2.4.

The main ingredient to establish Proposition 4.4 is the following domination result. It is the only result of the present paper for which we need to restrict the range of sub-critical  $\lambda, E$ , see the discussion in Section 2.5.

**Proposition 4.3.** Let  $\lambda, E \in \mathbb{R}$  be sub-critical parameters as in Proposition 2.1 and let T > 0. If  $\lambda(\lambda + 2E) \geq 0$  and is small enough (independently of T), there are constants  $\kappa \in (0,1), C > 0$  depending only on  $h, \lambda, E$  such that, for any sequence  $(\mathcal{O}_N)_N$  of events involving the dynamics up to time T:

$$\limsup_{N \to \infty} \mathbb{P}_{\lambda,T}^{curr,\nu_{1/2}^{N}}(\mathcal{O}_{N}) \leq C \limsup_{N \to \infty} \mathbb{P}_{h,\lambda,E}^{\nu_{h}^{N}}(\mathcal{O}_{N})^{\kappa}.$$
(4.23)

Proposition 4.3 allows for an expression of the current-biased dynamics in terms of expectations of bounded observables under  $\mathbb{P}_{h,\lambda+E}^N$  as follows. Let M > 0 and define  $F_M$  by:

$$F_M(x) = \begin{cases} e^x & \text{if } x \le M, \\ 0 & \text{if } x \ge 2M, \\ \text{smooth and bounded by } e^M & \text{otherwise,} \end{cases} \quad (4.24)$$

**Proposition 4.4.** Let  $\lambda, E \in \mathbb{R}$  be sub-critical so that  $\sigma h$  has leading eigenvalue strictly below 1. Assume that  $\lambda, E$  are such that the domination bound of Proposition 4.3 holds with exponent  $\kappa > 0$ .

There are then C, C' > 0 and  $\kappa' \in (0,1)$  depending only on  $h, \lambda, E$  such that, for each T > 0, each M > C' and each continuous and bounded  $G : \mathcal{D}(\mathbb{R}_+, \mathcal{S}'(\mathbb{T})) \to \mathbb{R}$ :

$$\lim_{N \to \infty} \sup_{N \to \infty} \left| \mathbb{E}_{\lambda, E, T}^{curr, \nu_{1/2}^{N}} \left[ G\left( (Y_{s}^{N})_{s \leq T} \right) \right] - \frac{\mathbb{E}_{h, \lambda + E}^{\nu_{h}^{N}} \left[ G\left( (Y_{s}^{N})_{s \leq T} \right) F_{M} \left( -\Pi_{0}^{N}(h) \right) F_{M} \left( -\Pi_{T}^{N}(h) \right) \right]}{\mathbb{E}_{h, \lambda + E}^{\nu_{h}^{N}} \left[ F_{M} \left( -\Pi_{0}^{N}(h) \right) F_{M} \left( -\Pi_{T}^{N}(h) \right) \right]} \right| \\ \leq \frac{C \|G\|_{\infty}}{M^{\kappa'}}. \tag{4.25}$$

#### 4.2.1 Proof of Proposition 4.4 assuming Proposition 4.3

*Proof.* Let  $T, M, \delta > 0, t \ge 0$ , and let G be as in Proposition 4.4. Introduce the sets:

$$S_t(M) := \left\{ \left| \Pi_t^N(h) \right| \le M \right\}, \qquad U_T(\delta) := \left\{ \left| \int_0^T \varepsilon_{h,\lambda,E}^N(\eta_t) \, dt \right| \le \delta \right\}, \tag{4.26}$$

with  $\varepsilon_{h,\lambda,E}^N$  given in Proposition 4.2. Theorem 2.7, the entropy inequality and the concentration bounds of Proposition A.4 give a control on  $S_t(M)^c$ : for some  $\gamma, C > 0$  depending only on  $h, \lambda, E$ ,

$$\mathbb{P}_{h,\lambda+E}^{\nu_h^N}(|\Pi_t^N(h)| \ge M) \le \frac{1}{M} \mathbb{E}_{h,\lambda+E}^{\nu_h^N}[|\Pi_t^N(h)|] \le \frac{\mathbf{H}^N(f_t)}{M\gamma} + \frac{1}{M\gamma} \log \nu_h^N\left(e^{\gamma^{-1}|\Pi^N(h)|}\right) \\
\le \frac{Ce^{Ct}}{N^{1/2}} + \frac{C}{M}.$$
(4.27)

Proposition 4.2 shows that  $\varepsilon_{h,\lambda,E}^N$  has small time integral, thus for a different constant C > 0:

$$\limsup_{N \to \infty} \mathbb{P}_{h,\lambda+E}^{\nu_h^N} \Big( (V_T(M,\delta))^c \Big) \le \frac{C}{M}, \quad V_T(M,\delta) := S_0(M) \cap S_T(M) \cap U_T(\delta).$$
(4.28)

The domination bound of Proposition 4.3 then implies:

$$\lim_{N \to \infty} \sup_{N \to \infty} \left| \mathbb{E}_{\lambda, E, T}^{curr, \nu_{1/2}^{N}} \left[ G \right] - \mathbb{E}_{\lambda, E, T}^{curr, \nu_{1/2}^{N}} \left[ G \middle| V_{T}(M, \delta) \right] \right| \leq \frac{C \|G\|_{\infty}}{M^{\kappa}}.$$
(4.29)

It is therefore enough to prove Proposition 4.4 for the current-biased dynamics conditioned on  $V_T(M, \delta)$ . The proof at this point is elementary but tedious. For short, write:

$$\mathcal{R}_T^N = \mathcal{R}_T^N\big((\eta_t)_{t \le T}\big) := -\Pi_T^N(h) - \Pi_0^N(h) + \int_0^T \varepsilon_{h,\lambda,E}^N(\eta_t) \, dt.$$
(4.30)

From the expression (4.7) of the current-biased dynamics in terms of  $\mathbb{P}_{h,\lambda+E}^N$ , we can write:

$$\mathbb{E}_{\lambda,E,T}^{curr,\nu_{1/2}^{N}}\left[G\big|V_{T}(M,\delta)\right] = \left(\mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}}\left[\mathbf{1}_{V_{T}(M,\delta)}e^{\mathcal{R}_{T}^{N}}\right]\right)^{-1}\mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}}\left[G\mathbf{1}_{V_{T}(M,\delta)}e^{\mathcal{R}_{T}^{N}}\right].$$
(4.31)

Recall definition (4.26) of  $U_T(\delta) \subset V_T(M, \delta)$ . On  $U_T(\delta)$ , the time integral of  $\varepsilon_{h,\lambda,E}^N$  is bounded by  $\delta$ . Moreover, on  $S_t(M)$ ,  $F_M(-\Pi_t^N(h)) = e^{-\Pi_t^N(h)}$ ,  $t \in \{0, T\}$ . Rewriting also the exponential containing  $\varepsilon_{h,\lambda,E}^N$  using  $e^x = 1 + e^x - 1$ , we find after elementary computations:

$$\left| \mathbb{E}_{\lambda,E,T}^{curr,\nu_{1/2}^{N}} \left[ G \middle| V_{T}(M,\delta) \right] - \frac{\mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}} \left[ G \mathbf{1}_{V_{T}(M,\delta)} F_{M} \left( -\Pi_{0}^{N}(h) \right) F_{M} \left( -\Pi_{T}^{N}(h) \right) \right]}{\mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}} \left[ \mathbf{1}_{V_{T}(M,\delta)} F_{M} \left( -\Pi_{0}^{N}(h) \right) F_{M} \left( -\Pi_{T}^{N}(h) \right) \right]} \right| \leq \|G\|_{\infty} c(\delta),$$

$$(4.32)$$

with  $c(\delta) = 2(e^{\delta} - 1)$ . To conclude the proof, it remains to remove the indicator function in the last equation. We start with the numerator, the denominator being similar.

Recall from Proposition 2.1 that h is either a negative kernel when  $\lambda(\lambda + 2E) \geq 0$ or a positive kernel if  $\lambda(\lambda + 2E) \leq 0$ . Recall also that we always work with sub-critical  $\lambda, E$  or equivalently that the leading eigenvalue of  $\sigma h$  strictly below 1 (see Proposition 2.1), say smaller than 1 - r for  $r = r(\lambda, E) > 0$ . Recall finally that the domination bound of Proposition 4.3 is assumed to hold for  $\lambda, E$ . Although this domination bound is established only for  $\lambda(\lambda + 2E) \geq 0$  corresponding to a negative kernel h, we conjecture it should hold throughout the sub-critical regime (see the discussion in Section 2.5) and therefore carry out the proof of (4.4) also in the case  $\lambda(\lambda + 2E) < 0$ .

Assume first that h is a positive kernel. This implies (recall  $\bar{\eta}_i^2 = \sigma = 1/4$  for each  $i \in \mathbb{T}_N$ ):

$$-\Pi^{N}(h) = -\frac{1}{4N} \sum_{i \neq j} \bar{\eta}_{i} \bar{\eta}_{j} h_{i-j} = -\frac{1}{4N} \sum_{i,j} \bar{\eta}_{i} \bar{\eta}_{j} h_{i-j} + \frac{h_{0}\sigma}{4} \le \frac{h_{0}\sigma}{4} \le \frac{\|h\|_{\infty}}{16}.$$
 (4.33)

From  $F_M(x) \leq e^x$  for each  $x \in \mathbb{R}$ , we get:

$$F_M(-\Pi_T(h)) \le e^{-\Pi_T(h)} \le e^{\|h\|_{\infty}/16}.$$
 (4.34)

The same holds for  $\Pi_0^N$ . Thus:

$$\begin{aligned} \left| \mathbb{E}_{h,\lambda+E}^{\nu_h^N} \Big[ G \mathbf{1}_{V_T(M,\delta)} F_M \Big( -\Pi_0^N(h) \Big) F_M \Big( -\Pi_T^N(h) \Big) \Big] \\ &- \mathbb{E}_{h,\lambda+E}^{\nu_h^N} \Big[ G F_M \Big( -\Pi_0^N(h) \Big) F_M \Big( -\Pi_T^N(h) \Big) \Big] \right| \\ &= \left| \mathbb{E}_{h,\lambda+E}^{\nu_h^N} \Big[ G \mathbf{1}_{V_T(M,\delta)^c} F_M \Big( -\Pi_0^N(h) \Big) F_M \Big( -\Pi_T^N(h) \Big) \Big] \right| \\ &\leq \|G\|_{\infty} e^{\|h\|_{\infty}/8} \mathbb{P}_{h,\lambda+E}^{\nu_h^N} \Big( V_T(M,\delta)^c \Big) \leq \frac{C(h,\lambda,E) \|G\|_{\infty} (1+o_N(1))}{M}. \end{aligned}$$
(4.35)

Assume now that h is a negative kernel. By the explicit formula for h in Proposition 2.1, we know that  $-\sigma h$  has eigenvalues bounded from above. Let  $r_h$  be the spectral radius of  $-\sigma h$ . Hölder inequality with exponents  $1 + (2r_h)^{-1}$ ,  $2r_h + 1$  and Cauchy-Schwarz inequality give:

$$\mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}} \left[ G\mathbf{1}_{V_{T}(M,\delta)^{c}} F_{M} \left( -\Pi_{0}^{N}(h) \right) F_{M} \left( -\Pi_{T}^{N}(h) \right) \right] \tag{4.36}$$

$$\leq \|G\|_{\infty} \mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}} \left[ F_{M} \left( -\Pi_{0}^{N}(h) \right)^{1+\frac{1}{2r_{h}}} F_{M} \left( -\Pi_{T}^{N}(h) \right)^{1+\frac{1}{2r_{h}}} \right]^{\frac{2r_{h}}{2r_{h}+1}} \mathbb{P}_{h,\lambda+E}^{\nu_{h}^{N}} \left( V_{T}(M,\delta)^{c} \right)^{\frac{1}{2r_{h}+1}}$$

$$\leq \frac{C(h,\lambda,E) \|G\|_{\infty}}{M^{\frac{1}{2r_{h}+1}}} \mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}} \left[ F_{M} \left( -\Pi_{0}^{N}(h) \right)^{2+\frac{1}{r_{h}}} \right]^{\frac{r_{h}}{2r_{h}+1}} \mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}} \left[ F_{M} \left( -\Pi_{T}^{N}(h) \right)^{2+\frac{1}{r_{h}}} \right]^{\frac{r_{h}}{2r_{h}+1}}.$$

We claim that the last two expectations are bounded independently of N, M, T. Indeed, as  $F_M(x) \leq e^x$ , the exponential moment involving  $\Pi_0(h)$  satisfies:

$$\mathbb{E}_{h,\lambda+E}^{\nu_h^N} \left[ F_M \left( -\Pi_0^N(h) \right)^{2+\frac{1}{r_h}} \right] \leq \nu_h^N \left[ \exp\left[ -\left(2+\frac{1}{r_h}\right) \Pi_0^N(h) \right] \right] \\
\leq \frac{1}{\mathcal{Z}_h^N} \nu_{1/2}^N \left[ \exp\left[ -\frac{1}{r_h} \Pi_0^N(h) \right] \right].$$
(4.37)

By definition of  $r_h$ ,  $-(\sigma/2r_h)h$  has leading eigenvalue at most 1/2. By Lemma A.1, the exponential moment in (4.37) is therefore bounded with N. Lemma A.1 also gives  $\mathcal{Z}_h^N \geq 1$ .

Consider now  $\Pi_T^N$ . As  $F_M$  is a bounded function, Pinsker's inequality in Proposition 3.2 gives:

$$\limsup_{N \to \infty} \left| \mathbb{E}_{h,\lambda+E}^{\nu_h^N} \Big[ F_M \big( -\Pi_T^N(h) \big)^{2+\frac{1}{2r_h}} \Big] - \nu_h^N \Big[ F_M \big( -\Pi^N(h) \big)^{2+\frac{1}{2r_h}} \Big] \right| = 0.$$
(4.38)

The bound is then the same as in (4.37).

Consider now the denominator. Recalling that  $F_M(-\Pi_t(h)) = e^{-\Pi_t(h)}$  on  $S_t(M)$ , it is enough to prove that, for M larger than some  $C(h, \lambda, E) > 0$ :

$$\inf_{M>C(h,\lambda,E)} \inf_{T>0} \liminf_{N\to\infty} \mathbb{E}_{h,\lambda+E}^{\nu_h^N} \Big[ \mathbf{1}_{V_T(M,\delta)} \exp\left[ -\Pi_0^N(h) - \Pi_T^N(h) \right] \Big] > 0.$$
(4.39)

Write  $\mathbf{1}_{V_T(\delta)} = 1 - \mathbf{1}_{V_T(\delta)^c}$ . The moment bound in Proposition 3.2, Jensen inequality and the convergence/stationarity results of Theorem 2.5 give the existence of c > 0 independent of N, T, M such that:

$$\lim_{N \to \infty} \inf_{h, \lambda+E} \left[ \exp\left[ -\Pi_0^N(h) - \Pi_T^N(h) \right] \right] \\
\geq 2 \liminf_{N \to \infty} \exp\left[ -\mathbb{E}_{h, \lambda+E}^{\nu_h^N} \left[ \Pi_0^N(h) + \Pi_T^N(h) \right] \right] \geq c > 0. \quad (4.40)$$

On the other hand, we know by (4.29) and (4.32) that:

$$\sup_{T>0} \limsup_{N\to\infty} \mathbb{E}_{h,\lambda+E}^{\nu_h^N} \Big[ \mathbf{1}_{V_T(M,\delta)^c} \exp\left[-\Pi_0^N(h) - \Pi_T^N(h)\right] \Big] \\ \leq \frac{C}{M^{\frac{1}{2r_h+1}}} \mathbf{1}_{\lambda(\lambda+2E)>0} + \frac{C}{M} \mathbf{1}_{\lambda(\lambda+2E)\leq 0}.$$
(4.41)

Putting together this bound and (4.29) and (4.32), we conclude that for all M larger than some  $C = C(h, \lambda, E) > 0$ :

$$\lim_{N \to \infty} \sup_{N \to \infty} \left| \mathbb{E}_{\lambda, E, T}^{curr, \nu_{1/2}^{N}} \left[ G \right] - \frac{\mathbb{E}_{h, \lambda+E}^{\nu_{h}^{N}} \left[ GF_{M} \left( -\Pi_{0}^{N}(h) \right) F_{M} \left( -\Pi_{T}^{N}(h) \right) \right]}{\mathbb{E}_{h, \lambda+E}^{\nu_{h}^{N}} \left[ F_{M} \left( -\Pi_{0}^{N}(h) \right) F_{M} \left( -\Pi_{T}^{N}(h) \right) \right]} \right| \\
\leq \|G\|_{\infty} c(\delta) + \frac{C\|G\|_{\infty}}{M^{\kappa'}},$$
(4.42)

where  $\kappa' = \min\{(2r_h + 1)^{-1}, \kappa\}$  if *h* is a negative kernel,  $\kappa' = \min\{1, \kappa\}$  if *h* is a positive kernel. Since  $c(\delta) = 2(e^{\delta} - 1)$ , taking  $\delta = M^{-\kappa'}$  concludes the proof of Proposition 4.4 assuming the domination bound of Proposition 4.3.

#### 4.2.2 Proof of Proposition 4.3: the domination bound

The time-independent domination bound of Proposition 4.3 is proven by first establishing a weaker domination result, Proposition 4.5, in terms of the dynamics  $\mathbb{P}^N_{\lambda+E}$  that has the correct macroscopic current but the wrong correlation structure. This first domination bound provides a control of the current-biased dynamics to order 1 in N, but not yet uniform in time, reflecting the fact that the correlation structure is not yet properly tuned.

**Proposition 4.5.** Let  $\lambda, E \in \mathbb{R}$  be sub-critical so that  $\sigma h$  has leading eigenvalue strictly below 1 (recall Proposition 2.1). Recall the definition of  $A_{\lambda,E}$  in (4.1).

Then, for each s > 0, if  $\lambda(\lambda + 2E) \ge 0$  and is small enough depending on s, there is  $C(s, \lambda, E) > 0$  satisfying:

$$\forall T > 0, \qquad \sup_{N \ge 1} \mathbb{E}_{\lambda+E}^{\nu_{1/2}^{N}} \left[ \exp\left[s \int_{0}^{T} A_{\lambda,E}(\eta_t) \, dt\right] \right] \le e^{C(s,\lambda,E)T}. \tag{4.43}$$

This implies that there is  $C(h, \lambda, E) > 0$  such that, for any event  $\mathcal{O}$  depending on the dynamics up to time T:

$$\mathbb{P}_{\lambda,E,T}^{curr,\nu_{1/2}^{N}}(\mathcal{O}) \leq e^{C(h,\lambda,E)T} \mathbb{P}_{\lambda+E}^{\nu_{1/2}^{N}}(\mathcal{O})^{1/2}, \qquad N \in \mathbb{N}_{\geq 1}.$$
(4.44)

*Proof.* Let us first prove (4.44) assuming (4.43). Fix  $\mathcal{O}$  as in the proposition. Lemma 4.1 relates the current-biased dynamics starting from  $\nu_{1/2}^N$  to  $\mathbb{P}_{\lambda+E}^{\nu_{1/2}^N}$ :

$$\mathbb{P}_{\lambda,E,T}^{curr,\nu_{1/2}^{N}}(\mathcal{O}) = \frac{\mathbb{E}_{\lambda+E}^{\nu_{1/2}^{N}} \Big[ \mathbf{1}_{\mathcal{O}} \exp\left[\int_{0}^{T} A_{\lambda,E}(\eta_{t}) dt\right] \Big]}{\mathbb{E}_{\lambda+E}^{\nu_{1/2}^{N}} \Big[ \exp\left[\int_{0}^{T} A_{\lambda,E}(\eta_{t}) dt\right] \Big]}.$$
(4.45)

Applying Cauchy-Schwarz inequality to the numerator and the bound (4.43), then Jensen inequality to the denominator, one finds:

$$\mathbb{P}_{\lambda,E,T}^{curr,\nu_{1/2}^{N}}(\mathcal{O}) \leq \frac{e^{C(s=2,\lambda,E)T/2} \nu_{1/2}^{N} \left(\exp[2\Pi^{N}(h)]\right)^{1/2}}{\exp\mathbb{E}_{\lambda+E}^{\nu_{1/2}^{N}} \left[\int_{0}^{T} A_{\lambda,E}(\eta_{t}) dt\right]} \mathbb{P}_{\lambda+E}^{N,\nu_{1/2}^{N}}(\mathcal{O})^{1/2}.$$
(4.46)

Since  $\sigma h$  has leading eigenvalue strictly below 1, Lemma A.1 gives:

$$\sup_{N} \nu_{1/2}^{N}(\exp[2\Pi^{N}(h)]) < \infty.$$
(4.47)

The numerator of (4.46) thus has the desired form (4.44). Moreover, recalling definition (4.1) of  $A_{\lambda,E}$  and the fact that  $\nu_{1/2}^N$  is invariant for  $\mathbb{P}^N_{\lambda+E}$  and product,

$$\mathbb{E}_{\lambda+E}^{\nu_{1/2}^{N}} \left[ \int_{0}^{T} A_{\lambda,E}(\eta_{t}) dt \right] = T \nu_{1/2}^{N} \left[ A_{\lambda,E} \right] = 0.$$
(4.48)

This concludes the proof of (4.44) assuming (4.43), which we now prove. For each  $0 \le m \le N$ , let  $U_m$  denote the uniform measure on the set  $\Omega_{N,m}$  of configurations with m particles. One can check that the measure  $U_m$  is invariant for the dynamics  $\mathbb{P}^N_{\lambda+E}$ . Let us rewrite  $A_{\lambda,E}$ , defined in (4.1), in terms of quantities with vanishing average under  $U_m$ . Define, for each mand  $\eta \in \Omega_{N,m}$ :

$$A^m_{\lambda,E}(\eta) := -\lambda(\lambda + 2E) \sum_{i \in \mathbb{T}_N} \bar{\eta}^m_i \bar{\eta}^m_{i+1}, \qquad \bar{\eta}^m_{\cdot} = \eta_{\cdot} - \frac{m}{N}.$$
(4.49)

Then, on  $\Omega_{N,m}$ :

$$A_{\lambda,E} = A_{\lambda,E}^m - \lambda(\lambda + 2E)N\left(\frac{m}{N} - \frac{1}{2}\right)^2 \le A_{\lambda,E}^m,\tag{4.50}$$

where the inequality follows from the assumption  $\lambda(\lambda + 2E) \ge 0$ . As a result, the exponential moment to estimate satisfies, for each s > 0:

$$\mathbb{E}_{\lambda+E}^{\nu_{1/2}^{N}}\left(\exp\left[s\int_{0}^{T}A_{\lambda,E}(\eta_{t})\,dt\right]\right) \leq \max_{0\leq m\leq N}\mathbb{E}_{\lambda+E}^{U_{m}}\left(\exp\left[s\int_{0}^{T}A_{\lambda,E}^{m}(\eta_{t})\,dt\right]\right).\tag{4.51}$$

Fix  $0 \le m \le N$  henceforth, and let us estimate the above exponential moment. Let  $\Gamma_{\lambda+E}$  denote the carré du champ operator associated with  $L_{\lambda+E}$ :

$$\Gamma_{\lambda+E}(g)(\eta) := \frac{1}{2} \sum_{i \in \mathbb{T}_N} c_{\lambda+E}(\eta, \eta^{i,i+1}) \left[ u(\eta^{i,i+1}) - u(\eta) \right]^2, \qquad \eta \in \Omega_N, \quad u : \Omega_N \to \mathbb{R}.$$
(4.52)

Feynman-Kac inequality (see Lemma 3.5) and the invariance of  $U_m$  give:

$$\frac{1}{T}\log \mathbb{E}_{\lambda+E}^{U_m} \Big[ \exp\left[s \int_0^T A_{\lambda,E}^m(\eta_t) \, dt\right] \Big] \\
\leq \sup_{f \ge 0: U_m(f)=1} \Big\{ U_m \big(f s A_{\lambda,E}\big) - \frac{N^2}{2} \, U_m \big(\Gamma_{\lambda+E}(\sqrt{f})\big) \Big\}.$$
(4.53)

An elementary computation shows that, for each density f for  $U_m$ :

$$U_m(\Gamma_{\lambda+E}(\sqrt{f})) = \cosh\left((\lambda+E)/N\right) D_{ex}^m(\sqrt{f}), \qquad D_{ex}^m(\sqrt{f}) := -U_m(\sqrt{f}L_0\sqrt{f}).$$
(4.54)

The supremum in (4.53) therefore reduces to:

$$\sup_{f \ge 0: U_m(f)=1} \left\{ \nu_{1/2}^N \left( f s A_{\lambda, E}^m \right) - \frac{N^2}{2} \cosh\left( (\lambda + E) / N \right) D_{ex}^m(\sqrt{f}) \right\}.$$
(4.55)

Fix a density  $f \ge 0$  for  $U_m$ . We aim to prove that this supremum is bounded by some  $C(s, \lambda, E) > 0$  provided  $\lambda(\lambda + 2E) \ge 0$  is small enough. To do so, we smooth out  $A_{\lambda,E}^m$  using the carre du champ in (4.55) as was done in the proof of Lemma 3.8. Notice first that an integration by parts gives, for each  $\eta \in \Omega_N$  and  $i \in \mathbb{T}_N$ :

$$\bar{\eta}_{i+1}^m = \frac{1}{N-1} \sum_{j \neq i} \bar{\eta}_j^m + \sum_{j \neq i, i-1} (\bar{\eta}_j^m - \bar{\eta}_{j+1}^m) \phi_{N-1}(j-i), \qquad (4.56)$$

with:

$$\phi_{N-1}(a) := \frac{N - 1 - (a \mod N)}{N - 1}, \qquad a \in \mathbb{Z}.$$
(4.57)

Fix a density f for  $\nu_{1/2}^N$ . Using (4.56) on  $A_{\lambda,E}^m$  for each  $i \in \mathbb{T}_N$  and the integration by parts Lemma B.1 (which applies to  $U_m$  by Remark B.2) with  $\alpha = 1/2$ ,  $\tilde{h} = 0$  and  $\tilde{\lambda} = \lambda + E$ , we find that there is a numerical constant c > 0 such that:

$$U_m(fsA_{\lambda,E}^m) \leq \frac{N^2}{4} D_{ex}^m(\sqrt{f}) - \frac{s\lambda(\lambda+2E)}{N-1} U_m\left(f\sum_{i\neq j} \bar{\eta}_i^m \bar{\eta}_j^m\right) + \frac{cs^2\lambda^2(\lambda+2E)^2}{N} U_m\left[f\sum_{j\in\mathbb{T}_N} \left(\frac{1}{\sqrt{N}}\sum_{i\neq j} \phi_{N-1}(j-i)\bar{\eta}_i^m\right)^2\right].$$
(4.58)

Adding and removing the (deterministic) i = j term in the first sum and using  $\lambda(\lambda + 2E) \ge 0$ , we get:

$$\frac{s\lambda(\lambda+2E)}{N-1}U_m\left(f\sum_{i\neq j}\bar{\eta}_i^m\bar{\eta}_j^m\right) \le s\lambda(\lambda+2E)\frac{N}{N-1}\frac{m}{N}\left(1-\frac{m}{N}\right).$$
(4.59)

Since  $\rho(1-\rho) \leq 1/4$  for any  $\rho \in [0,1]$ , the bound (4.43) will be proven, recalling (4.55), if we can show that, for  $\lambda(\lambda + 2E)$  small enough:

$$\frac{cs^2\lambda^2(\lambda+2E)^2}{N}U_m\left(f\sum_{j\in\mathbb{T}_N}\left(\frac{1}{\sqrt{N}}\sum_{i\neq j}\phi_{N-1}(j-i)\bar{\eta}_i^m\right)^2\right) \le C(s,\lambda,E) + \frac{N^2}{4}D_{ex}^m(\sqrt{f}), \quad (4.60)$$

where  $C(s, \lambda, E) > 0$  is independent of f, m, N.

To prove (4.60), we use the entropy inequality for each  $j \in \mathbb{T}_N$  to get, for any  $\gamma > 0$ :

$$\frac{cs^{2}\lambda^{2}(\lambda+2E)^{2}}{N}U_{m}\left(f\sum_{j\in\mathbb{T}_{N}}\left(\frac{1}{\sqrt{N}}\sum_{i\neq j}\phi_{N-1}(j-i)\bar{\eta}_{i}^{m}\right)^{2}\right) \\
\leq \frac{U_{m}(f\log f)}{\gamma} \\
+ \frac{1}{\gamma N}\sum_{j\in\mathbb{T}_{N}}\log U_{m}\left(\exp\left(\gamma cs^{2}\lambda^{2}(\lambda+2E)^{2}\left(\frac{1}{\sqrt{N}}\sum_{i\neq j}\phi_{N-1}(j-i)\bar{\eta}_{i}^{m}\right)^{2}\right]\right). \quad (4.61)$$

The entropic term can be estimated with the log-Sobolev inequality of Lemma A.5: for  $\gamma$  larger than some  $\gamma_0 > 0$  independent of m,

$$\frac{U_m(f\log f)}{\gamma} \le \frac{N^2}{4} D_{ex}^m(\sqrt{f}). \tag{4.62}$$

On the other hand, the function  $\phi_{N-1}(j-\cdot)$  is Lipschitz uniformly in  $j \in \mathbb{T}_N$ , in the sense:

$$\sup_{N \ge 2} \max_{i,j \in \mathbb{T}_N} N \left| \phi_{N-1}(j-i-1) - \phi_{N-1}(j-i) \right| = 2.$$
(4.63)

Lemma A.6 then implies that, for  $\lambda(\lambda + 2E)$  smaller than a constant depending only on  $s, \gamma$ :

$$\frac{1}{\gamma N} \sum_{j \in \mathbb{T}_N} \log U_m \left[ \exp\left(\gamma c s^2 \lambda^2 (\lambda + 2E)^2 \left(\frac{1}{\sqrt{N}} \sum_{i \neq j} \phi_{N-1} (j-i) \bar{\eta}_i^m\right)^2 \right] \right) \le \frac{3}{\gamma}.$$
(4.64)

Putting the last estimate together with (4.58)-(4.60)-(4.62) inside (4.55) concludes the proof of Proposition 4.5.

We can now use Proposition 4.5, together with the finely tuned correlation structure of  $\mathbb{P}^{N}_{h,\lambda+E}$  to prove the time-independent domination bound of Proposition 4.3.

Proof of Proposition 4.3. Let T > 0 and  $(\mathcal{O}_N)_N$  be a sequence of measurable sets involving the dynamics up to time T only. Recall the expression of  $\mathbb{P}_{\lambda,E,h}^{curr,\nu_{1/2}^N}$  in terms of  $\mathbb{P}_{h,\lambda+E}^{\nu_h^N}$  from Proposition 4.2:

$$\mathbb{P}_{\lambda,E,h}^{curr,\nu_{1/2}^{N}}(\mathcal{O}_{N}) = \frac{\mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}} \left[ \mathbf{1}_{\mathcal{O}_{N}} e^{-\Pi_{T}^{N}(h) - \Pi_{0}^{N}(h) + \int_{0}^{T} \varepsilon_{h,\lambda,E}^{N}(\eta_{t}) dt} \right]}{\mathbb{E}_{h,\lambda+E}^{\nu_{h}^{N}} \left[ e^{-\Pi_{T}^{N}(h) - \Pi_{0}^{N}(h) + \int_{0}^{T} \varepsilon_{h,\lambda,E}^{N}(\eta_{t}) dt} \right]}.$$
(4.65)

The idea is to use the (time-dependent) domination bound of Proposition 4.5 to reduce  $\mathcal{O}_N$  to a set  $\mathcal{O}_N \cap V_T^N(\delta)$  (see (4.67)) on which the exponential moment in the numerator in (4.65) cannot be too large when  $N \gg 1$ . This will allow us to use entropy estimate of Theorem 2.7 to bound these exponential moments uniformly in time and N. The arguments are similar to those used to prove Proposition 4.4.

Recall from (4.26) the definition of the sets  $S_t(M), U_T(\delta)$  for  $M, \delta > 0$  and  $t \ge 0$ :

$$S_t(M) = \left\{ |\Pi_t(h)| \le M \right\}, \qquad U_T(\delta) = \left\{ \left| \int_0^T \varepsilon_{h,\lambda,E}^N(\eta_t) \, dt \right| \le \delta \right\}.$$
(4.66)

Choosing  $M = \log \log N$ , define:

$$V_T^N(\delta) = S_T(\log \log N) \cap U_T(\delta).$$
(4.67)

Proposition 4.5 shows that  $V_T^N(\delta)^c$  is unlikely under the current biased dynamics:

$$\limsup_{N \to \infty} \mathbb{P}_{\lambda+E}^{\nu_{1/2}^N} \left( V_T^N(\delta)^c \right) = 0 \quad \Rightarrow \quad \limsup_{N \to \infty} \mathbb{P}_{\lambda,E,T}^{curr,\nu_{1/2}^N} \left( V_T^N(\delta)^c \right) = 0.$$
(4.68)

It is therefore enough to prove the domination bound for  $(\mathcal{O}_N \cap V_T^N(\delta))_N$  rather than  $(\mathcal{O}_N)_N$ .

On  $V_T^N(\delta)$ , we can estimate the exponential moments of  $\Pi_T^N(h)$ ,  $\varepsilon_{h,\lambda,E}^N$  in the numerator as we show below. Let us bound numerator and denominator in (4.65) separately. For the denominator, Jensen inequality gives the lower bound:

$$\exp \mathbb{E}_{h,\lambda+E}^{\nu_h^N} \Big[ -\Pi_T^N(h) - \Pi_0^N(h) + \int_0^T \varepsilon_{h,\lambda,E}^N(\eta_t) \, dt \Big].$$
(4.69)

The expectation of  $\Pi_0^N(h)$ ,  $\Pi_T^N(h)$  is bounded uniformly in N, T as in (4.40). Together with the estimate of  $\varepsilon_{h,\lambda,E}^N$  in Proposition 4.2, we obtain the following bound on the denominator of (4.65):

$$\inf_{T>0} \liminf_{N\to\infty} \mathbb{E}_{h,\lambda+E}^{\nu_h^N} \left[ e^{-\Pi_T^N(h) - \Pi_0^N(h) + \int_0^T \varepsilon_{h,\lambda,E}^N(\eta_t) \, dt} \right] > 0.$$
(4.70)

Consider now the numerator in (4.65) for the set  $\mathcal{O}_N \cap V_T^N(\delta)$ . On  $V_T^N(\delta)$ , the time integral of  $\varepsilon_{h,\lambda,E}^N$  is bounded by  $\delta$ , so only the exponential moments of  $-\Pi_0^N(h), -\Pi_T^N(h)$  remain. As in (4.36), one can use Hölder inequality to estimate exponential moments of  $-\Pi_0^N(h), -\Pi_T^N(h)$ separately. The key observation is that, on the set  $S_T(\log \log N)$ ,  $\exp[-\Pi_T(h)]$  is bounded by  $\log N = o(N^{1/4})$ , therefore one can still use Pinsker's inequality in Proposition 3.2 to reduce the computation of this exponential moment to an estimate under  $\nu_h^N$ . This is done exactly as in (4.36) and following equations, so we conclude the proof of Proposition 4.3 as in (4.37)-(4.38):

$$\limsup_{N \to \infty} \mathbb{P}_{\lambda, E, h}^{curr, \nu_{1/2}^{N}} (\mathcal{O}_{N}) = \limsup_{N \to \infty} \mathbb{P}_{\lambda, E, h}^{curr, \nu_{1/2}^{N}} (\mathcal{O}_{N} \cap V_{T}^{N}(\delta))$$
$$\leq C(h, \lambda, E) \limsup_{N \to \infty} \mathbb{P}_{h, \lambda + E}^{\nu_{h}^{N}} (\mathcal{O}_{N})^{\kappa}, \qquad (4.71)$$

with  $\kappa = (2r_h + 1)^{-1}$  and  $r_h$  the spectral radius of  $-\sigma h$  ( $r_h = 0$  if h is a positive kernel).  $\Box$ 

## 5 Time decorrelation and fluctuations for the currentbiased dynamics

In this section, we carry out the second part of Step 3 as outlined in Section 2.4, establishing time decorrelation estimates. This concludes the proof of the main result, Theorem 2.4.

Throughout, let  $\lambda, E$  be sub-critical in the sense of Proposition 2.1 and let  $h = h_{\lambda,E}$  be the associated solution of (2.21).

Let  $\tau \geq 0$  and let  $G : \mathcal{D}(\mathbb{R}_+, \mathcal{S}'(\mathbb{T})) \to \mathbb{R}$  be a bounded continuous function. We need to prove:

$$\lim_{k \to \infty} \lim_{T \to \infty} \lim_{N \to \infty} \int G\left( (Y_s^N)_{t \le s \le t+\tau} \right) d\mathbb{P}_{\lambda, E, T}^{curr, \nu_{1/2}^N} = \mathbf{E}_h^{\nu_h^\infty} \left[ G\left( (Y_s)_{0 \le s \le \tau} \right) \right], \tag{5.1}$$

where  $\nu_h^{\infty}$  is the centred Gaussian field on  $\mathcal{S}'(\mathbb{T})$  with covariance  $C_h = (\sigma^{-1}\mathrm{id}-h)^{-1}$  ( $\sigma := 1/4$ ) and  $\mathbf{E}_h^{\nu_h^{\infty}}$  denotes the expectation associated with the couple  $(Y, \Pi)$  of limiting fields as given by Theorem 2.5.

The proof of (5.1) is carried out in several steps. We first start from the expression of the current-biased dynamics obtained in Proposition 4.4 to take the large N limit and reduce to quantities involving only the limiting processes  $Y, \Pi$ ; this is done below. We then study the large time behaviour of the limiting fluctuation process. This is done through a classical approximation in terms of finite-dimensional diffusions following [31], in Section 5.1–5.2, together with a large time analysis of these approximations in Section 5.3.

Let M > 0. Since  $F_M$  (recall (4.24)) is continuous and bounded, Proposition 4.4 and Theorem 2.5 give:

$$\lim_{N \to \infty} \sup_{N \to \infty} \left| \mathbb{E}_{\lambda, E, T}^{curr, \nu_{1/2}^{N}} \left[ G \right] - \frac{\mathbf{E}_{h}^{\nu_{h}^{\infty}} \left[ GF_{M} \left( -\Pi_{T}(h) \right) F_{M} \left( -\Pi_{0}(h) \right) \right]}{\mathbf{E}_{h}^{\nu_{h}^{\infty}} \left[ F_{M} \left( -\Pi_{T}(h) \right) F_{M} \left( -\Pi_{0}(h) \right) \right]} \right| \leq \frac{C(h) \|G\|_{\infty}}{M^{\kappa'}}, \quad (5.2)$$

It is convenient to rewrite the ratio of expectations in (5.2) in terms of the fluctuation field only. For any  $\delta > 0$ , let  $h_{\delta}$  denote the truncation of the Fourier series of h given in Proposition 2.1 at some  $n_{\delta} \in \mathbb{N}$  satisfying:

$$\|h_{\delta} - h\|_{\infty} \le \delta, \qquad h_{\delta} := \sum_{j=0}^{n_{\delta}} \langle e_j, h \rangle e_j.$$
(5.3)

As usual, we identify  $h_{\delta}$  with the function  $(x, y) \in \mathbb{T}^2 \mapsto h_{\delta}(x - y)$ . Equation (3.29) in Proposition 3.4 quantifies the error made when replacing h by  $h_{\delta}$  in (5.2): writing  $1 = \mathbf{1}_B + \mathbf{1}_{B^c}$ with  $B^c = \{ |\Pi_0(h - h_{\delta})| \leq \delta^{1/2}, |\Pi_T(h - h_{\delta})| \leq \delta^{1/2} \}$  and using that  $F_M$  is Lipschitz,

$$\begin{aligned} \left| \mathbf{E}_{h}^{\nu_{h}^{\infty}} \left[ GF_{M} \left( -\Pi_{T}(h) \right) F_{M} \left( -\Pi_{0}(h) \right) \right] - \mathbf{E}_{h}^{\nu_{h}^{\infty}} \left[ GF_{M} \left( -\Pi_{T}(h_{\delta}) \right) F_{M} \left( -\Pi_{0}(h_{\delta}) \right) \right] \right| \\ & \leq 2 \|G\|_{\infty} \|F_{M}\|_{\infty}^{2} \mathbf{P}_{h}^{\nu_{h}^{\infty}} \left( |\Pi_{0}(h-h_{\delta})| \geq \delta^{1/2} \text{ or } |\Pi_{T}(h-h_{\delta})| \geq \delta^{1/2} \right) \\ & + 2\delta^{1/2} \|G\|_{\infty} \|F_{M}\|_{\infty} \|F'_{M}\|_{\infty} \\ & \stackrel{(3.29)}{\leq} C(\|G\|_{\infty}, M) \delta^{1/2}. \end{aligned}$$

$$\tag{5.4}$$

The function  $F_M(-\Pi_T(h_{\delta}))$  is now a function of  $Y_T$  in view of the link (2.34) between  $\Pi_T, Y_T$ . Since the right-hand side above does not depend on time, vanishes when  $\delta$  is small and  $\delta$  does not depend on M, the proof of the main result, Theorem 2.4, is reduced to the proof of the following proposition.

**Proposition 5.1.** Let  $G : \mathcal{D}(\mathbb{R}_+, \mathcal{S}'(\mathbb{T})) \to \mathbb{R}$  be a continuous and bounded function, let  $\tau \geq 0$ , and let  $H_1, H_2 : \mathcal{S}'(\mathbb{T}) \to \mathbb{R}$  be continuous and bounded. Then:

$$\lim_{t \to \infty} \lim_{T \to \infty} \frac{\mathbf{E}_{h}^{\nu_{h}^{\infty}} \left[ G\left( (Y_{s})_{t \le s \le t + \tau} \right) H_{1}(Y_{0}) H_{2}(Y_{T}) \right]}{\mathbf{E}_{h}^{\nu_{h}^{\infty}} \left[ H_{1}(Y_{0}) H_{2}(Y_{T}) \right]} = \mathbf{E}_{h}^{\nu_{h}^{\infty}} \left[ G\left( (Y_{s})_{0 \le s \le \tau} \right) \right].$$
(5.5)

The result also holds if  $t, \tau$  are diverging functions of T provided  $T - \tau$  diverges with T.

In Proposition 5.1, the microscopic model does not appear any more. All that is required is an analysis of the long time behaviour of the limiting fluctuation process  $(Y_t)_{t\geq 0}$ , carried out in the next sections via an approximation procedure.

## 5.1 The finite-dimensional approximation

In this section, we follow [31] to construct, for each  $n \in \mathbb{N}_{\geq 1}$ , a diffusion process  $Y^{(n)} \in C(\mathbb{R}_+, \mathcal{S}'(\mathbb{T}))$ . This process is finite-dimensional in a sense made clear below, relaxes to its invariant measure exponentially fast, and  $(Y^{(n)})_n$  will be shown to converge weakly to Y. in the next sections.

Recall from (3.27) the definition of the eigenvectors  $(e_{\ell})_{\ell \in \mathbb{N}}$  of the Laplacian on the torus, and notice that the fact that  $Y_t$  is a bounded linear form for each  $t \geq 0$  implies:

$$\forall t \ge 0, \forall f \in C^{\infty}(\mathbb{T}), \qquad Y_t(f) = \sum_{\ell \in \mathbb{N}} \left\langle e_\ell, f \right\rangle Y_t(e_\ell).$$
(5.6)

To construct an approximation  $Y^{(n)}_{\cdot}$   $(n \in \mathbb{N}_{\geq 1})$ , let us make some preliminary remarks and introduce notations. Recall the definition of the self-adjoint operator  $\mathcal{L}_h$  appearing in the martingale problem of Theorem 2.5 when  $\tilde{h}$  there is equal to h.

For  $n \in \mathbb{N}_{\geq 1}$ , let  $\pi_n$  be the projection operator on  $\operatorname{span}(e_1, \dots, e_n)$ . In the following, if A is a bounded linear operator on  $C^{\infty}(\mathbb{T})$ , its projection  $\pi_n A \pi_n$  is assimilated when useful with the matrix  $(\langle e_k, A e_\ell \rangle)_{1 \leq k, \ell \leq n}$ . Recall that the covariance operator  $C_h$  defining the centred Gaussian measure  $\nu_h^{\infty}$  is given by:

$$C_h = (\sigma^{-1} \mathrm{id} - h)^{-1}, \qquad \sigma = 1/4.$$
 (5.7)

For each  $n \in \mathbb{N}_{\geq 1}$ , let  $C_n := \pi_n C_h \pi_n$  and let  $\nu^{(n)}$  be the centred normal distribution on  $\mathbb{R}^n$  with covariance matrix  $C_n$ :

$$\nu^{(n)}(dx) = p^{(n)}(x) \, dx, \qquad p^{(n)}(x) = \frac{1}{(2\pi)^{n/2} \det(C_n)^{1/2}} \exp\left[-\frac{x^T C_n^{-1} x}{2}\right], \quad x \in \mathbb{R}^n.$$
(5.8)

Let us now define the approximation process  $(Y_t^{(n)})_{t\geq 0}$ . First, by the martingale characterisation of fluctuations in Theorem 2.5, observe that  $Y_t(1) = Y_0(1)$  for all  $t \geq 0$  with probability 1,

with 1 denoting here the function constant equal to 1. We therefore only need to approximate Y for functions with vanishing average.

For  $Y^{(n)}$  to be a good approximation of Y, it will have to satisfy the same martingale problem in the large n limit. We build  $Y^{(n)}$  by appropriate finite-dimensional truncations of the drift and diffusion operator of Y. It turns out to be convenient to do so by approximating the drift, on the one hand, and the covariance  $C_h$  of the invariant measure on the other hand. The diffusion coefficient is then fixed by the other two. Introduce thus the diffusion  $y^{(n)}$  on  $\mathbb{R}^n$  given by:

$$dy_t^{(n)} = B_n y_t^{(n)} dt + \sqrt{2} D_n^{1/2} dW_t^{(n)}, \qquad (5.9)$$

with  $(W_t^{(n)})_{t\geq 0}$  a standard Brownian motion on  $\mathbb{R}^n$ , and where  $B_n$ ,  $D_n$  respectively approximate the drift and diffusion part:

$$B_{n} := \pi_{n} \mathcal{L}_{h} \pi_{n} = \left( \left\langle e_{k}, \mathcal{L}_{h} e_{\ell} \right\rangle \right)_{1 \le k, \ell \le n},$$
  
$$D_{n} := -\frac{1}{2} \left[ B_{n} C_{n} + C_{n} B_{n}^{T} \right] = -\frac{1}{2} \left[ B_{n} C_{n} + C_{n} B_{n} \right].$$
(5.10)

Note that  $B_n, C_n, D_n$  all commute since h commutes with the Laplacian (h is translation invariant, recall Proposition 2.1). Note also that  $y_{\cdot}^n$  is a reversible diffusion, admitting  $\nu^{(n)}$  as its unique invariant measure as can be checked from direct computations. The process  $Y_{\cdot}^{(n)}$  ( $n \in \mathbb{N}_{\geq 1}$ ) is then defined as follows:

$$Y_t^{(n)}(f) = \langle 1, f \rangle Y_0(1) + \sum_{j=1}^n \langle e_j, f \rangle y_t^{(n)}(j), \qquad f \in C^{\infty}(\mathbb{T}), t \ge 0.$$
(5.11)

Let  $\mathbf{P}^{(n)}, \mathbf{E}^{(n)}$  denote the probability/expectation associated with  $Y^{(n)}$ . Notice that there is a continuous function  $\Phi^{(n)}$  such that  $Y^{(n)} = \Phi^{(n)}(y^{(n)})$ . Therefore, abusing notations, we will also write  $\mathbf{P}^{(n)}, \mathbf{E}^{(n)}$  for the probability and expectation associated with  $y^{(n)}$ . Unless an initial condition is specified, the processes are assumed to start from the invariant measure  $\nu^{(n)}$  of  $y^{(n)}$  (or the corresponding measure on distributions for  $Y^{(n)}$ ).

## 5.2 Convergence to Y.

Here we prove:

**Proposition 5.2.** Assume  $y_0^{(n)}$  has law  $\nu^{(n)}$  and is independent from  $Y_0(1)$ . Then  $Y_{\cdot}^{(n)}$  converges weakly to  $Y_{\cdot}$  in  $C(\mathbb{R}_+, \mathcal{S}'(\mathbb{T}))$ .

For deterministic initial conditions, the convergence in Proposition 5.2 is proven in [31] (on  $\mathbb{R}^d$ , adapting the proof to the torus  $\mathbb{T}$  is straightforward) as stated next.

**Proposition 5.3.** [31, Theorem (1.23)]. Assume  $Y_0^{(n)}$  starts at a deterministic  $X_0^{(n)} \in \mathcal{S}'(\mathbb{T})$ with  $\lim_n X_0^{(n)} = X_0 \in \mathcal{S}'(\mathbb{T})$ . Then  $Y_{\cdot}^{(n)}$  converges weakly in  $C(\mathbb{R}_+, \mathcal{S}'(\mathbb{T}))$  to the process  $Y_{\cdot}^{X_0}$ starting at  $X_0$  and satisfying the same martingale problem as  $Y_{\cdot}$ , written out in Theorem 2.5. To prove Proposition 5.2, it is therefore enough to prove that, if  $y^{(n)}$  is distributed according to  $\nu^{(n)}$  and independent from  $Y_0(1)$ , then  $Y_0^n$  converges weakly to the centred Gaussian field  $\nu_h^\infty$  with covariance  $C_h = (\sigma^{-1} \mathrm{id} - h)^{-1}$ . This is an elementary Gaussian computation, given next. Let  $n \in \mathbb{N}$ . By definition,

$$Y_0^{(n)}(f) = \langle 1, f \rangle Y_0(1) + \sum_{j=1}^n \langle e_j, f \rangle y_0^{(n)}(k), \qquad f \in C^\infty(\mathbb{T}).$$
(5.12)

with  $y_0^{(n)}$  a normal distribution with covariance  $C_n$ . As  $Y_0(1)$  and  $y_0^{(n)}$  are independent,  $Y_0^{(n)}$  is a centred Gaussian field with covariance  $\sigma \text{id} + C_n$ . To prove that  $Y_0^{(n)}$  converges to the Gaussian field  $Y_0$  with covariance  $C_h$ , it is enough to check convergence of the covariances. Take thus  $f_1, f_2 \in C^{\infty}(\mathbb{T})$ , and write, as  $y_0^{(n)}$  has vanishing average:

$$\mathbf{E}^{(n)}\Big[Y_0^{(n)}(f_1)Y_0^{(n)}(f_2)\Big] = \sigma \langle 1, f_1 \rangle \langle 1, f_2 \rangle + \sum_{j,\ell=1}^n \langle e_j, f_1 \rangle \langle e_\ell, f_2 \rangle \langle e_j, C_h e_\ell \rangle$$
$$\xrightarrow[n \to \infty]{} \sigma \langle 1, f_1 \rangle \langle 1, f_2 \rangle + \sum_{j,\ell=1}^\infty \langle e_j, f_1 \rangle \langle e_\ell, f_2 \rangle \langle e_j, C_h e_\ell \rangle.$$
(5.13)

To prove convergence of the covariances to those of  $Y_0$ , we need the last line to be equal to:

$$\sum_{j,\ell=0}^{\infty} \langle e_j, f_1 \rangle \langle e_\ell, f_2 \rangle \langle e_j, C_h e_\ell \rangle = \int_{\mathbb{T}} f_1(x) (C_h f_2)(x) \, dx.$$
(5.14)

For this to be true, it is enough to prove that  $\langle e_j, C_h 1 \rangle = \langle 1, C_h e_j \rangle = 0$  for each  $j \in \mathbb{N}_{\geq 1}$ . We show in Proposition C.1 that  $\int_{\mathbb{T}} h(x) dx = 0$ . It follows that the symmetric operator  $C_h$  admits 1 has an eigenvector, thus the orthogonal decomposition  $\mathbb{L}^2(\mathbb{T}) = 1 \oplus 1^{\perp}$  (for the usual inner product in  $\mathbb{L}^2(\mathbb{T})$ ) is stable under  $C_h$ . This concludes the proof of Proposition 5.2.

## 5.3 Long time asymptotics and conclusion of the proof

To conclude the proof of Proposition 5.1, it is enough to prove that Y decorrelates in time in the following sense.

**Proposition 5.4.** Let  $\lambda, E \in \mathbb{R}$  be such that  $\sigma h$  has leading eigenvalue strictly below 1. Let  $\tau \geq 0$  and  $G : \mathcal{D}(\mathbb{T}_+, \mathcal{S}'(\mathbb{T})) \to \mathbb{R}$ ,  $H_1, H_2 : \mathcal{S}'(\mathbb{T}) \to \mathbb{R}$  be bounded continuous functions. There is C > 0 and  $c_h > 0$  such that, for any  $t \geq 0$  with  $t + \tau < T$ ,

$$\mathbf{E}_{h,\lambda+E}^{\nu_h^{\infty}} \Big[ H_1(Y_0) H_2(Y_T) G\big((Y_s)_{t \le s \le t+\tau}\big) \Big] - \nu^{\infty} \Big[ H_1(Y) \Big] \nu^{\infty} \Big[ H_2(Y) \Big] \mathbf{E}_{h,\lambda+E}^{\nu_h^{\infty}} \Big[ G\big((Y_s)_{0 \le s \le \tau}\big) \Big] \Big| \\
\le C \exp \Big[ -c_h \min\{t, T-\tau-t\} \Big].$$
(5.15)

As  $Y_{\cdot}^{(n)}$  converges weakly to  $Y_{\cdot}$ , it is enough to prove Proposition 5.4 for  $Y_{\cdot}^{(n)}$ ,  $\nu^{(n)}$  instead of  $Y_{\cdot}$ ,  $\nu^{\infty}$  with the same constants  $C, c_h$  for each  $n \in \mathbb{N}_{\geq 1}$ . In particular, Proposition 5.4 is a direct consequence of the following lemma (recall the definition of  $\mathbf{E}^{(n)}$  from below (5.11)). **Lemma 5.5.** With the notations of Proposition 5.4 and  $c_h$  given by (5.18) below, for any  $\delta > 0$ , there are constants  $C_1, C_2$  depending only on  $H_1, H_2, h, \delta$  such that, for any  $t \ge \delta$ :

$$\sup_{n \in \mathbb{N}_{\geq 1}} \left| \mathbf{E}^{(n)} \Big[ H_1(Y_0^{(n)}) G\big( (Y^{(n)})_{s \geq t} \big) \Big] - \nu^{(n)} (H_1) \mathbf{E}^{(n)} \Big[ G\big( (Y^{(n)})_{s \geq 0} \big) \Big] \right| \le C_1 e^{-c_h t}, \qquad (5.16)$$

and for  $t + \tau < T$ :

$$\sup_{n \in \mathbb{N}_{\geq 1}} \left| \mathbf{E}^{(n)} \Big[ H_2(Y_T^{(n)}) G\big( (Y^{(n)})_{s \leq t+\tau} \big) \Big] - \nu^{(n)} (H_2) \mathbf{E}^{(n)} \Big[ G\big( (Y^{(n)})_{s \leq t+\tau} \big) \Big] \right| \leq C_2 e^{-c_h (T-t-\tau)}.$$
(5.17)

To prove Lemma 5.5, the key element is the fact that the generator  $\mathcal{L}_h$  is for sub-critical  $\lambda, E$ .

**Lemma 5.6.** Let  $\lambda, E \in \mathbb{R}$  be sub-critical in the sense of Proposition 2.1. Then the generator  $\mathcal{L}_h$  is gapped: there is  $c_h > 0$  such that, for any  $f \in C^{\infty}(\mathbb{T})$  with  $\int_{\mathbb{T}} f(x) dx = 0$ :

$$\int_{\mathbb{T}} f(x) \big( -\mathcal{L}_h f \big)(x) \, dx \ge c_h \int_{\mathbb{T}} f(x)^2 \, dx.$$
(5.18)

Let  $u \in \mathbb{R}^n$  with  $||u||_2 = 1$  and write  $f = \sum_{\ell=1}^n u_\ell e_\ell$ . Then the gap (5.18) carries over to  $B_n$ :

$$u^{T}(-B_{n})u = -\int_{\mathbb{T}} f(x) \left(\mathcal{L}_{h}f\right)(x) dx \ge c_{h}.$$
(5.19)

*Proof.* Recall that  $\mathcal{L}_h$  acts on  $f \in C^{\infty}(\mathbb{T})$  according to:

$$\mathcal{L}_h f(x) = f''(x) - \sigma \int_{\mathbb{T}} h'(x-y) f'(y) \, dy = (\mathrm{id} - \sigma h) \Delta f(x).$$
(5.20)

As h is translation invariant,  $\Delta$  and h commute and are thus diagonalisable in the real orthonormal basis  $(e_{\ell})_{\ell \in \mathbb{N}}$  of eigenvectors of the Laplacian. The assumption that  $\lambda, E$  are sub-critical implies by Proposition 2.1 that  $\sigma h$  has leading eigenvalue  $\rho_{\sigma h}$  strictly smaller than 1. As  $-\Delta$  has gap  $4\pi^2$  on  $\mathbb{T}$ , this implies:

$$-\int_{\mathbb{T}} f(x)\mathcal{L}_h f(x) \, dx \ge c_h \int_{\mathbb{T}} f(x)^2 \, dx, \quad c(h) \ge 4\pi^2 (1-\rho_{\sigma h}). \tag{5.21}$$

*Proof of Lemma 5.5.* The proof is an explicit Gaussian computation for finite dimensional Ornstein-Uhlenbeck processes.

Let us start with some notations. Let  $\nu_t^{(n)}(x, dy) = p_t^{(n)}(x, y) dy$  denote the law of  $y_t^{(n)}$ starting from x, and recall that the Gaussian measure  $\nu^{(n)}(dy) = p^{(n)}(y) dy$  is the invariant measure of the diffusion  $y_{\cdot}^{(n)}$ . For two probability measures  $\pi_1(dx) = \rho_1(x) dx$ ,  $\pi_2(dx) = \rho_2(x) dx$  on  $\mathbb{R}^n$ , let  $\|\pi_1 - \pi_2\|_{TV}$  denote their total variation distance:

$$\|\pi_1 - \pi_2\|_{TV} = \frac{1}{2} \int_{\mathbb{R}^n} |\rho_1(x) - \rho_2(x)| \, dx.$$
(5.22)

Let  $\Psi^{(n)}$  be the continuous function such that  $Y_t^{(n)} = \Psi^{(n)}(y_t^{(n)})$  for each  $t \ge 0$ . We first prove decorrelation at time 0, i.e. (5.16). By the Markov property for  $y_t^{(n)}$ ,

$$\begin{aligned} \left| \mathbf{E}^{(n)} \Big[ H_1(Y_0^{(n)}) G\big((Y^{(n)})_{s \ge t}\big) \Big] - \nu^{(n)}(H_1) \mathbf{E}^{(n)} \Big[ G\big((Y^{(n)})_{s \ge 0}\big) \Big] \right| \\ &= \left| \int \nu^{(n)}(dx) H_1(\Psi^{(n)}(x)) \Big( \mathbf{E}^{\nu_t^{(n)}(x)} \Big[ G\big((\Psi^{(n)}(y_s^{(n)})\big)_{s \ge 0}\big) \Big] - \mathbf{E}^{(n)} \Big[ G\big((\Psi^{(n)}(y_s^{(n)})\big)_{s \ge 0}\big) \Big] \Big) \right| \\ &\le 2 \|G\|_{\infty} \|H_1\|_{\infty} \int \nu^{(n)}(dx) \|\nu_t^{(n)}(x) - \nu^{(n)}\|_{TV}. \end{aligned}$$
(5.23)

Let us compute this total variation distance. Writing  $\nu_t^{(n)}(x) = p_t^{(n)}(x, y) dy$ , an elementary elementary computation gives  $p^{(n)}(y) = \lim_{t \to \infty} p_t^{(n)}(x, y)$  and:

$$p_t^n(x,y) = \frac{1}{(2\pi)^{n/2} \det(C_n(t))^{1/2}} \exp\left[-\frac{1}{2}(y - e^{tB_n}x)^T C_n(t)^{-1}(y - e^{tB_n}x)\right],$$
(5.24)

with, recalling (5.10) and the fact that  $B_n, D_n, C_n$  commute:

$$C_n(t) := \int_0^t D_n^{1/2} e^{\frac{s}{2}(B_n + B_n^T)} D_n^{1/2} \, ds = C_n - \int_t^\infty D_n^{1/2} e^{sB_n} D_n^{1/2} \, ds.$$
(5.25)

The last integral is well defined as  $B_n$  is gapped, recall (5.19). Let  $x \in \mathbb{R}^n$  and  $t \ge 1$ . Pinsker's inequality (see Proposition 3.2, the inequality is valid on general measurable spaces) bounds the total variation distance in terms of the relative entropy:

$$\|\nu_t^{(n)}(x) - \nu^{(n)}\|_{TV} \le \sqrt{2H(\nu_t^{(n)}(x)|\nu^{(n)})},\tag{5.26}$$

with:

$$H(\nu_t^{(n)}(x)|\nu^{(n)}) := \int \log\left(\frac{p_t^{(n)}(x,y)}{p^{(n)}(y)}\right) p_t^{(n)}(x,y) \, dy.$$
(5.27)

The relative entropy between Gaussian measures can be computed explicitly:

$$H(\nu_t^{(n)}(x)|\nu^{(n)}) = \frac{1}{2} \operatorname{tr}(C_n^{-1}C_n(t)) - \frac{n}{2} + \frac{1}{2}(e^{tB_n}x)^T C_n^{-1}(e^{tB_n}x) - \frac{1}{2}\log\det\left(C_n^{-1}C_n(t)\right).$$
(5.28)

Let us bound each term above. Consider first the trace. The explicit formula (5.25) for  $C_n(t), C_n$  gives, as  $B_n$  is invertible:

$$\frac{1}{2} \operatorname{tr} \left( C_n^{-1} C_n(t) \right) - \frac{n}{2} = -\frac{1}{2} \operatorname{tr} \left( -2B_n \int_t^\infty e^{2sB_n} \, ds \right) = -\frac{1}{2} \operatorname{tr} \left( e^{2tB_n} \right) \\ = -\frac{1}{2} e^{-2c_h t} \operatorname{tr} \left( e^{2t(B_n + c_h \operatorname{id})} \right).$$
(5.29)

Since  $\Delta$  and h commute by translation invariance of h, the spectrum of  $B_n$  is known explicitly. In particular, if  $0 \leq \lambda_1^n \leq \ldots \leq \lambda_n^n(B_n)$  denote the eigenvalues of  $-B_n$  then:

$$\lambda_{2\ell}^n, \lambda_{2\ell+1}^n \ge \ell^2 c_h, \qquad 0 \le \ell \le \lfloor n/2 \rfloor.$$
(5.30)

As a result,

$$\operatorname{tr}\left(e^{2t(B_n+c_{\ell}\mathrm{id})}\right) \leq \sum_{\ell=1}^{\lfloor n/2 \rfloor} 2e^{-(\ell^2-1)c_h t},\tag{5.31}$$

which is bounded uniformly in n and  $t \ge 1$ . Recalling (5.29), we have thus proven:

$$\forall t \ge 1, \qquad \sup_{n} \left| \frac{1}{2} \operatorname{tr} \left( C_n^{-1} C_n(t) \right) - \frac{n}{2} \right| \le C e^{-2c_h t}, \qquad C = C(h) > 0.$$
 (5.32)

The determinant in (5.28) is treated similarly:

$$\frac{1}{2}\log\det\left(C_n^{-1}C_n(t)\right) = \frac{1}{2}\log\det\left(1 - e^{2tB_n}\right) \le Ce^{-2c_h t}, \qquad C = C(h) > 0. \tag{5.33}$$

All in all, recalling (5.28), we find that (5.23) is bounded by:

$$C\|G\|_{\infty}\|H_1\|_{\infty}\int\nu^{(n)}(dx)\left[e^{-2c_ht}+(e^{tB_n}x)^TC_n^{-1}(e^{tB_n}x)\right]^{1/2}.$$
(5.34)

Equation (5.34) can be bounded using Cauchy-Schwarz inequality and the following inequality, valid for any matrices A, B:

$$\operatorname{tr}(A^T B) \le \operatorname{tr}(A^T A)^{1/2} \operatorname{tr}(B^T B)^{1/2}.$$
 (5.35)

The resulting bound on (5.34) reads:

$$C\|G\|_{\infty}\|H_{1}\|_{\infty}\left[e^{-2c_{h}t} + \int \nu^{(n)}(dx)(e^{tB_{n}}x)^{T}C_{n}^{-1}e^{tB_{n}}x\right]^{1/2}$$
  
$$= C\|G\|_{\infty}\|H_{1}\|_{\infty}\left[e^{-2c_{h}t} + \operatorname{tr}\left(C_{n}e^{tB_{n}}C_{n}^{-1}e^{tB_{n}}\right)\right]^{1/2}$$
  
$$\leq C\|G\|_{\infty}\|H_{1}\|_{\infty}\left[e^{-2c_{h}t} + \operatorname{tr}\left(e^{2tB_{n}}\right)\right]^{1/2}$$
  
$$\leq C'\|G\|_{\infty}\|H_{1}\|_{\infty}e^{-c_{h}t}.$$
 (5.36)

This proves the first inequality (5.16) in the lemma. This and the reversibility of  $y^{(n)}$  immediately give the second inequality (5.17).

## 6 Correlations under the current-biased dynamics

In this section, we prove Proposition 2.3 on the correlation structure of the current-biased dynamics at times  $1 \ll t \ll T$ . We then show heuristically that the result agrees with the prediction of [15], where correlations were computed at fixed density of particles.

**Proposition 6.1.** Let  $\lambda, E \in \mathbb{R}$  be sub-critical as in Proposition 2.1. Let  $k_{curr}$  denote the kernel encoding long-range correlations under the current-biased dynamics at times far away from 0 and T, in the sense that:

$$\lim_{T \to \infty} \lim_{N \to \infty} \mathbb{E}_{\lambda, E, T}^{curr, \nu_{1/2}^{N}} \left[ Y_{T/2}^{N}(f_1) Y_{T/2}^{N}(f_2) \right] = \int f_1(x) (\sigma i d + k_{curr}) f_2(x) \, dx.$$
(6.1)

Then  $k_{curr}$  satisfies  $(\sigma^{-1}id - h)^{-1} = \sigma id + k_{curr}$  and is explicitly given by:

$$k_{curr}(x) = \sqrt{2\sigma} \sum_{\ell \ge 1} \left[ \frac{1}{\sqrt{1 + \frac{\lambda(\lambda + 2E)\sigma}{\pi^2 \ell^2}}} - 1 \right] \sqrt{2} \cos(2\pi \ell x), \qquad x \in \mathbb{T}.$$
 (6.2)

Proof. Theorem 2.4 shows that the current-biased dynamics has the same fluctuations as the dynamics  $\mathbb{P}_{h,\lambda+E}^{\nu_h^N}$  in the sense of convergence of bounded observables. To determine the correlation structure of the current-biased dynamics (and in particular prove that the limits in (6.1) make sense), we check that Theorem 2.4 also holds for observables of the form  $Y_t^N(f_1)Y_t^N(f_2)$  ( $t \ge 0, f_1, f_2 \in C^{\infty}(\mathbb{T})$ ), then compute this covariance using the characterisation fo limiting fluctuations in Theorem 2.5.

Let  $f_1, f_2 \in C^{\infty}(\mathbb{T})$ . The moment bound (3.3) on  $\Pi_t^N$  at each time t and the relationship (2.12) between  $\Pi_t^N$  and  $Y_t^N$  give uniform integrability of  $(Y_t^N(f_1)Y_t^N(f_2))_N$  under  $\mathbb{P}_{h,\lambda+E}^{\nu_h^N}$ . Combining with the time decorrelation of Proposition 5.4, valid with the same proof for unbounded test functions  $G(Y_{T/2}) = Y_{T/2}(f_1)Y_{T/2}(f_2)$  that have bounded moments, we get:

$$\lim_{T \to \infty} \lim_{N \to \infty} \mathbb{E}_{\lambda, E, T}^{curr, \nu_{1/2}^{N}} \left[ Y_{\frac{T}{2}}^{N}(f_{1}) Y_{\frac{T}{2}}^{N}(f_{2}) \right] = \mathbf{E}_{h, \lambda + E}^{\nu^{\infty}} \left[ Y_{0}(f_{1}) Y_{0}(f_{2}) \right], \tag{6.3}$$

where we used the stationarity of Y. under  $\mathbf{E}_{h,\lambda+E}^{\nu^{\infty}}$ . Since  $Y_0$  is a Gaussian field with covariance  $C_h = (\sigma^{-1} \mathrm{id} - h)^{-1}$ , there is  $k_{curr} \in C^{\infty}([0,1]) \cap C^0(\mathbb{T})$  such that  $C_h = \sigma \mathrm{id} + k_{curr}$  (because  $C_h - \sigma \mathrm{id} = \sigma h C_h$  can be checked to be a kernel operator on  $\mathbb{L}^2(\mathbb{T})$ ; the regularity of the associated kernel is inherited from h).

It remains to find an expression for this kernel. We do so using Ito formula on  $Y_t(f_1)Y_t(f_2)$  $(t \ge 0)$  and the stationarity of  $(Y_s)_{s\ge 0}$ . Recall the martingale problem characterising  $(Y_s)_{s\ge 0}$ in Theorem 2.5. Taking expectations, we find:

$$\int_{\mathbb{T}} f_1(x)(\sigma \mathrm{id} + k_{curr}) \mathcal{L}_h f_2(x) \, dx + \sigma \int_{\mathbb{T}} f_1'(x) f_2'(x) \, dx = 0, \tag{6.4}$$

where  $\mathcal{L}_h f_2 = f_2'' - \sigma \int h'(\cdot - y) f_2'(y) dy$ . Taking  $f_1 = f_2 = f$  with  $f'(x) = e^{-2i\pi\ell x}$   $(x \in \mathbb{T})$ for  $\ell \in \mathbb{Z} \setminus \{0\}$  and evaluating at x = 0 yields, with  $c_\ell(u) = \int_{\mathbb{T}} u(x) e^{-2i\pi\ell x} dx$  the  $\ell^{\text{th}}$  Fourier coefficient of a function  $u \in \mathbb{L}^2(\mathbb{T})$ :

$$-2\iota \pi \ell c_{\ell}(k_{curr}) + \sigma c_{\ell}(k_{curr})c_{\ell}(h') + \sigma^2 c_{\ell}(h') = 0.$$
(6.5)

Since  $c_{\ell}(h') = 2\iota \pi \ell c_{\ell}(h)$  is known by Proposition 2.1, we obtain:

$$c_{\ell}(k_{curr}) = \frac{\sigma^2 c_{\ell}(h')}{2\iota \pi \ell - \sigma c_{\ell}(h')} = \sigma \left[\frac{1}{\sqrt{1 + \frac{\lambda(\lambda + 2E)\sigma}{\pi^2 \ell^2}}} - 1\right].$$
(6.6)

**Comparison with [15]**. The conjecture in [15] (see Equation 61 there with the values D = 1,  $\sigma(\rho) = 2\rho(1-\rho)$ ,  $j_0 = \sigma(\rho)(\lambda + E)$ ,  $\bar{j}_0 = \sigma(\rho)E$ ) is the following. Let  $\rho \in [0, 1]$ , and consider the current-biased dynamics (1.6) starting from the uniform measure with  $p = \lfloor N\rho \rfloor$ .

Then, in the large N limit, the two-point correlation kernel  $k_{curr}^{\rho}$  satisfies, for each  $f \in C^{\infty}(\mathbb{T})$ and  $x \in \mathbb{T}$ :

$$k_{curr}^{\rho}f(x) = \rho(1-\rho) \int_{\mathbb{T}} \sum_{\ell \ge 1} 2\cos\left(2\pi\ell(x-y)\right) \left(1 + \left[\frac{1}{\sqrt{1 + \frac{\rho(1-\rho)\lambda(\lambda+2E)}{\pi^{2}\ell^{2}}}} - 1\right]\right) f(y) \, dy.$$
(6.7)

To be precise, the conjecture in [15] is formulated for the dynamics conditioned to having current  $j_0$ , rather than for our current-biased dynamics tilted by  $\lambda$  times the current. In the large N, large T limit, the two should be equivalent with  $j_0 = \rho(1-\rho)(\lambda+E)$  at least in the absence of dynamical phase transition [46].

Using the identity  $\delta(\cdot) - 1 = \sum_{\ell \ge 1} 2\cos(2\pi\ell \cdot)$ , with  $\delta(\cdot)$  the Dirac distribution, (6.7) becomes:

$$C_{curr}^{\rho}f(x) = \rho(1-\rho)f(x) + k_{curr}^{\rho}f(x),$$
(6.8)

with the translation-invariant kernel  $k_{curr}^{\rho}$  given for  $x \in \mathbb{T}$  by:

$$k_{curr}^{\rho}(x) = -\rho(1-\rho) + \rho(1-\rho) \sum_{\ell \ge 1} 2\cos(2\pi\ell x) \left[ \frac{1}{\sqrt{1 + \frac{\rho(1-\rho)\lambda(\lambda+2E)}{\pi^2\ell^2}}} - 1 \right].$$
 (6.9)

The next proposition states that the expression (6.2) of  $k_{curr}$  and the  $k_{curr}^{\rho}$  of [15] are compatible, in the sense that one recovers  $k_{curr}$  upon averaging  $k_{curr}^{\rho}$  on all densities.

**Proposition 6.2.** Let  $\mathcal{B}_N$  denote the Binomial law with parameters N, 1/2. Then:

$$k_{curr}(x) = \lim_{N \to \infty} \left[ \mathbb{E}_{\mathcal{B}_N} \left[ k_{curr}^{p/N}(x) \right] + N \operatorname{Var}_{\mathcal{B}_N} \left[ p/N \right] \right] = \lim_{N \to \infty} \mathbb{E}_{\mathcal{B}_N} \left[ k_{curr}^{p/N}(x) \right] + \frac{1}{4}, \qquad (6.10)$$

**Remark 6.3.** The decomposition of  $k_{curr}$  as the sum of two terms is not artificial. Indeed, on the one hand  $k_{curr}$  is the limit of N times a certain covariance, and the right-hand side of (6.10) similarly involves two rescaled covariances. On the other hand  $k_{curr}$  can be seen as the limiting covariance under the convolution of a binomial measure determining the density of particles, and a measure associated with the steady state of the current biased simple exclusion at fixed density. The decomposition of  $k_{curr}$  then follows from the general fact that for any three probability measures  $\pi, \mu, \nu$  such that  $\pi = \mu \star \nu$  and any test functions F, G for which the following quantities make sense:

$$Cov_{\pi}(F,G) = Cov_{\mu}(\mathbb{E}_{\nu}[F],\mathbb{E}_{\nu}[G]) + \mathbb{E}_{\mu}[Cov_{\nu}(F,G)].$$
(6.11)

Proof. Note that Fourier coefficients of  $k_{curr}^{\rho}$  in (6.9) scale like  $o(\ell^{-2})$  uniformly in  $\rho$ , thus  $(\rho, x) \mapsto k_{curr}^{\rho}(x)$  is continuous on  $[0, 1] \times \mathbb{T}$  by Dirichlet's normal convergence theorem. Let  $x \in \mathbb{T}$ . As  $\rho \in [0, 1] \mapsto k_{curr}^{\rho}(x)$  is continuous and bounded, the law of large numbers gives, recalling the expression (6.2) of  $k_{curr}$ :

$$\lim_{N \to \infty} \mathbb{E}_{\mathcal{B}_N} \left[ k_{curr}^{p/N}(x) \right] = -\frac{1}{4} + k_{curr}(x).$$
(6.12)

On the other hand, elementary computations give  $N \operatorname{Var}_{\mathcal{B}_N}[p/N] = N/4$ .

## Acknowledgements

The author thanks Thierry Bodineau for many discussions and comments at all stages of this work. This work was partially carried out at the University of Cambridge with support from the European Research Council under the European Union's Horizon 2020 research and innovation programme (grant agreement No. 851682 SPINRG).

## A Concentration of measure

#### A.1 Concentration for discrete Gaussian measures

Let  $g: \mathbb{T}^2 \to \mathbb{R}$  be a bounded function. Recall that  $\nu_{1/2}^N$  is the product Bernoulli measure with density 1/2 and that  $\nu_g^N$  is the following discrete measure:

$$\nu_g^N := (\mathcal{Z}_g^N)^{-1} e^{2\Pi^N(g)} \nu_{1/2}^N, \quad \mathcal{Z}_g^N \text{ a normalisation factor.}$$
(A.1)

We often decompose g as  $g = g_- + g_+$  with  $g_-(g_+)$  a negative (positive) kernel on  $\mathbb{L}^2(\mathbb{T})$ :

$$\forall f \in \mathbb{L}^2(\mathbb{T}), \qquad \pm \int_{\mathbb{T}^2} f(x) g_{\pm}(x, y) f(y) \, dx \, dy \ge 0. \tag{A.2}$$

When  $g_{\pm} \in C^0(\mathbb{T}^2)$ , it is known that the definition (A.2) is equivalent to asking for the matrix  $\pm (g_{\pm}(x_i, x_j))_{1 \leq i,j \leq n}$  to be positive for any  $n \in \mathbb{N}_{\geq 1}$  and any  $x_1, \ldots, x_n \in \mathbb{T}$ , see e.g. [29]. In the following we will often assume  $g_{\pm} \in C_D^1(\mathbb{T}^2) \subset C^0(\mathbb{T}^2)$ , the set of continuous functions with restrictions to  $\{x \leq y\}, \{x \geq y\}$  that are  $C^1$  (the derivatives of the restrictions on the diagonal need not coincide). Recall also the notations:

$$\sigma := 1/4, \qquad \bar{\eta}_i := \eta_i - \frac{1}{2}, \quad i \in \mathbb{T}_N.$$
(A.3)

**Lemma A.1.** Let  $n \in \mathbb{N}$  and let  $I \subset \mathbb{T}_N$  with |I| = 2n + 1. Then:

$$\nu_g^N \left(\prod_{i \in I} \bar{\eta}_i\right) = 0. \tag{A.4}$$

Moreover, assume that the g of (A.1) satisfies  $g_{\pm} \in C_D^1(\mathbb{T}^2)$  and that, for some  $\delta > 0$ :

$$\forall f \in \mathbb{L}^2(\mathbb{T}), \qquad \sigma \int_{\mathbb{T}^2} f(x)g_+(x,y)f(y)\,dx\,dy \le (1-\delta)\int_{\mathbb{T}} f(x)^2\,dx. \tag{A.5}$$

Then:

$$1 \le \sup_{N \in \mathbb{N}_{\ge 1}} \mathcal{Z}_g^N < \infty \quad and \quad \sup_{J \subset \mathbb{T}_N : |J| = 2n} \nu_g^N \Big(\prod_{j \in J} \bar{\eta}_j\Big) = O(N^{-n}), \qquad n \in \mathbb{N}_{\ge 1}.$$
(A.6)

A similar result was obtained in [13, Lemma A.1] under the stronger assumption that  $\|\sigma g_+\|_2 < 1$ . Here the assumption (A.5) is optimal.

*Proof.* The mapping  $\eta \in \Omega_N \mapsto (1 - \eta_i)_{i \in \mathbb{T}_N} \in \Omega_N$  leaves  $\nu_g^N$  invariant due to the 1/2 density, but it changes the sign of  $\prod_{i \in I} \overline{\eta_i}$  (recall  $\overline{\eta_i} := \eta_i - 1/2$ ). This proves (A.4). Moreover, Jensen inequality implies:

$$\mathcal{Z}_g^N \ge \exp\left[\nu_{1/2}^N [2\Pi^N(g)]\right] = 1. \tag{A.7}$$

The estimate on even correlations is proven in Lemma A.2 in [13] provided  $\sup_N \mathcal{Z}_g^N < \infty$ . We thus only prove this bound.

Recall that  $g_{-} \subset C^{0}(\mathbb{T}^{2})$  being a negative kernel implies:

$$2\Pi^{N}(g_{-}) := \frac{1}{2N} \sum_{i \neq j} (g_{-})_{i,j} \bar{\eta}_{i} \bar{\eta}_{j} = \frac{1}{2N} \sum_{i,j} (g_{-})_{i,j} \bar{\eta}_{i} \bar{\eta}_{j} - \frac{\sigma}{2N} \sum_{i} (g_{-})_{i,i} \le \frac{\sigma}{2} \|g_{-}\|_{\infty}.$$
(A.8)

Thus:

$$\begin{aligned} \mathcal{Z}_{g}^{N} &\leq \nu_{1/2}^{N} \Big( \exp\left[\frac{1}{2N} \sum_{i \neq j \in \mathbb{T}_{N}} (g_{+})_{i,j} \bar{\eta}_{i} \bar{\eta}_{j} + \frac{1}{2N} \sum_{i,j \in \mathbb{T}_{N}} (g_{-})_{i,j} \bar{\eta}_{i} \bar{\eta}_{j} \right] \Big) e^{\sigma \|g_{-}\|_{\infty}/2} \\ &\leq \nu_{1/2}^{N} \Big( \exp\left[\frac{1}{2N} \sum_{i \neq j \in \mathbb{T}_{N}} (g_{+})_{i,j} \bar{\eta}_{i} \bar{\eta}_{j} \right] \Big) e^{\sigma \|g_{-}\|_{\infty}/2}. \end{aligned}$$
(A.9)

It is thus enough to bound  $\mathcal{Z}_g^N$  when  $g = g_+$ , i.e.  $g_- = 0$ , which we now assume. To make use of Assumption (A.5) on g, it is convenient to replace  $(g_{i,j})_{(i,j)\in\mathbb{T}_N^2}$  by the matrix  $\hat{g}^N$ :

$$\hat{g}^{N}(i,j) = N^{2} \int_{(\frac{i}{N},\frac{j}{N}) + \frac{1}{N}[0,1)^{2}} g(x,y) \, dx \, dy, \qquad (i,j) \in \mathbb{T}_{N}^{2}. \tag{A.10}$$

The advantage of  $\hat{g}^N$  over  $(g_{i,j})_{i,j}$  is that the spectrum of  $\sigma \hat{g}^N/N$  is included in the spectrum of  $\sigma g$ . Indeed, if  $u \in \mathbb{R}^N$  is an eigenvector of  $\sigma \hat{g}^N$ , one can check that the following  $f \in \mathbb{L}^2(\mathbb{T})$  is an eigenvector for  $\sigma g$  associated with the same eigenvalue:

$$f(x) = \sqrt{N} \mathbf{1}_{\frac{1}{N}[i,i+1)}(x) u_i, \qquad x \in \mathbb{T}.$$
(A.11)

As  $g = g_+ \in C_D^1(\mathbb{T}^2) \subset C^0(\mathbb{T}^2)$ , replacing g by  $\hat{g}^N$  is not restrictive as far as bounding  $\mathcal{Z}_g^N$  is concerned:

$$\mathcal{Z}_g^N \le C(g) \,\tilde{\mathcal{Z}}_g^N, \qquad \tilde{\mathcal{Z}}_g^N := \nu_{1/2}^N \Big( \exp\left[\frac{1}{2N} \sum_{i,j \in \mathbb{T}_N} \bar{\eta}_i \bar{\eta}_j \hat{g}^N(i,j)\right] \Big). \tag{A.12}$$

Let us estimate  $\tilde{\mathcal{Z}_g^N}$ . Expanding the exponential in (A.12) yields:

$$\nu_{1/2}^{N} \left( \exp\left[\frac{1}{2N} \sum_{i,j \in \mathbb{T}_{N}} \hat{g}^{N}(i,j)\bar{\eta}_{i}\bar{\eta}_{j}\right] \right) \\ = \sum_{n=0}^{\infty} \frac{1}{2^{n}n!N^{n}} \sum_{i_{1},\dots,i_{2n}} \nu_{1/2}^{N} \left( \hat{g}^{N}(i_{1},i_{2})\dots\hat{g}^{N}(i_{2n-1},i_{2n}) \prod_{j=1}^{2n} \bar{\eta}_{i_{j}} \right).$$
(A.13)

Let  $P_{2n}$  denote the set of pairings of  $\{1, ..., 2n\}$  elements. As  $\nu_{1/2}^N$  is a product measure, the average in (A.13) vanishes whenever an index  $i_1, ..., i_{2n}$  appears only once, i.e. vanishes unless



Figure 2: Pairing of eight points (black dots) and associated bijection on four points (red dots) obtained by fusing points 2i - 1, 2i into a single point i ( $i \in \{1, 2, 3, 4\}$ ). Black lines link paired points. Lines are conserved when the points are fused (in red) and determine the cycles of the resulting bijection (here (1)(234)).

there is a pairing  $p \in P_{2n}$  such that  $i_j = i_{p(j)}$  for  $j \in \{1, ..., 2n\}$ . Thus, using  $\bar{\eta}_i^2 = \sigma$  for each  $i \in \mathbb{T}_N$ :

$$\frac{1}{N^{n}} \sum_{i_{1},...,i_{2n}} \nu_{1/2}^{N} \Big[ \hat{g}^{N}(i_{1},i_{2})...\hat{g}^{N}(i_{2n-1},i_{2n}) \prod_{j=1}^{2n} \bar{\eta}_{i_{j}} \Big] \\
= \frac{\sigma^{n}}{N^{n}} \sum_{p \in P_{2n}} \sum_{i_{1},...,i_{2n}} \hat{g}^{N}(i_{1},i_{2})...\hat{g}^{N}(i_{2n-1},i_{2n}) \mathbf{1}_{(i_{j})_{j \leq 2n} \text{ paired by } p}$$
(A.14)

To use the assumption on the spectrum of  $\sigma g$ , we now express the right-hand side in terms of the trace of  $\hat{g}^N$ . To do so, we need some notations. For  $p \in P_{2n}$ , let b(p) be the bijection on  $\{1, ..., n\}$  obtained from p by identifying integers 2i - 1, 2i  $(1 \le i \le n)$ , so that b(p)(i) = jif either 2i - 1 or 2i is paired to either 2j - 1 or 2j  $(1 \le i, j \le n)$ , see Figure 2. This is not a bijective mapping as discussed further along the proof. Let also  $c_{\ell}$   $(1 \le \ell \le n)$  denote the number of cycles with  $\ell$  points in b(p), and note that  $\sum_{\ell} \ell c_{\ell} = n$  by definition. For each  $p \in P_{2n}$ , one has:

$$\sum_{i_1,\dots,i_{2n}} \prod_{j=1}^n \hat{g}^N(i_{2j-1},i_{2j}) \mathbf{1}_{(i_j)_{j\leq 2n} \text{ paired by } p} = \prod_{\ell=1}^n \left[ \sum_{i\in\mathbb{T}_N} \left( \hat{g}^N \right)^\ell(i,i) \right]^{c_\ell}.$$
 (A.15)

As a result,

$$\frac{1}{N^{n}} \sum_{i_{1},...,i_{2n}} \nu_{1/2}^{N} \Big[ \hat{g}^{N}(i_{1},i_{2})...\hat{g}^{N}(i_{2n-1},i_{2n}) \prod_{j=1}^{2n} \bar{\eta}_{i_{j}} \Big] \\
= \sum_{p \in P_{n}} \prod_{\ell=1}^{n} \Big[ \frac{\sigma^{\ell}}{N^{\ell}} \sum_{i \in \mathbb{T}_{N}} \left( \hat{g}^{N} \right)^{\ell}(i,i) \Big]^{c_{\ell}} = \sum_{p \in P_{n}} \prod_{\ell=1}^{n} \Big[ \operatorname{tr} \left( \left( \frac{\sigma}{N} \hat{g}^{N} \right)^{\ell} \right) \Big]^{c_{\ell}}. \quad (A.16)$$

Letting  $\mathbb{E}_{P_{2n}}$  denote the average with respect to the uniform measure on pairings, we obtain:

$$\mathcal{Z}_{g}^{N} \leq \sum_{n=0}^{\infty} \frac{|P_{2n}|}{2^{n} n!} \mathbb{E}_{P_{2n}} \left[ \prod_{\ell=1}^{n} \left[ \operatorname{tr} \left( \left( \frac{\sigma}{N} \hat{g}^{N} \right)^{\ell} \right) \right]^{c_{\ell}} \right].$$
(A.17)

The fact that  $|P_{2n}| = 2^{-n}(2n)!/n!$  and the Stirling equivalent give:

$$\frac{(2n)!}{2^{2n}(n!)^2} = \frac{(1+o_n(1))}{\sqrt{\pi n}} \le \frac{C}{\sqrt{\pi(n+1)}}, \qquad n \ge 0$$
$$\Rightarrow \quad \tilde{\mathcal{Z}}_g^N \le \frac{C}{\sqrt{\pi}} \sum_{n=0}^\infty \mathbb{E}_{P_{2n}} \left[ \prod_{\ell=1}^n \left[ \operatorname{tr} \left( \left( \frac{\sigma}{N} \hat{g}^N \right)^\ell \right) \right]^{c_\ell} \right]. \tag{A.18}$$

Let us estimate this expectation. For small values of  $\ell$ , the boundedness of g implies:

$$\operatorname{tr}\left[\frac{\sigma}{N}\hat{g}^{N}\right] \leq \|\sigma g\|_{\infty}.\tag{A.19}$$

Since all eigenvalues of  $\sigma \hat{g}^N/N$  are in  $(0, 1 - \varepsilon)$  by (A.5) and the discussion below (A.10), the trace will be dominated by the leading eigenvalue for large values of  $\ell$  and we prove next the existence of  $\ell_0 \in \mathbb{N}_{\geq 1}$ , independent of N, such that:

$$\forall \ell \ge \ell_0, \qquad \operatorname{tr}\left[\left(\frac{\sigma}{N}\hat{g}^N\right)^\ell\right] \le (1-\varepsilon)^{\ell/2}.$$
 (A.20)

Write first, for any  $\ell \in \mathbb{N}_{\geq 1}$ :

$$\operatorname{tr}\left[\left(\frac{\sigma}{N}\hat{g}^{N}\right)^{\ell}\right] \leq (1-\varepsilon)^{\ell-1}\operatorname{tr}\left[\frac{\sigma}{N}\hat{g}^{N}\right] \leq (1-\varepsilon)^{\ell-1}\|\sigma g\|_{\infty}.$$
 (A.21)

Choose  $\ell_1 = \ell_1(g) \in \mathbb{N}_{\geq 1}$  such that the above is smaller than 1 for  $\ell \geq \ell_1$ . The claim (A.20) follows for  $\ell \geq \ell_0 = 2\ell_1$ , as:

$$\operatorname{tr}\left[\left(\frac{\sigma}{N}\hat{g}^{N}\right)^{\ell}\right] \leq (1-\varepsilon)^{\ell-\ell_{1}}\operatorname{tr}\left[\left(\frac{\sigma}{N}\hat{g}^{N}\right)^{\ell_{1}}\right] \leq (1-\varepsilon)^{\ell-\ell_{1}} \leq (1-\varepsilon)^{\ell/2}.$$
 (A.22)

It follows that the sum in (A.18) satisfies, recalling  $\sum_{\ell} \ell c_{\ell} = n$ :

$$\sum_{n=0}^{\infty} \mathbb{E}_{P_{2n}} \left[ \prod_{\ell=1}^{n} \operatorname{tr} \left[ \left( \frac{\sigma}{N} \hat{g}^{N} \right)^{\ell} \right]^{c_{\ell}} \right] \\ \leq C(\ell_{0}, \|g\|_{\infty}) + \sum_{n \geq \ell_{0}} \mathbb{E}_{P_{2n}} \left[ \exp \left[ \sum_{\ell=1}^{\ell_{0}} \ell c_{\ell} \log(\|\sigma g\|_{\infty}) + \frac{1}{2} \log(1-\varepsilon) \left( n - \sum_{\ell=1}^{\ell_{0}} \ell c_{\ell} \right) \right] \right] \\ \leq C(\ell_{0}, \|g\|_{\infty}) + \sum_{n \geq \ell_{0}} (1-\varepsilon)^{n/2} \prod_{\ell=1}^{\ell_{0}} \mathbb{E}_{P_{2n}} \left[ \exp \left[ \ell_{0}^{2} c_{\ell} \left[ \sigma \|g\|_{\infty} - \frac{1}{2} \log(1-\varepsilon) \right] \right] \right], \quad (A.23)$$

where we used Hölder inequality in the last line on the sum  $\sum_{\ell=1}^{\ell_0}$ . It remains to estimate this exponential moment, which now only involves small cycles.

This is done by rewriting it as an exponential moments of cycles of general permutations on n elements, which is a well known quantity. To do so, notice that each bijection on nelements is associated with  $2^{\sum_{\ell} (\ell-1)c_{\ell}} = 2^{n-\sum_{\ell} c_{\ell}}$  different pairings in  $P_{2n}$ , corresponding to the  $2^{(\ell-1)c_{\ell}}$  ways of joining edges in each cycle. Thus, if  $\mathbb{E}_{S_n}$  denote the uniform measure on permutations with n elements, writing thanks to Jensen inequality:

$$\mathbb{E}_{P_{2n}}[\cdot] = \mathbb{E}_{S_n} \left[ 2^{n - \sum_{\ell} c_{\ell}} \right]^{-1} \mathbb{E}_{S_n} \left[ \cdot 2^{n - \sum_{\ell} c_{\ell}} \right] \le 2^{\sum_{\ell} \mathbb{E}_{S_n} \left[ c_{\ell} \right]} \mathbb{E}_{S_n} \left[ \cdot \right], \tag{A.24}$$

it is enough to prove that the following quantity has slower-than-geometric growth in n:

$$2^{\sum_{\ell} \mathbb{E}_{S_n}[c_{\ell}]} \prod_{\ell=1}^{\ell_0} \mathbb{E}_{S_n} \bigg[ \exp\bigg[\ell_0^2 c_{\ell} \big[\sigma \|g\|_{\infty} - \frac{1}{2}\log(1-\varepsilon)\big] \bigg] \bigg].$$
(A.25)

The statistics of the number  $c_{\ell}$  of cycles of length  $\ell$  under  $\mathbb{E}_{S_n}$  is well known [42, Theorem 2]. In particular, the moment of order  $k \in \mathbb{N}$  is smaller than the same moment for a Poisson distribution of parameter  $1/\ell$ , which has moment generating function defined on the full real line. The above exponential moment is thus bounded for each  $n > \ell_0$  in terms of  $\ell_0, g$  only. Also,  $\mathbb{E}_{S_n}[c_{\ell}] = 1/\ell$  for each  $n > \ell_0$ . This concludes the proof:

$$2^{\sum_{\ell} \mathbb{E}_{S_n}[c_{\ell}]} \prod_{\ell=1}^{\ell_0} \mathbb{E}_{S_n} \left[ \exp\left[\ell_0^2 c_{\ell} \left[\sigma \|g\|_{\infty} - \frac{1}{2}\log(1-\varepsilon)\right] \right] \right] \le C(\ell_0, g) 2^{\log n}, \quad n \ge \ell_0.$$
(A.26)

In the following, let  $I_N$  denote the identity matrix on  $\mathbb{R}^N$  and define  $g^N$  as the matrix:

$$g^{N}(i,j) = g_{i,j}, \qquad g^{N}(i,i) = 0, \qquad i \neq j \in \mathbb{T}_{N}.$$
 (A.27)

Write also  $g_{\pm}^{N}$  for the matrices similarly defined in terms of  $g_{\pm}$ .

**Proposition A.2** (Approximate Wick theorem for  $\nu_g^N$ ). Assume that  $g = g_+ + g_-$  satisfies the hypotheses of Lemma A.1. For  $n \in \mathbb{N}_{\geq 1}$ , let  $P_{2n}$  be the set of pairings of 2n elements. For each  $I \subset \mathbb{T}_N$  with |I| = 2n, there is then an error term  $e_N^I$  such that:

$$\nu_g^N \Big(\prod_{i \in I} \bar{\eta}_i\Big) = \sum_{p \in P_{2n}} \prod_{e=(i,p(i)) \text{ pair in } p} \left(\sigma^{-1} I_N - N^{-1} g^N\right)^{-1} (i,p(i)) + e_N^I, \tag{A.28}$$

where pairs are counted only once in the product, and with:

$$\sup_{|J|=2n} |e_N^J| = O(N^{-n-1}).$$
(A.29)

Proof. We claim that the hypothesis on  $g_+$  ensures that  $\sigma^{-1}I_N - N^{-1}g^N$  is invertible for all large enough  $N \in \mathbb{N}_{\geq 1}$ . Indeed, the continuity of  $g_-$  implies that  $g_-^N$  has eigenvalues in  $\mathbb{R}_-$ , see e.g. [29, Theorem 2.3]. On the other hand, recall the notation  $\hat{g}_+^N$  from (A.10). Since  $g_+ \in C_D^1(\mathbb{T}^2)$ , one has:

$$\frac{\sigma}{N} \max_{i \neq j} \left| [\hat{g}_{+}^{N} - g_{+}^{N}](i,j) \right| = O(N^{-2}), \qquad \frac{\sigma}{N} \max_{i} \left| [\hat{g}_{+}^{N} - g_{+}^{N}](i,i) \right| = O(N^{-1}).$$
(A.30)

It follows that the matrix  $\frac{\sigma}{N}(\hat{g}_+^N - g_+^N)$  has eigenvalues bounded by C/N for some C > 0. Therefore, for any vector  $u \in \mathbb{R}^N$  with  $||u||_2 = 1$ , by definition of  $\hat{g}_+^N$ :

$$\frac{\sigma}{N} \sum_{i,j} u_i u_j g_+^N(i,j) \le \frac{\sigma}{N} \sum_{i,j} u_i u_j \hat{g}_+^N(i,j) + \frac{C}{N} \le 1 - \delta + \frac{C}{N}.$$
 (A.31)

If N is large enough, the spectral radius of  $\frac{\sigma}{N}g_+^N$  is thus smaller than  $1 - \delta/2$  and the matrix  $I_N - \frac{\sigma}{N}g^N$  is invertible as claimed. We henceforth work with such an N.

For any  $J \subset \mathbb{T}_N$ , write  $\bar{\eta}_J$  for the product  $\prod_{j \in J} \bar{\eta}_j$ . The key part of the proof is the observation (A.32) below: the correlations  $\nu_g^N(\bar{\eta}_a\bar{\eta}_J)$  can be expressed in terms of all other correlations of the form  $\nu_g^N(\bar{\eta}_b\bar{\eta}_J)$  ( $b \in \mathbb{T}_N$ ), up to a small error that depends on the cardinal |J| of J. The claim of Proposition A.2 is then established by recursion on |J|.

Let  $n \in \mathbb{N}_{\geq 1}$ . The set I in the proposition will correspond here to sets of the form  $\{a\} \cup J$ with  $a \notin J$  and  $J \subset \mathbb{T}_N$  with |J| = 2n - 1. Let  $\bar{\eta}_{\cdot} \bar{\eta}_J$  denote the vector  $(\bar{\eta}_b \bar{\eta}_J)_{b \in \mathbb{T}_N}$ . Assume for the moment that we have proven the following averaging identity, for each J with |J| = 2n - 1and each  $a \notin J$ :

$$\nu_g^N(\bar{\eta}_a\bar{\eta}_J) = \left[\frac{\sigma}{N}g^N\nu_g^N(\bar{\eta}_.\bar{\eta}_J)\right](a) + \tilde{e}_N^J(a), \tag{A.32}$$

with  $\tilde{e}_N^{n,J} = (\tilde{e}_N^J(i))_{i \in \mathbb{T}_N}$  a vector satisfying:

$$\sup_{|J'|=2n-1} \sup_{i \notin J'} |\tilde{e}_N^{J'}(i)| = O(N^{-n-1}), \qquad \tilde{e}_N^{J'}(i) = 0 \quad (i \in J').$$
(A.33)

Let us then use (A.32) to prove the claim of Proposition A.2 by recursion. Fix  $J \subset \mathbb{T}_N$  with |J| = 2n - 1. Equation (A.32) is valid for  $a \notin J$ . Let us first extend it to general  $a \in \mathbb{T}_N$ . Using  $(\bar{\eta}_{\cdot})^2 = \sigma$ , with the convention  $\bar{\eta}_{\emptyset} := 1$ :

$$\nu_g^N(\bar{\eta}_a\bar{\eta}_J) = \mathbf{1}_J(a)\sigma\nu_g^N(\bar{\eta}_{J\backslash\{a\}}) + \mathbf{1}_{J^c}(a)\sigma\big[N^{-1}g^N\nu_g^N(\bar{\eta}_{J}\bar{\eta}_{J})\big](a) + \tilde{e}_N^J(a).$$

Rearranging terms, this becomes:

$$\begin{split} \left[ \left( \sigma^{-1} I_N - N^{-1} g^N \right) \nu_g^N(\bar{\eta}_. \bar{\eta}_J) \right] (a) \\ &= \mathbf{1}_J(a) \left( \nu_g^N(\bar{\eta}_{J \setminus \{a\}}) - \frac{1}{N} \sum_{i \neq a} g_{a,i} \nu_g^N(\bar{\eta}_i \bar{\eta}_J) \right) + \sigma^{-1} \tilde{e}_N^J(a) \\ &= \mathbf{1}_J(a) \left( \nu_g^N(\bar{\eta}_{J \setminus \{a\}}) + O(N^{-n}) \right) + O(N^{-n-1}), \end{split}$$
(A.34)

where the  $O(N^{-n}), O(N^{-n-1})$  are uniform on a, J as given respectively by Lemma A.1 and (A.33). We now proceed with the proof of the claim of Proposition A.2 by recursion assuming (A.32).

Take first n = 1,  $a \in \mathbb{T}_N$  and  $J = \{b\} \subset \mathbb{T}_N$ . We will make use of the following lemma, proven afterwards.

**Lemma A.3.** Let  $\mathcal{M}$  be the set of sequences of matrices on  $\mathbb{R}^{N \times N}$  indexed by  $N \in \mathbb{N}_{\geq 1}$  and defined as follows.  $(A_N)_{N \geq N_0} \in \mathcal{M}$  for some  $N_0 \in \mathbb{N}_{\geq 1}$  if and only if there are sequences  $(D_N)_N, (R_N)_N$  of matrices such that, for each  $N \geq N_0$ ,  $D_N$  is diagonal,  $R_N(i, i) = 0$  for  $1 \leq i \leq N$ , and  $A_N = D_N + N^{-1}R_N$ . Moreover,  $(R_N)_N, (D_N)_N$  are such that there is c > 0 with:

for each 
$$N \ge N_0$$
,  $\inf_{1 \le i \le N} |D_N(i,i)| \ge c$ ,  $\sup_{1 \le i,j \le N} |R_N(i,j)| \le c^{-1}$ . (A.35)

It then holds that if  $(A_N)_{N\geq N_0} \in \mathcal{M}$  and  $A_N$  is invertible  $(N\geq N_0)$ , then  $(A_N^{-1})_{N\geq N_0} \in \mathcal{M}$ .

By Lemma A.3, the matrix  $\sigma^{-1}I_N - N^{-1}g^N$  has inverse satisfying:

$$\sup_{i \in \mathbb{T}_N} \left| \left( \sigma^{-1} I_N - N^{-1} g^N \right)^{-1}(i,i) \right| = O_N(1), \quad \sup_{i \neq j \in \mathbb{T}_N} \left| \left( \sigma^{-1} I_N - N^{-1} g^N \right)^{-1}(i,j) \right| = O(N^{-1}).$$
(A.36)

The last equation and (A.34) with  $J = \{b\}$  yield:

$$\nu_g^N(\bar{\eta}_a\bar{\eta}_b) = \left(\sigma^{-1}I_N - N^{-1}g^N\right)^{-1}(a,b) + e_N^{\{b\}}(a), \tag{A.37}$$

with:

$$e_N^{\{b\}}(a) := \left(\sigma^{-1}I_N - N^{-1}g^N\right)^{-1}(a,b) \left[-\frac{1}{N}\sum_{i\neq b}g_{b,i}\nu_g^N(\bar{\eta}_i\bar{\eta}_b) + \sigma^{-1}\tilde{e}_N^{\{b\}}(b)\right].$$
 (A.38)

The error term indeed satisfies  $\sup_{a\neq b} |e_N^{\{b\}}(a)| = O(N^{-2})$  by (A.36). Assume now  $n \geq 1$ and the claim of Proposition A.2 to hold for each subset of  $\mathbb{T}_N$  containing 2n elements. Let  $J \subset \mathbb{T}_N$  with |J| = 2n + 1 and  $a \in \mathbb{T}_N$ . Then, by (A.34),

$$\nu_g^N(\bar{\eta}_a\bar{\eta}_J) = \sum_{j\in J} \left(\sigma^{-1}I_N - N^{-1}g^N\right)^{-1}(a,j)\nu_g^N(\bar{\eta}_{J\setminus\{j\}}) + e_N^J(a),$$
(A.39)

with  $e_N^J = (e_N^J(i))_{i \in \mathbb{T}_N}$  the vector given by:

$$e_N^{J\cup}(a) := \sum_{j \in J} \left( \sigma^{-1} I_N - N^{-1} g^N \right)^{-1}(a, j) \left[ -\frac{1}{N} \sum_{i \in \mathbb{T}_N} g_{j,i} \nu_g^N(\bar{\eta}_i \bar{\eta}_J) + \sigma^{-1} \tilde{e}_N^J(j) \right].$$
(A.40)

The recursion hypothesis and (A.36) imply that  $e_N^J(a) = O(N^{-n-2})$  uniformly on J, a. The recursion hypothesis applied to each  $J \setminus \{j\}$   $(j \in J)$  then turns (A.39) into:

$$\nu_g^N(\bar{\eta}_a\bar{\eta}_J) = \sum_{j\in J} \left(\sigma^{-1}I_N - N^{-1}g^N\right)(a,j) \sum_{p\in P_{|J\setminus\{j\}|}} \prod_{i\in J\setminus\{j\}} \left(\sigma^{-1}I_N - N^{-1}g^N\right)^{-1}(i,p(i)) + e_N^J(a).$$
(A.41)

Since the sum on  $j \in J$  and  $p \in P_{|J \setminus \{j\}|} = P_{2n}$  in the left-hand side exactly enumerates elements of the set  $P_{2n+2}$  of pairings of 2n + 2 elements, Proposition A.2 is proven assuming (A.32):

$$\nu_g^N(\bar{\eta}_a\bar{\eta}_J) = \sum_{p \in P_{2n+2}} \prod_{i \in J \cup \{a\}} \left( \sigma^{-1}I_N - N^{-1}g^N \right)^{-1}(i, p(i)) + e_N^J(a).$$
(A.42)

We now prove the identity (A.32). Let  $J \subset \mathbb{T}_N$  with |J| = 2n - 1  $(n \ge 1)$ , and let  $a \in \mathbb{T}_N \setminus J$ . Recall that  $\nu_g^N \propto \exp[2\Pi^N(g)]\nu_{1/2}^N$ , and decompose  $\Pi^N(g)$  as follows:

$$2\Pi^{N}(g) = \frac{\bar{\eta}_{a}}{N} \sum_{i \neq a} \bar{\eta}_{i} g_{a,i} + 2G_{\{a\}^{c}}(\eta), \qquad G_{\{a\}^{c}}(\eta) = \frac{1}{4N} \sum_{i \neq j \in \mathbb{T}_{N} \setminus \{a\}} \bar{\eta}_{i} \bar{\eta}_{j} g_{i,j}.$$
(A.43)

Using  $e^x = 1 + x \int_0^1 e^{tx} dx$  with  $x = \frac{\bar{\eta}_a}{N} \sum_{i \neq a} \bar{\eta}_i g_{a,i}$ , we find:

$$\mathcal{Z}_{g}^{N}\nu_{g}^{N}(\bar{\eta}_{a}\bar{\eta}_{J}) = \nu_{1/2}^{N} \left( e^{2G_{\{a\}^{c}}} \bar{\eta}_{J}\bar{\eta}_{a} \left[ 1 + \frac{\bar{\eta}_{a}}{N} \sum_{i \neq a} \bar{\eta}_{i}g_{a,i} \int_{0}^{1} \exp\left[\frac{t\bar{\eta}_{a}}{N} \sum_{i \neq a} \bar{\eta}_{i}g_{a,i}\right] dt \right] \right).$$
(A.44)

Under the product measure  $\nu_{1/2}^N$ ,  $\bar{\eta}_a$  is independent of  $\bar{\eta}_J$ ,  $G_{\{a\}^c}$  (and has average 0). It follows that  $\nu_{1/2}^N(\bar{\eta}_a\bar{\eta}_J\exp[2G_{\{a\}^c}])$  vanishes. Since also  $(\bar{\eta}_a)^2 = \sigma$ , (A.44) becomes, reconstructing  $2\Pi^N(g)$  from  $2G_{\{a\}^c}$  through (A.43):

$$\nu_g^N(\bar{\eta}_a\bar{\eta}_J) = \sigma\nu_g^N\left(\frac{\bar{\eta}_J}{N}\sum_{i\neq a}\bar{\eta}_i g_{a,i}\int_0^1 \exp\left[\frac{(t-1)\bar{\eta}_a}{N}\sum_{i\neq a}\bar{\eta}_i g_{a,i}\right]dt\right).$$
(A.45)

To obtain (A.32), we further expand the exponential in (A.45), using the following formula: for each  $t \in [0, 1]$ ,

$$\exp\left[\frac{(t-1)\bar{\eta}_a}{N}\sum_{i\neq a}\bar{\eta}_i g_{a,i}\right] = 1 + T_t^1(\eta) + T_t^2(\eta),$$
(A.46)

with:

$$T_t^1(\eta) := \sum_{p=1}^{2n} \frac{1}{p!} \left( (t-1)\bar{\eta}_a \right)^p \left( \frac{1}{N} \sum_{i \neq a} \bar{\eta}_i g_{a,i} \right)^p$$
$$T_t^2(\eta) := \int_0^1 \frac{s^{2n}}{(2n)!} \left( (t-1)\bar{\eta}_a \right)^{2n+1} \left( \frac{1}{N} \sum_{i \neq a} \bar{\eta}_i g_{a,i} \right)^{2n+1} \exp\left[ \frac{s(t-1)\bar{\eta}_a}{N} \sum_{i \neq a} \bar{\eta}_i g_{a,i} \right] ds. \quad (A.47)$$

Injecting this decomposition in (A.45) yields:

$$\nu_g^N(\bar{\eta}_a\bar{\eta}_J) = \left[\frac{\sigma}{N}g^N\nu_g^N(\bar{\eta}_{\cdot}\bar{\eta}_J)\right](a) + \sigma\nu_g^N\left(\int_0^1\frac{\bar{\eta}_J}{N}\sum_{i\neq a}\bar{\eta}_i g_{a,i}(T_t^1(\eta) + T_t^2(\eta))\,dt\right).$$
(A.48)

The first term in the right-hand side above is exactly the first term in the right-hand side of (A.32). We thus need to prove that the contributions of  $T_t^1, T_t^2$  are of order  $O(N^{-n-1})$  uniformly on J, a.

Both  $T_t^1$  and  $T_t^2$  are readily estimated through Lemma A.1. For  $T_t^1$ , one has for each  $t \in [0, 1]$  and each  $1 \le p \le 2n$ :

$$\nu_g^N \left( \frac{\bar{\eta}_J}{N} \sum_{i \neq a} \bar{\eta}_i g_{a,i} \cdot \frac{1}{p!} \left( (t-1)\bar{\eta}_a \right)^p \left( \frac{1}{N} \sum_{i \neq a} \bar{\eta}_i g_{a,i} \right)^p \right) = \begin{cases} O(N^{-n-(p+1)/2}) & \text{if } p \text{ is odd,} \\ O(N^{-n-p/2}) & \text{if } p \text{ is even,} \end{cases}$$
(A.49)

and the estimate above is uniform in  $t \in [0, 1]$ ,  $a \in \mathbb{T}_N$  and sets J with cardinal 2n - 1.

For  $T_t^2$ , bounding  $\bar{\eta}_a, \bar{\eta}_J$  by 1 and the exponential by  $e^{\|g\|_{\infty}}$ , one similarly concludes on the existence of  $C_{2n+1} > 0$  such that:

$$\left| \nu_{g}^{N} \left[ \bar{\eta}_{J} \frac{\bar{\eta}_{a}}{N} \sum_{i \neq a} \bar{\eta}_{i} g_{a,i} \int_{0}^{1} T_{t}^{2}(\eta) dt \right] \right| \\
\leq C_{2n+1} e^{\|g\|_{\infty}} \nu_{g}^{N} \left[ \left( \frac{1}{N} \sum_{i \neq a} \bar{\eta}_{i} g_{a,i} \right)^{2n+2} \right] = O(N^{-n-1}). \quad (A.50)$$

The error term is again uniform on t, a, J. This concludes the proof of (A.32):

$$\nu_g^N(\bar{\eta}_a\bar{\eta}_J) = \sigma \left[ N^{-1}g^N \nu_g^N(\bar{\eta}_{\cdot}\bar{\eta}_J) \right](a) + \tilde{e}_N^J(a), \tag{A.51}$$

where  $\tilde{e}_N^J$  satisfies  $\sup_{|J|=2n-1} \sup_{i \notin J} |\tilde{e}_N^J(i)| = O(N^{-n-1})$  and is defined as follows:

$$\tilde{e}_{N}^{J}(a) := \mathbf{1}_{J^{c}}(a) \int_{0}^{1} \nu_{g}^{N} \left[ \bar{\eta}_{J} \frac{\bar{\eta}_{a}}{N} \sum_{i \neq a} \bar{\eta}_{i} g_{a,i} \left( T_{t}^{1}(\eta) + T_{t}^{2}(\eta) \right) \right] dt.$$
(A.52)

Proof of Lemma A.3. Let  $N_0 \in \mathbb{N}_{\geq 1}$  and  $(A_N)_{N \geq N_0} \in \mathcal{M}$  be a family of invertible matrices. Then  $(A_N(i,j))_{1 \leq i,j \leq n}$  is also invertible for each  $n \leq N$ . This implies (see Theorem 4.3.1 in [20]) the existence of a lower triangular matrix  $L_N$  with diagonal equal to 1, and an upper triangular matrix  $U_N$  such that:

$$A_N = L_N U_N, \qquad N \ge N_0. \tag{A.53}$$

 $L_N$  is invertible by assumption, thus  $U_N$  is invertible by invertibility of  $A_N$ . The claim of Lemma A.3 therefore boils down to proving that  $(U_N^{-1}L_N^{-1})_N \in \mathcal{M}$ . Let us first prove that  $(L_N)_N, (U_N)_N \in \mathcal{M}$ . The coefficients of these matrices can be computed explicitly: for each  $N \geq N_0$  and each  $1 \leq i \leq 2N - 1$ ,

$$U_N(i,j) = A_N(i,j) - \sum_{\ell=1}^{i-1} L_N(i,\ell) U_N(\ell,j) \quad \text{for } j \ge i,$$
(A.54)

$$L_N(j,i) = \frac{1}{U_N(i,i)} \Big( A_N(j,i) - \sum_{\ell=1}^{i-1} L_N(j,\ell) U_N(\ell,i) \Big) \quad \text{for } j > i.$$
(A.55)

In particular, one can check that  $(L_N)_N, (U_N)_N$  indeed belong to  $\mathcal{M}$ .

Let us now check that  $(L_N^{-1})_N, (U_N^{-1})_N$  also belong to  $\mathcal{M}$ . The inverse  $L_N^{-1}$  of  $L_N$  is also lower triangular, and can be computed straightforwardly: for  $N \ge N_0$ ,

$$\forall i \ge j, \qquad [L_N^{-1}](i,j) = \frac{1}{A_N(i,i)} \Big( \delta_{i,j} - \sum_{\ell=1}^{i-1} L_N(i,\ell) [L_N^{-1}](\ell,j) \Big), \tag{A.56}$$

where:

$$\delta_{i,j} := \begin{cases} 1 & \text{if } i = j, \\ 0 & \text{otherwise.} \end{cases}$$
(A.57)

The same result holds for the upper-triangular  $U_N$ . A recursion thus gives  $(L_N^{-1})_N, (U_N^{-1})_N \in \mathcal{M}$ . It remains to notice that  $\mathcal{M}$  is stable under matrix product:  $(B_N)_N, (C_N)_N \in \mathcal{M}$  implies  $(B_N C_N)_N \in \mathcal{M}$ . This concludes the proof of Lemma A.3.

The following proposition gives exponential concentration results under  $\nu_g^N$ . These are useful when applying the entropy inequality.

**Proposition A.4.** Assume that g satisfies the hypotheses of Lemma A.1. Let  $J \subset \mathbb{Z}$  contain  $0, n \in \mathbb{N}$  and  $\phi_n : \mathbb{T}_N^n \to \mathbb{R}$ . We assume N to be large enough that J can be viewed as a subset of  $\mathbb{T}_N$ . Define the functions:

$$X_{\phi_n}^{n,J}(\eta) := \sum_{(i_1,...,i_n) \in (\mathbb{T}_N)^n} \phi_n(i_1,...,i_n) \Big(\prod_{j \in J} \bar{\eta}_{i_1+j}\Big) \bar{\eta}_{i_2}...\bar{\eta}_{i_n}, \qquad \eta \in \Omega_N.$$
(A.58)

Let  $W_{\phi_n}^{n,J} := N^{-(n-1)} X_{\phi_n}^{n,J}$  if  $n \ge 2$ , and  $W_{\phi_1}^{1,J} := N^{-1/2} X_{\phi_1}^{0,J}$ . Then, for any  $n \ge 0$ , there are constants  $\gamma^{n,J}, C^{n,J} > 0$  depending on n, J but independent

Then, for any  $n \ge 0$ , there are constants  $\gamma^{n,J}$ ,  $C^{n,J} > 0$  depending on n, J but independent of  $\phi_n, N$ , such that:

$$\forall N \in \mathbb{N}_{\geq 1}, \forall \gamma < \gamma^{n,J}, \qquad \log \nu_g^N \left[ \exp \left[ \frac{\gamma |W_{\phi_n}^{n,J}|}{\sup_N \|\phi_n\|_\infty} \right] \right] \leq C^{1,J} \mathbf{1}_{n=1} + \mathbf{1}_{n\geq 2} \frac{C^{n,J}}{N^{\frac{n-2}{2}}}.$$
(A.59)

*Proof.* The proof can be found in Theorem A.3 and Corollary A.4 in [13]. It is written for negative kernels only but this is just used to reduce to the case g = 0. The assumption that  $\sigma g_+$  has largest eigenvalue  $\lambda_{\max}$  strictly below 1 similarly enables one to reduce to this case. Indeed, for  $\varepsilon_g > 0$  such that  $(1 + \varepsilon_g)\lambda_{\max} < 1$ ,  $\sup_N \mathcal{Z}^N_{(1+\varepsilon_g)g} < \infty$  by Lemma A.1. The identity  $\mathbb{E}[X] = \int_0^\infty \mathbb{P}(X > t) dt$  for a non-negative random variable X and a change of variable thus give:

$$\nu_g^N \left[ \exp\left[\frac{\gamma |W_{\phi_n}^{n,J}|}{\sup_N \|\phi_n\|_{\infty}}\right] \right] \le 1 + \int_0^\infty e^t \nu_g^N \left(\frac{\gamma |W_{\phi_n}^{n,J}|}{\sup_N \|\phi_n\|_{\infty}} > t\right) dt$$
$$\le 1 + \left(\mathcal{Z}_{(1+\varepsilon_g)g}^N\right)^{\frac{1}{1+\varepsilon_g}} \int_0^\infty e^t \nu_{1/2}^N \left(\frac{\gamma |W_{\phi_n}^{n,J}|}{\sup_N \|\phi_n\|_{\infty}} > t\right)^{\frac{\varepsilon_g}{1+\varepsilon_g}} dt. \quad (A.60)$$

## A.2 Concentration for the uniform measure

Recall that  $\Omega_{N,m}$  is the set of configurations with m particles on the torus  $\mathbb{T}_N$   $(0 \le m \le N)$ . Recall also that  $U_m$  denotes the uniform measure on  $\Omega_{N,m}$ . Concentration results under  $U_m$  are given in Lemma A.6, relying on the following logarithmic Sobolev inequality.

**Lemma A.5** (Theorem 4 in [41]). For  $m \in \{0, ..., N\}$ , let  $D_{ex}^m$  denote the Dirichlet form of symmetric simple exclusion with m particles:

$$D_{ex}^{m}(g) = \frac{1}{2} U_m \Big[ \sum_{i \in \mathbb{T}_N} \Big[ g(\eta^{i,i+1}) - g(\eta) \Big]^2 \Big], \qquad g : \Omega_{N,m} \to \mathbb{R}.$$
(A.61)

There is  $C_{LS} > 0$  independent of m, N such that, for any density f for  $U_m$ :

$$U_m[f\log f] \le C_{LS} N^2 D_{ex}^m(\sqrt{f}). \tag{A.62}$$

For  $\phi : \mathbb{T}_N \to \mathbb{R}$ , define its discrete gradient at point  $i \in \mathbb{T}_N$  by:

$$\partial^N \phi(i) = N[\phi(i+1) - \phi(i)]. \tag{A.63}$$

The log-Sobolev inequality provides concentration results for Lipschitz functions.

**Lemma A.6.** Let  $\phi : \mathbb{T}_N \to \mathbb{R}$  be a Lipschitz function, in the sense that:

$$\sup_{N} \|\partial^{N}\phi\|_{\infty} < \infty.$$
 (A.64)

Similarly to (A.4), define:

$$W_m(\eta) = W^{1,\{0\}}_{\phi,m}(\eta) := \frac{1}{N^{1/2}} \sum_{i \in \mathbb{T}_N} \phi(i)\bar{\eta}^m_i.$$
 (A.65)

There is then  $\gamma_0 > 0$  independent of  $m, \phi$  such that, for any  $\gamma < \gamma_0$ :

$$U_m \left( \exp\left[\frac{\gamma(W_m)^2}{\|\partial^N \phi\|_{\infty}^2}\right] \right) \le 3.$$
 (A.66)

*Proof.* Note that, for any  $i \in \mathbb{T}_N$ , one has:

$$W_m(\eta^{i,i+1}) - W_m(\eta) = (\eta_i - \eta_{i+1}) \frac{\partial^N \phi(i)}{N^{3/2}}.$$
 (A.67)

The Herbst argument (see e.g. Proposition 2.3 in Ledoux's course [40]) then gives, for some C > 0 independent of  $m, \phi$ :

$$U_m\left(\exp\left[\frac{\theta W_m}{\|\partial^N \phi\|_{\infty}}\right]\right) \le e^{C\theta^2/2}, \qquad \theta \in \mathbb{R}.$$
 (A.68)

The identity  $\mathbb{E}[e^X] \leq 1 + \int_0^\infty e^t \mathbb{P}(X > t) dt$  applied to  $X = \gamma(W_m^2) / \|\partial^N \phi\|_\infty^2$  for  $\gamma$  small enough depending only on C and a Chernov bound conclude the proof.

## **B** Integration by parts formula

Let  $\tilde{\lambda} \in \mathbb{R}$  and  $\tilde{h} \in C_D^{\infty}(\mathbb{T}^2)$  be symmetric. In this section, we provide an integration by parts formula under  $\nu_{\tilde{h}}^N$  and  $U_{m,\tilde{h}}$  for  $0 \leq m \leq N$ . Let  $f : \Omega_N \to \mathbb{R}$ , and define  $\Gamma_{\tilde{h},\tilde{\lambda}}^{i,i+1}(f)$  for  $i \in \mathbb{T}_N$ as the quantity:

$$\Gamma_{\tilde{h},\tilde{\lambda}}^{i,i+1}(f)(\eta) := \frac{1}{2} c_{\tilde{h},\tilde{\lambda}}(\eta,\eta^{i,i+1}) \left[ \nabla_{i,i+1} f(\eta) \right]^2, \tag{B.1}$$

with:

$$\nabla_{i,i+1} f(\eta) = f(\eta^{i,i+1}) - f(\eta), \qquad \eta \in \Omega_N.$$
(B.2)

**Lemma B.1.** (Integration by parts under  $\nu_{\tilde{h}}^N$ ) Let  $i \in \mathbb{T}_N$ , let f be a density for  $\nu_{\tilde{h}}^N$  and let  $u: \Omega_N \to \mathbb{R}$  satisfy  $u(\eta^{i,i+1}) = u(\eta)$ , i.e. u is invariant under a jump between sites i, i + 1. There is then a constant  $C = C(\tilde{h}, \tilde{\lambda}) > 0$  such that, for any  $\alpha > 0$ :

$$\nu_{\bar{h}}^{N} \left( f u(\bar{\eta}_{i+1} - \bar{\eta}_{i}) \right) \leq \alpha N^{2} \nu_{\bar{h}}^{N} \left( \Gamma_{\bar{h},\tilde{\lambda}}^{i,i+1}(\sqrt{f}) \right) + \frac{C}{\alpha N^{2}} \nu_{\bar{h}}^{N} \left( f |u|^{2} \right) \\ + \nu_{\bar{h}}^{N} \left( f u(\bar{\eta}_{i+1} - \bar{\eta}_{i}) \left( \exp \left[ -\frac{2(\bar{\eta}_{i+1} - \bar{\eta}_{i})}{N} B_{i}^{\tilde{h}} \right] - 1 \right) \right).$$
(B.3)

**Remark B.2.** The claim of the lemma also holds if  $\nu_{\tilde{h}}^N$  is replaced by the uniform measure  $U_m$  on configurations with  $0 \le m \le N$  particles, with the same proof.

*Proof.* A direct computation using the fact that  $\eta \mapsto \eta^{i,i+1}$  is bijective gives:

$$\nu_{\tilde{h}}^{N} \left( f u(\bar{\eta}_{i+1} - \bar{\eta}_{i}) \right) = \frac{1}{2} \sum_{\eta \in \Omega_{N}} \nu_{\tilde{h}}^{N}(\eta) u(\eta) (\bar{\eta}_{i+1} - \bar{\eta}_{i}) \left[ f(\eta) - \frac{\nu_{\tilde{h}}^{N}(\eta^{i,i+1})}{\nu_{\tilde{h}}^{N}(\eta)} f(\eta^{i,i+1}) \right]$$
(B.4)

and, for each  $\eta \in \Omega_N$ :

$$\frac{\nu_{\tilde{h}}^{N}(\eta^{i,i+1})}{\nu_{\tilde{h}}^{N}(\eta)} = \exp\left[-\frac{2(\bar{\eta}_{i+1} - \bar{\eta}_{i})}{N}B_{i}^{\tilde{h}}\right], \qquad B_{i}^{\tilde{h}} = \frac{1}{2N}\sum_{j\notin\{i,i+1\}}\bar{\eta}_{j}\partial_{1}^{N}\tilde{h}_{i,j}. \tag{B.5}$$

Thus, writing  $e^x = 1 + e^x - 1$  for  $x = -2N^{-1}(\bar{\eta}_{i+1} - \bar{\eta}_i)B_i^{\tilde{h}}$  and again changing variables between  $\eta$  and  $\eta^{i,i+1}$ :

$$\nu_{\tilde{h}}^{N} \left( fu(\bar{\eta}_{i+1} - \bar{\eta}_{i}) \right) \le \nu_{\tilde{h}}^{N} \left( |\nabla_{i,i+1}f| |u| \right) + \nu_{\tilde{h}}^{N} \left( fu(\bar{\eta}_{i+1} - \bar{\eta}_{i}) \left( e^{-\frac{2(\bar{\eta}_{i+1} - \bar{\eta}_{i})}{N} B_{i}^{\tilde{h}}} - 1 \right) \right).$$
(B.6)

Note that jump rates are bounded below:

$$c_{\tilde{h},\tilde{\lambda}}(\eta,\eta_{i+1}) \ge c_0 c(\eta,\eta_{i+1}), \qquad \eta \in \Omega_N, i \in \mathbb{T}_N.$$
(B.7)

The claim then follows from the lower bound on the jump rates, the fact that  $\sup_{N,i} |B_i^{\tilde{h}}| < \infty$ and the inequality  $|ab| \leq \frac{\kappa}{2}a^2 + \frac{1}{2\kappa}b^2$  applied to  $\kappa = 2\alpha N^2/c_0$ ,  $a = \sqrt{f(\eta^{i,i+1})} - \sqrt{f(\eta)}$  and  $b = u(\sqrt{f(\eta^{i,i+1})} + \sqrt{f(\eta)})$ .

# C On some partial differential equations involving the bias h

## C.1 Existence and regularity of the bias h

In this section, we prove Proposition 2.1, i.e. the existence and properties of  $h \in C(\mathbb{T}) \cap C^{\infty}([0,1])$  solving:

$$\begin{cases} h''(x) - \frac{\sigma}{2} \int_{\mathbb{T}} h'(x-y)h'(y) \, dy = 0 \quad \text{for } x \in (0,1), \\ h'(0_+) - h'(1_-) = 2\lambda(\lambda + 2E). \end{cases}$$
(C.1)

Define:

$$\mathcal{H} := \left\{ \psi \in \mathbb{H}^1(\mathbb{T}) : \int_{\mathbb{T}} \psi(x) \, dx = 0 \right\}.$$
(C.2)

Then  $\mathcal{H}$  equipped with the norm  $\psi \mapsto \|\psi'\|_2$  is a Hilbert space. Upon integrating (C.1) against a test function  $f \in C^{\infty}(\mathbb{T})$  and integrating by parts, (C.1) becomes:

$$\int_{\mathbb{T}} h(x) f''(x) \, dx - 2\lambda(\lambda + 2E) f(0) - \frac{\sigma}{2} \int_{\mathbb{T}^2} f''(x) h(x - y) h(y) \, dx \, dy = 0.$$
(C.3)

**Proposition C.1.** Recall the definition of sub-criticality in Proposition 2.1. There is a unique family  $(h_{\lambda,E})_{\lambda,E}$  sub-critical of solutions of (2.21) in  $\mathcal{H}$ . These solutions are in  $C^0(\mathbb{T}) \cap C^\infty([0,1])$  and:

 $(\lambda, E) \mapsto ||h_{\lambda,E}||_2$  is a continuous function vanishing on the line  $\lambda = 0.$  (C.4)

For each sub-critical  $\lambda, E$ , the Fourier series of  $h_{\lambda,E}$  reads:

$$\forall x \in \mathbb{T}, \qquad h(x) = \frac{\sqrt{2}}{\sigma} \sum_{\ell \ge 1} \left( 1 - \left[ 1 + \frac{\sigma\lambda(\lambda + 2E)}{\pi^2 \ell^2} \right]^{1/2} \right) \sqrt{2} \cos(2\pi\ell x). \tag{C.5}$$

Moreover, the kernel  $(x, y) \mapsto h_{\lambda, E}(x - y)$  is positive if  $\lambda(\lambda + 2E) \leq 0$ , negative otherwise, with eigenvalues given by its Fourier coefficients.

*Proof.* We look for a solution h to (C.3) in  $\mathbb{L}^2(\mathbb{T})$  by computing its Fourier decomposition. For  $\ell \in \mathbb{Z}$ , let  $c_{\ell}(h) = \int_{\mathbb{T}} h(x)e^{-2\iota\pi\ell x} dx$ , with  $\iota^2 = -1$ . We look for h with  $c_0(h) = 0$  by assumption. Note that, for any  $\psi, \tilde{\psi} \in \mathbb{L}^2(\mathbb{T})$ , one has:

$$c_{\ell} \left( \int_{\mathbb{T}} \psi(x-y) \tilde{\psi}(y) \, dy \right) = c_{\ell}(\psi) c_{\ell}(\tilde{\psi}).$$
 (C.6)

Taking  $f = e^{-2\iota \pi \ell}$  for  $\ell \neq 0$ , Equation (C.3) becomes:

$$-(2\pi\ell)^2 c_\ell(h) - 2\lambda(\lambda + 2E) + \frac{\sigma}{2}(2\pi\ell)^2 c_\ell(h)^2 = 0.$$
 (C.7)

This gives:

$$c_{\ell}(h) = \frac{1}{\sigma} \left( 1 + \eta_{\ell} \sqrt{1 + \frac{\sigma \lambda(\lambda + 2E)}{\pi^2 \ell^2}} \right), \qquad \eta_{\ell} \in \{-1, 1\}.$$
(C.8)

Note that  $h \in \mathbb{L}^2(\mathbb{T}) \supset \mathcal{H}$  only if all  $\eta_\ell$  are equal to -1 for  $\ell$  large enough. In addition, we must have  $\eta_\ell = -1$  for any  $\ell \neq 1$  in order for h to vanish on the line  $\lambda = 0$ . This gives existence and uniqueness of a family  $(h_{\lambda,E})$  of solutions to (C.3) in  $\mathcal{H}$  such that  $(\lambda, E) \mapsto ||h_{\lambda,E}||_2$ is continuous and vanishes on the line  $\lambda = 0$ . Note further that  $c_\ell(h_{\lambda,E}) = o(\ell^{-2})$ , thus  $h_{\lambda,E} \in \mathbb{H}^1(\mathbb{T}) \subset C^0(\mathbb{T})$  by Dirichlet's normal convergence theorem and (C.5) holds pointwise. Moreover, the Fourier coefficients  $c_\ell(h_{\lambda,E})$  are the eigenvalues of the kernel operator  $h_{\lambda,E}$  all have the same sign, the sign of  $-\lambda(\lambda + 2E)$ .

It remains to observe that  $h_{\lambda,E} \in C^{\infty}([0,1])$ . Indeed, defining  $h''_{\lambda,E}$  on (0,1) through (C.1) gives  $h''_{\lambda,E} \in \mathbb{H}^1((0,1))$ , thus  $h'_{\lambda,E} \in \mathbb{H}^2((0,1))$ , with  $\mathbb{H}^n((0,1))$  the usual Sobolev space of functions with n weak derivatives in  $\mathbb{L}^2((0,1))$ . Iterating we get  $h_{\lambda,E} \in \mathbb{H}^n((0,1))$  for any  $n \in \mathbb{N}$ , thus  $h_{\lambda,E} \in C^{\infty}([0,1])$  by Sobolev embedding (Theorem 4.12 in [1]).

## C.2 Forwards and backwards Fokker-Planck equations

Here, we establish well-posedness and regularity for solutions of certain Fokker-Planck equations. These are used to characterise limiting fluctuations in Section 3.1.2.

For a symmetric  $h \in C_D^{\infty}(\mathbb{T}^2)$ , recall that  $\mathcal{L}_{\tilde{h}}$  is the operator acting on  $f \in C^{\infty}(\mathbb{T})$  according

to:

$$\mathcal{L}_{\tilde{h}}f(x) = f''(x) + \sigma \int_{\mathbb{T}} \partial_2 \tilde{h}(x, y) f'(y) dy$$
  
=  $f''(x) - \sigma m_{\tilde{h}}f(x) - \sigma \int_{\mathbb{T}} \partial_2^2 \tilde{h}(x, y) f(y) dy, \qquad x \in \mathbb{T},$  (C.9)

with  $m_{\tilde{h}} \in C^{\infty}(\mathbb{T})$  the function:

$$m_{\tilde{h}}(x) := \left(\partial_1 - \partial_2\right) \tilde{h}(x_-, x), \quad x \in \mathbb{T}.$$
 (C.10)

For  $f \in C^{\infty}(\mathbb{T})$  and  $t \geq 0$ , consider the following forwards and backwards Fokker-Planck equations:

$$\begin{cases} \partial_s u_s - \mathcal{L}_{\tilde{h}} u_s = 0 & \text{for } s \le t, \\ u_0 = f, \end{cases} \qquad \begin{cases} \partial_s v_s + \mathcal{L}_{\tilde{h}} v_s = 0 & \text{for } s \le t, \\ v_t = f. \end{cases}$$
(C.11)

**Proposition C.2.** Let  $f \in C^{\infty}(\mathbb{T})$  and  $t \geq 0$ . Then (C.11) admits solutions u, v in  $C^{\infty}([0, t] \times \mathbb{T})$ .

Proof. Let  $f \in C^{\infty}(\mathbb{T})$ . The Laplacian operator  $\Delta$  generates a strongly continuous semi-group on each of the Sobolev spaces  $\mathbb{H}^{\ell}(\mathbb{T})$  of functions with  $\ell$  weak derivatives in  $\mathbb{L}^{2}(\mathbb{T})$  ( $\ell \in \mathbb{N}$ ). On the other hand,  $\tilde{h} \in C_{D}^{\infty}(\mathbb{T}^{2})$  means that the operator  $f \mapsto m_{\tilde{h}}f - \sigma \partial_{2}^{2}\tilde{h}f$  is bounded in  $\mathbb{H}^{\ell}(\mathbb{T})$ . By Theorem III.1.3 in [25], it follows that  $\mathcal{L}_{\tilde{h}}$  generates a strongly continuous semi-group  $(\mathcal{P}_{\tilde{h}}^{*}(s))_{s\geq 0}$  on each  $\mathbb{H}^{\ell}(\mathbb{T})$  ( $\ell \in \mathbb{N}$ ). In particular,  $(u_{s})_{s\leq t} := (\mathcal{P}_{\tilde{h}}^{\infty}(s)f)_{s\leq t}$  is in  $C^{\infty}([0,t] \times \mathbb{T})$  and solves the first equation in (C.11). Setting  $v_{s} := u_{t-s}$  for  $s \leq t$  yields a solution of the second equation in (C.11) with the desired regularity.  $\Box$ 

## References

- [1] R. A. Adams and J. J. Fournier. *Sobolev Spaces*. Elsevier, 2003.
- [2] T. Agranov, M. E. Cates, and R. L. Jack. Tricritical Behavior in Dynamical Phase Transitions. *Physical Review Letters*, 131(1):017102, July 2023.
- [3] Y. Baek, Y. Kafri, and V. Lecomte. Dynamical Symmetry Breaking and Phase Transitions in Driven Diffusive Systems. *Physical Review Letters*, 118(3):030604, Jan. 2017.
- [4] L. Bertini, A. De Sole, D. Gabrielli, G. Jona-Lasinio, and C. Landim. Large deviations for the boundary driven symmetric simple exclusion process. *Mathematical Physics*, *Analysis and Geometry*, 6(3):231-267, 2003.
- [5] L. Bertini, A. De Sole, D. Gabrielli, G. Jona-Lasinio, and C. Landim. Current Fluctuations in Stochastic Lattice Gases. *Physical Review Letters*, 94(3):030601, Jan. 2005.
- [6] L. Bertini, A. De Sole, D. Gabrielli, G. Jona-Lasinio, and C. Landim. Large Deviations of the Empirical Current in Interacting Particle Systems. *Theory of Probability & Its Applications*, 51(1):2–27, Jan. 2007.
- [7] L. Bertini, A. De Sole, D. Gabrielli, G. Jona-Lasinio, and C. Landim. Macroscopic fluctuation theory. *Reviews of Modern Physics*, 87(2):593-636, June 2015.
- [8] L. Bertini, D. Gabrielli, and C. Landim. Large deviations for diffusions: Donsker-Varadhan meet Freidlin-Wentzell, Nov. 2022.
- [9] L. Bertini, D. Gabrielli, and C. Landim. Concurrent Donsker-Varadhan and hydrodynamical large deviations. *The Annals of Probability*, 51(4):1298-1341, July 2023.
- [10] L. Bertini, C. Landim, and M. Mourragui. Dynamical large deviations for the boundary driven weakly asymmetric exclusion process. *The Annals of Probability*, 37(6):2357–2403, Nov. 2009.
- [11] L. Bertini, A. D. Sole, D. Gabrielli, G. Jona-Lasinio, and C. Landim. Non Equilibrium Current Fluctuations in Stochastic Lattice Gases. *Journal of Statistical Physics*, 123(2):237-276, Apr. 2006.
- [12] P. Billingsley. Convergence of Probability Measures, 2nd Edition. Wiley, July 1999.
- [13] T. Bodineau and B. Dagallier. Large deviations for out of equilibrium correlations in the symmetric simple exclusion process, Dec. 2022.
- [14] T. Bodineau and B. Derrida. Distribution of current in nonequilibrium diffusive systems and phase transitions. *Physical Review E*, 72(6):066110, Dec. 2005.
- [15] T. Bodineau, B. Derrida, V. Lecomte, and F. van Wijland. Long Range Correlations and Phase Transitions in Non-equilibrium Diffusive Systems. *Journal of Statistical Physics*, 133(6):1013–1031, Dec. 2008.
- [16] T. Bodineau and M. Lagouge. Large deviations of the empirical currents for a boundarydriven reaction diffusion model. *The Annals of Applied Probability*, 22(6):2282–2319, 2012.
- [17] R. Chetrite and H. Touchette. Nonequilibrium Microcanonical and Canonical Ensembles and Their Equivalence. *Physical Review Letters*, 111(12):120601, Sept. 2013.
- [18] R. Chetrite and H. Touchette. Nonequilibrium Markov Processes Conditioned on Large Deviations. Annales Henri Poincaré, 16(9):2005–2057, Sept. 2015.
- [19] R. Chetrite and H. Touchette. Variational and optimal control representations of conditioned and driven processes. *Journal of Statistical Mechanics: Theory and Experiment*, 2015(12):P12001, Dec. 2015.
- [20] P. G. Ciarlet. Introduction to Numerical Linear Algebra and Optimisation. Cambridge Texts in Applied Mathematics. Cambridge University Press, Cambridge, 1989.
- [21] A. De Masi, N. Ianiro, and E. Presutti. Small deviations from local equilibrium for a process which exhibits hydrodynamical behavior. I. Journal of Statistical Physics, 29(1):57-79, Sept. 1982.

- [22] B. Derrida and T. Sadhu. Large deviations conditioned on large deviations I: Markov chain and Langevin equation. *Journal of Statistical Physics*, 176(4):773-805, Aug. 2019.
- [23] B. Derrida and T. Sadhu. Large Deviations Conditioned on Large Deviations II: Fluctuating Hydrodynamics. Journal of Statistical Physics, 177(1):151–182, Oct. 2019.
- [24] M. D. Donsker and S. R. S. Varadhan. Asymptotic evaluation of certain markov process expectations for large time, I. Communications on Pure and Applied Mathematics, 28(1):1–47, 1975.
- [25] K.-J. Engel and R. Nagel. One-Parameter Semigroups for Linear Evolution Equations. Graduate Texts in Mathematics. Springer-Verlag, New York, 2000.
- [26] C. P. Espigares, P. L. Garrido, and P. I. Hurtado. Dynamical phase transition for current statistics in a simple driven diffusive system. *Physical Review E*, 87(3):032115, Mar. 2013.
- [27] R. M. L. Evans. Rules for Transition Rates in Nonequilibrium Steady States. Physical Review Letters, 92(15):150601, Apr. 2004.
- [28] J. Farfan, C. Landim, and M. Mourragui. Hydrostatics and dynamical large deviations of boundary driven gradient symmetric exclusion processes. *Stochastic Processes and their Applications*, 121(4):725–758, Apr. 2011.
- [29] J. C. Ferreira and V. A. Menegatto. Eigenvalues of Integral Operators Defined by Smooth Positive Definite Kernels. *Integral Equations and Operator Theory*, 64(1):61–81, May 2009.
- [30] P. Gonçalves and M. Jara. Quadratic fluctuations of the symmetric simple exclusion. ALEA, 16:605-632, 2019.
- [31] R. A. Holley and D. W. Stroock. Generalized Ornstein-Uhlenbeck Processes and Infinite Particle Branching Brownian Motions. *Publications of the Research Institute for Mathematical Sciences*, 14(3):741–788, 1978.
- [32] R. L. Jack. Ergodicity and large deviations in physical systems with stochastic dynamics. The European Physical Journal B, 93(4):74, Apr. 2020.
- [33] R. L. Jack and P. Sollich. Large Deviations and Ensembles of Trajectories in Stochastic Models. Progress of Theoretical Physics Supplement, 184:304–317, Mar. 2010.
- [34] R. L. Jack and P. Sollich. Effective interactions and large deviations in stochastic processes. The European Physical Journal Special Topics, 224(12):2351–2367, Sept. 2015.
- [35] J. Jacod and A. N. Shiryaev. Limit Theorems for Stochastic Processes. Grundlehren Der Mathematischen Wissenschaften. Springer-Verlag, Berlin Heidelberg, 2 edition, 2003.
- [36] M. Jara and O. Menezes. Non-equilibrium fluctuations of interacting particle systems. arxiv preprint arXiv:1810.09526, 2018.

- [37] M. Jara and O. Menezes. Non-equilibrium fluctuations for a reaction-diffusion model via relative entropy. *Markov Processes and Related Fields*, 26(1):95–124, 2020.
- [38] C. Kipnis and C. Landim. Scaling Limits of Interacting Particle Systems. Springer Berlin Heidelberg, 1999.
- [39] S. Kusuoka and Y. Tamura. Gibbs measures for mean field potentials. Journal of the Faculty of Science. University of Tokyo. Section IA. Mathematics, 31(1):223-245, 1984.
- [40] M. Ledoux. Concentration of measure and logarithmic Sobolev inequalities. In J. Azéma, M. Émery, M. Ledoux, and M. Yor, editors, *Séminaire de Probabilités XXXIII*, Lecture Notes in Mathematics, pages 120–216, Berlin, Heidelberg, 1999. Springer.
- [41] T.-Y. Lee and H.-T. Yau. Logarithmic Sobolev inequality for some models of random walks. The Annals of Probability, 26(4):1855–1873, Oct. 1998.
- [42] T. Lengyel. The moments of the number of cycles of a random permutation by simple enumeration. Sankhyā. Series A. Methods and Techniques, 59(1):133-137, 1997.
- [43] I. Mitoma. Tightness of Probabilities On C([0, 1]; Y') and D([0, 1]; Y'). The Annals of Probability, 11(4):989-999, 1983.
- [44] H. Spohn. Long range correlations for stochastic lattice gases in a non-equilibrium steady state. Journal of Physics A: Mathematical and General, 16(18):4275, 1983.
- [45] N. Tizón-Escamilla, V. Lecomte, and E. Bertin. Effective driven dynamics for onedimensional conditioned Langevin processes in the weak-noise limit. *Journal of Statistical Mechanics: Theory and Experiment*, 2019(1):013201, Jan. 2019.
- [46] H. Touchette. Equivalence and Nonequivalence of Ensembles: Thermodynamic, Macrostate, and Measure Levels. *Journal of Statistical Physics*, 159(5):987–1016, June 2015.
- [47] P. Tsobgni Nyawo and H. Touchette. Large deviations of the current for driven periodic diffusions. *Physical Review E*, 94(3):032101, Sept. 2016.
- [48] H.-T. Yau. Relative entropy and hydrodynamics of Ginzburg-Landau models. Letters in Mathematical Physics, 22(1):63-80, May 1991.
- [49] L. Zarfaty and B. Meerson. Statistics of large currents in the Kipnis-Marchioro-Presutti model in a ring geometry. *Journal of Statistical Mechanics: Theory and Experiment*, 2016(3):033304, Mar. 2016.